UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): June 17, 2015

RESTORGENEX CORPORATION

(Exact name of registrant as specified in its charter)

Delaware

(State or other jurisdiction of incorporation)

> 2150 E. Lake Cook Road, Suite 750 **Buffalo Grove, Illinois** (Address of principal executive offices)

000-24477 (Commission File Number)

30-0645032 (I.R.S. Employer Identification No.)

60089

(Zip Code)

(847) 777-8092

(Registrant's telephone number, including area code)

Not Applicable

(Former name or former address, if changed since last report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

- o Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)
- Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)
- Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))
- o Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Item 1.01 Entry into a Material Definitive Agreement.

The information set forth under Item 5.02 of this Current Report on Form 8-K is incorporated herein by reference.

Item 3.03 Material Modification to Rights of Security Holders.

On June 18, 2015, RestorGenex Corporation ("RestorGenex") changed its state of incorporation from the State of Nevada to the State of Delaware (the "Reincorporation") pursuant to a plan of conversion, dated June 18, 2015 (the "Plan of Conversion"). The Reincorporation was accomplished by the filing of (i) articles of conversion (the "Nevada Articles of Conversion") with the Secretary of State of the State of Nevada, and (ii) a certificate of conversion (the "Delaware Certificate of Incorporation") with the Secretary of State of the State of Delaware. Pursuant to the Plan of Conversion, RestorGenex also adopted new bylaws under Delaware corporate law (the "Delaware Bylaws").

The Reincorporation was previously submitted to a vote of, and approved by, RestorGenex's stockholders at its 2015 Annual Meeting of Stockholders held on June 17, 2015 (the "Annual Meeting") as described under Item 5.07 of this Current Report on Form 8-K. Upon the effectiveness of the Reincorporation:

- the affairs of RestorGenex ceased to be governed by the Nevada Revised Statutes, RestorGenex's existing Articles of Incorporation and its existing Bylaws, and the affairs of RestorGenex became subject to the General Corporation Law of the State of Delaware, the Delaware Certificate of Incorporation and the Delaware Bylaws;
- each outstanding share of the Nevada corporation's common stock converted into an outstanding share of the Delaware corporation's common stock;
- each outstanding option to acquire shares of the Nevada corporation's common stock converted into an equivalent option to acquire, upon the same terms and conditions (including the vesting schedule and $exercise\ price\ per\ share\ applicable\ to\ each\ such\ option),\ the\ same\ number\ of\ shares\ of\ the\ Delaware\ corporation's\ common\ stock;$
- each outstanding warrant or other right to acquire shares of the Nevada corporation's common stock converted into an equivalent warrant or other right to acquire, upon the same terms and conditions the same number of shares of the Delaware corporation's common stock;
- each employee benefit, stock option or other similar plan of the Nevada corporation continued to be an employee benefit, stock option or other similar plan of the Delaware corporation; and
- each director and officer of the Nevada corporation continued to hold his or her respective position with the Delaware corporation.

RestorGenex effected the Reincorporation because the corporate laws of the State of Delaware are more comprehensive, widely-used and extensively interpreted than the corporate laws of other states, including Nevada. As a result of the flexibility and responsiveness of the Delaware corporate laws to the legal and business needs of corporations, many corporations have incorporated in Delaware or have changed their corporate domiciles to Delaware in a manner similar to the Reincorporation. The Delaware judiciary has become particularly familiar with corporate law matters and a substantial body of court decisions has developed construing the laws of Delaware, thus providing greater clarity and predictability

with respect to RestorGenex's corporate legal and governance affairs. Although the Delaware Certificate of Incorporation and the Delaware Bylaws are substantially similar to provisions from RestorGenex's articles of incorporation and bylaws under Nevada corporate law, certain rights of RestorGenex's stockholders are different as a result of the Reincorporation, as described in RestorGenex's Definitive Proxy Statement on Schedule 14A for the Annual Meeting filed with the Securities and Exchange Commission on April 24, 2015, under the section entitled "Proposal No. 6 — Approval of Reincorporation of RestorGenex to the State of Delaware Rights of Stockholders Prior to and After Reincorporation from Nevada to Delaware," which description is incorporated herein by reference

The Reincorporation did not affect any of RestorGenex's contracts with any third parties, and RestorGenex's rights and obligations under such contractual arrangements continue to be rights and obligations of RestorGenex after the Reincorporation. The Reincorporation did not result in any change in headquarters, business, jobs, management, location of any of the offices or facilities, number of employees, assets, liabilities or net worth (other than as a result of the costs incident to the Reincorporation) of RestorGenex.

The foregoing descriptions of the Plan of Conversion, the Nevada Articles of Conversion, the Delaware Certificate of Conversion, the Delaware Certificate of Incorporation and the Delaware Bylaws do not purport to be complete and are qualified in their entirety by reference to the full text of the Plan of Conversion, the Nevada Articles of Conversion, the Delaware Certificate of Conversion, the Delaware Certificate of Incorporation and the Delaware Bylaws, copies of which are filed as Exhibits 2.1, 3.1, 3.2, 3.3 and 3.4, respectively, to this Current Report on Form 8-K and are incorporated herein by reference.

In connection with the Reincorporation, RestorGenex adopted a new form of common stock certificate, a copy of which is filed as Exhibit 4.1 to this Current Report on Form 8-K and is incorporated herein by reference, and a new form of indemnification agreement with its directors and officers, a copy of which is filed as Exhibit 10.1 to this Current Report on Form 8-K and is incorporated herein by reference.

Item 5.02 Departure of Directors or Certain Officers; Election of Directors; Appointment of Certain Officers; Compensatory Arrangements of Certain Officers.

Indemnification Agreements with Directors and Officers

In connection with the Reincorporation, RestorGenex entered into new indemnification agreements with each of its directors and officers, effective as of June 18, 2015, which replace the previous indemnification agreements entered into between RestorGenex and its directors and officers.

The indemnification agreements are substantially similar to those previously entered into between RestorGenex and its directors and officers, except that the new indemnification agreements are governed by Delaware law and have been modified to conform to Delaware law. The indemnification agreements require RestorGenex, among other things, to indemnify the director or officer against specified expenses and liabilities, such as attorneys' fees, judgments, fines and settlements, paid by the individual in connection with any action, suit or proceeding arising out of the individual's status or service as a director or officer of RestorGenex, other than liabilities arising from actions not taken in good faith or of which indemnification would be otherwise unlawful, and to advance expenses incurred by the individual in connection with any proceeding against the individual with respect to which the individual may be entitled to indemnification by RestorGenex.

3

The foregoing description of the indemnification agreements entered into between RestorGenex and each of its directors and officers is qualified in its entirety by reference to the full text of the form of indemnification agreement, which is filed as Exhibit 10.1 to this Current Report on Form 8-K and is incorporated herein by reference.

RestorGenex Corporation 2015 Equity Incentive Plan

At the Annual Meeting on June 17, 2015, RestorGenex's stockholders, upon recommendation of the Board of Directors, approved the RestorGenex Corporation 2015 Equity Incentive Plan (the "2015 Plan"). The material terms of the 2015 Plan are as described in RestorGenex's Definitive Proxy Statement on Schedule 14A for the Annual Meeting filed with the Securities and Exchange Commission on April 24, 2015, under the section entitled "Proposal No. 5 — Approval of RestorGenex Corporation 2015 Equity Incentive Plan," which description is incorporated herein by reference. The description of the 2015 is qualified in its entirety by reference to the full text of the 2015 Plan, which is filed as Exhibit 10.2 to this Current Report on Form 8-K and is incorporated herein by reference.

In connection with the approval of the 2015 Plan, RestorGenex adopted new forms of option agreements to be used for future option grants under the 2015 Plan, copies of which are filed as Exhibits 10.3 and 10.4 to this Current Report on Form 8-K and are incorporated herein by reference.

Item 5.03 Amendments to Articles of Incorporation or Bylaws; Change in Fiscal Year.

The information set forth under Item 3.03 of this Current Report on Form 8-K is incorporated herein by reference.

Item 5.07 Submission of Matters to a Vote of Security Holders.

Proposal No. 1— Election of directors for a term of one year

The 2015 Annual Meeting of Stockholders of RestorGenex was held on June 17, 2015. The final results of the stockholder vote on each proposal brought before the meeting were as follows:

Sol J. Barer, Ph.D.	11,069,594	41,930	708,164	
Isaac Blech	11,069,594	41,930	708,164	
Rex Bright	11,085,994	25,530	708,164	
Stephen M. Simes	11,019,897	91,627	708,164	
Nelson K. Stacks	11,069,594	41,930	708,164	
	For	Against	Abstain	Broker Non-Votes
Proposal No. 2—Ratification of selection of Deloitte & Touche LLP as independent				
registered public accounting firm	11,770,911	25,400	23,377	0
				Broker

Withhold

Non-Votes

	For	Against	Abstain	Non-Votes
Proposal No. 3—Advisory vote on executive compensation	10,875,073	197,301	39,150	708,164

4

	1 Year	2 Years	3 Years	Abstain
Proposal No. 4—Indicate on an advisory basis, whether future votes to approve the compensation				
of executive officers should occur every one, two or three years	10,109,843	920,029	48,152	33,500
				Broke
	For	Against	Abstain	Non-Vo
Proposal No. 5—Approval of the RestorGenex Corporation 2015 Equity Incentive Plan	10 846 195	190 552	74 777	708.1

rioposarrior of ripprovar of the restor deficit dorporation 2015 Equity meetave rian	10,040,133	130,332	7-1,777	700,104
				Broker
	For	Against	Abstain	Non-Votes
Proposal No. 6—Approval of the reincorporation of RestorGenex from the State of				
Nevada to the State of Delaware	11,098,223	400	12,901	708,164

	FUI	Against	AUStaili	INUIT- VUICS
Proposal No. 7—Adjournment of the meeting, if necessary, to solicit additional proxies if				
there are not sufficient votes in favor of Proposal No. 6.	11,757,864	18,447	43,377	0

the required vote.

With respect to Proposal No. 1, each of the director nominees was elected by RestorGenex's stockholders by the required vote and Proposal Nos. 2, 3, 5, 6 and 7 were approved by RestorGenex's stockholders by

With respect to Proposal No. 4, RestorGenex's stockholders expressed a preference for an advisory vote on executive compensation to be held every year. As this vote is consistent with the recommendation of the Board of Directors of RestorGenex, the Board of Directors has determined that RestorGenex will hold a non-binding advisory vote on executive compensation every year.

Item 9.01. Financial Statements and Exhibits

(d) Exhibits. The following exhibits are being filed with this Current Report on Form 8-K:

Exhibit No.	Description
2.1	Plan of Conversion, dated June 18, 2015 (filed herewith)
3.1	Articles of Conversion, as filed with the Secretary of State of the State of Nevada on June 18, 2015 (filed herewith)
3.2	Certificate of Conversion, as filed with the Secretary of State of the State of Delaware on June 18, 2015 (filed herewith)
3.3	Certificate of Incorporation of RestorGenex Corporation (filed herewith)
3.4	Bylaws of RestorGenex Corporation (filed herewith)
4.1	Specimen Stock Certificate (filed herewith)

Exhibit No.	Description
10.1	Form of Indemnification Agreement between RestorGenex Corporation and Each of its Directors and Officers (filed herewith)
10.2	RestorGenex Corporation 2015 Equity Incentive Plan (filed herewith)
10.3	Form of Incentive Stock Option Agreement under the RestorGenex 2015 Equity Incentive Plan (filed herewith)
10.4	Form of Non-Statutory Stock Option Agreement under the RestorGenex 2015 Equity Incentive Plan (filed herewith)
	6

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, as amended, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Dated: June 18, 2015 RESTORGENEX CORPORATION

> By: Name: /s/ Phillip B. Donenberg
> Phillip B. Donenberg
> Chief Financial Officer and Secretary

Title:

RESTORGENEX CORPORATION CURRENT REPORT ON FORM 8-K

EXHIBIT INDEX

Exhibit No.	Description	Method of Filing
2.1	Plan of Conversion, dated June 18, 2015	Filed herewith
3.1	Articles of Conversion, as filed with the Secretary of State of the State of Nevada on June 18, 2015	Filed herewith
3.2	Certificate of Conversion, as filed with the Secretary of State of the State of Delaware on June 18, 2015	Filed herewith
3.3	Certificate of Incorporation of RestorGenex Corporation	Filed herewith
3.4	Bylaws of RestorGenex Corporation	Filed herewith
4.1	Specimen Stock Certificate	Filed herewith
10.1	Form of Indemnification Agreement between RestorGenex Corporation and Each of its Directors and Officers	Filed herewith
10.2	RestorGenex Corporation 2015 Equity Incentive Plan	Filed herewith
10.3	Form of Incentive Stock Option Agreement under the RestorGenex Corporation 2015 Equity Incentive Plan	Filed herewith
10.4	Form of Non-Statutory Stock Option Agreement under the RestorGenex Corporation 2015 Equity Incentive Plan	Filed herewith
	8	

PLAN OF CONVERSION OF RESTORGENEX CORPORATION A NEVADA CORPORATION INTO RESTORGENEX CORPORATION,

A DELAWARE CORPORATION

THIS PLAN OF CONVERSION, dated as of June 17, 2015 (including all of the Exhibits attached hereto, this "Plan"), is hereby adopted by RestorGenex Corporation, a Nevada corporation, in order to set forth the terms, conditions and procedures governing the conversion of RestorGenex Corporation from a Nevada corporation to a Delaware corporation pursuant to Section 265 of the General Corporation Law of the State of Delaware, as amended (the "<u>DGCL</u>"), and Section 92A.120 of the Nevada Revised Statutes, as amended (the "<u>NRS</u>").

RECITALS

WHEREAS, RestorGenex Corporation is a corporation organized and existing under the laws of the State of Nevada (the "Converting Entity");

WHEREAS, the Board of Directors of the Converting Entity has determined that it would be advisable and in the best interests of the Converting Entity and its stockholders for the Converting Entity to convert from a Nevada corporation to a Delaware corporation pursuant to Section 265 of the DGCL and Sections 92A.120 and 92A.250 of the NRS;

WHEREAS, the form, terms and provisions of this Plan have been authorized, approved and adopted by the Board of Directors of the Converting Entity;

WHEREAS, the Board of Directors of the Converting Entity has submitted this Plan to the stockholders of the Converting Entity for approval; and

WHEREAS, this Plan has been authorized, approved and adopted by the holders of a majority of the voting power of the stockholders of the Converting Entity.

NOW, THEREFORE, the Converting Entity hereby adopts this Plan as follows:

PLAN OF CONVERSION

Conversion; Effect of Conversion

- Upon the Effective Time (as defined in Section 3 below), the Converting Entity shall be converted from a Nevada corporation to a Delaware corporation pursuant to Section 265 of the DGCL and Sections 92A.120 and 92A.250 of the NRS (the "Conversion") and the Converting Entity, as converted to a Delaware corporation (the "Converted Entity"), shall thereafter be subject to all of the provisions of the DGCL, except that notwithstanding Section 106 of the DGCL, the existence of the Converted Entity shall be deemed to have commenced on the date the Converting Entity commenced its existence in the State of Nevada.
- Upon the Effective Time, by virtue of the Conversion and without any further action on the part of the Converting Entity or its stockholders, the Converted Entity shall, for all purposes of the laws of the (b) State of Delaware, be deemed to be the same entity as the Converting Entity existing immediately prior to the Effective Time. Upon the Effective Time, by virtue of the Conversion and without any further action on the part of the Converting Entity or its stockholders, for all purposes of the laws of the State of Delaware, all of the rights, privileges and powers of the Converting Entity existing immediately prior to the Effective Time, and all property, real, personal and mixed, and all debts due to the Converting Entity existing immediately prior to the Effective Time, as well as all other things and causes of action belonging to the Converting Entity existing immediately prior to the Effective Time, shall remain vested in the Converted Entity and shall be the property of the Converted Entity and the title to any real property vested by deed or otherwise in the Converting Entity existing immediately prior to the Effective Time shall not revert or be in any way impaired by reason of the Conversion; but all rights of creditors and all liens upon any property of the Converting Entity existing immediately prior to the Effective Time shall be preserved unimpaired, and all debts, liabilities and duties of the Converting Entity existing immediately prior to the Effective Time shall remain attached to the Converted Entity upon the Effective Time, and may be enforced against the Converted Entity to the same extent as if said debts, liabilities and duties had originally been incurred or contracted by the Converted Entity in its capacity as a corporation of the State of Delaware. The rights, privileges, powers and interests in property of the Converting Entity existing immediately prior to the Effective Time, as well as the debts, liabilities and duties of the Converting Entity existing immediately prior to the Effective Time, shall not be deemed, as a consequence of the Conversion, to have been transferred to the Converted Entity upon the Effective Time for any purpose of the laws of the State of Delaware.
- (c) The Conversion shall not be deemed to affect any obligations or liabilities of the Converting Entity incurred prior to the Conversion or the personal liability of any person incurred prior to the Conversion.
- (d) Upon the Effective Time, the name of the Converted Entity shall remain unchanged and continue to be "RestorGenex Corporation".
- (e) The Converting Entity intends for the Conversion to constitute a tax-free reorganization qualifying under Section 368(a) of the Internal Revenue Code of 1986, as amended.
- Filings. As promptly as practicable following the adoption of this Plan by the Board of Directors and the stockholders of the Converting Entity, the Converting Entity shall cause the Conversion to be effective by:
 - executing and filing (or causing the execution and filing of) Articles of Conversion pursuant to Section 92A.205 of the NRS, substantially in the form of (a)

Exhibit A hereto (the "Nevada Articles of Conversion"), with the Secretary of State of the State of Nevada;

- (b) executing and filing (or causing the execution and filing of) a Certificate of Conversion pursuant to Sections 103 and 265 of the DGCL, substantially in the form of Exhibit B hereto (the "Delaware Certificate of Conversion"), with the Secretary of State of the State of Delaware; and
- executing and filing (or causing the execution and filing of) a Certificate of Incorporation of the Converted Entity, substantially in the form of Exhibit C hereto (the "Delaware Certificate of Incorporation"), with the Secretary of State of the State of Delaware.
- Effective Time. The Conversion shall become effective upon the last to occur of the filing of the Nevada Articles of Conversion, the Delaware Certificate of Conversion and the Delaware Certificate of Incorporation (the time of the effectiveness of the Conversion, the "Effective Time").

Effect of Conversion

- Effect on Common Stock. Upon the Effective Time, by virtue of the Conversion and without any further action on the part of the Converting Entity or its stockholders, each share of Common Stock, (a) \$0.001 par value per share, of the Converting Entity ("Converting Entity Common Stock") that is issued and outstanding immediately prior to the Effective Time shall convert into one validly issued, fully paid and nonassessable share of Common Stock, \$0.001 par value per share, of the Converted Entity ("Converted Entity Common Stock").
- (b) Effect on Outstanding Stock Options. Upon the Effective Time, by virtue of the Conversion and without any further action on the part of the Converting Entity or its stockholders, each option to acquire shares of Converting Entity Common Stock outstanding immediately prior to the Effective Time shall convert into an equivalent option to acquire, upon the same terms and conditions (including the vesting schedule and exercise price per share applicable to each such option) as were in effect immediately prior to the Effective Time, the same number of shares of Converted Entity Common Stock.
- Effect on Outstanding Warrants or Other Rights. Upon the Effective Time, by virtue of the Conversion and without any further action on the part of the Converting Entity or its stockholders, each (c) warrant or other right to acquire shares of Converting Entity Common Stock outstanding immediately prior to the Effective Time shall convert into an equivalent warrant or other right to acquire, upon the same terms and conditions (including the vesting schedule and exercise price per share applicable to each such warrant or other right) as were in effect immediately prior to the Effective Time, the same number of shares of Converted Entity Common Stock.

- Effect on Stock Certificates. All of the outstanding certificates representing shares of Converting Entity Common Stock immediately prior to the Effective Time shall be deemed for all purposes to (d) continue to evidence ownership of and to represent the same number of shares of Converted Entity Common Stock.
- Effect on Employee Benefit, Equity Incentive or Other Similar Plans. Upon the Effective Time, by virtue of the Conversion and without any further action on the part of the Converting Entity or its stockholders, each employee benefit plan, equity incentive plan or other similar plan to which the Converting Entity is a party shall continue to be a plan of the Converted Entity. To the extent that any (e) such plan provides for the issuance of Converting Entity Common Stock, upon the Effective Time, such plan shall be deemed to provide for the issuance of Converted Entity Common Stock.

- (f) Effect of Conversion on Directors and Officers. Upon the Effective Time, by virtue of the Conversion and without any further action on the part of the Converting Entity or its stockholders, the members of the Board of Directors and the officers of the Converting Entity holding their respective offices in the Converting Entity existing immediately prior to the Effective Time shall continue in their respective offices as members of the Board of Directors and officers, respectively, of the Converted Entity.
- Further Assurances. If, at any time after the Effective Time, the Converted Entity shall determine or be advised that any deeds, bills of sale, assignments, agreements, documents or assurances or any other acts or things are necessary, desirable or proper, consistent with the terms of this Plan, (a) to vest, perfect or confirm, of record or otherwise, in the Converted Entity its right, title or interest in, to or under any of the rights, privileges, immunities, powers, purposes, franchises, properties or assets of the Converted Entity existing immediately prior to the Effective Time, or (b) to otherwise carry out the purposes of this Plan, the Converted Entity and its officers and directors (or their designees), are hereby authorized to solicit in the name of the Converted Entity any third party, to execute and deliver, in the name and on behalf of the Converted Entity, all such deeds, bills of sale, assignments, agreements, documents or other documents required to be delivered by any third party, to execute and deliver, in the name and on behalf of the Converted Entity, all such other acts and things necessary, desirable or proper to vest, perfect or confirm its right, title or interest in, to or under any of the rights, privileges, immunities, powers, purposes, franchises, properties or assets of the Converting Entity existing immediately prior to the Effective Time and otherwise to carry out the purposes of this Plan.
- 6. <u>Delaware Bylaws</u>. Upon the Effective Time, the bylaws of the Converted Entity shall be the Bylaws of RestorGenex Corporation, substantially in the form of Exhibit D hereto.
- 7. <u>Delaware Indemnification Agreements.</u> As promptly as practicable following the Effective Time, the Converted Entity shall enter into an Indemnification Agreement substantially in the form of <u>Exhibit E</u> hereto with each member of the Board of Directors of the Converted Entity and each executive officer of the Converted Entity.

- 8. <u>Copy of Plan of Conversion</u>. After the Conversion, a copy of this Plan will be kept on file at the offices of the Converted Entity, and any stockholder of the Converted Entity (or former stockholder of the Converting Entity) may request a copy of this Plan at no charge at any time.
- 9. <u>Termination</u>. At any time prior to the Effective Time, this Plan may be terminated and the transactions contemplated hereby may be abandoned by action of the Board of Directors of the Converting Entity if, in the opinion of the Board of Directors of the Converting Entity, such action would be in the best interests of the Converting Entity and its stockholders. In the event of termination of this Plan, this Plan shall become void and of no further force or effect
- 10. Third-Party Beneficiaries. This Plan shall not confer any rights or remedies upon any person other than as expressly provided herein.
- 11. <u>Severability.</u> Whenever possible, each provision of this Plan will be interpreted in such manner as to be effective and valid under applicable law, but if any provision of this Plan is held to be prohibited by or invalid under applicable law, such provision will be ineffective only to the extent of such prohibition or invalidity, without invalidating the remainder of this Plan.

[Remainder of page intentionally left blank]

5

IN WITNESS WHEREOF, the undersigned hereby causes this Plan to be duly executed as of the date hereof.

RESTORGENEX CORPORATION

a Nevada corporation

By: /s/ Stephen M. Simes

Stephen M. Simes
Its: Chief Executive Officer





BARBARA K. CEGAVSKE Secretary of State 202 North Carson Street Carson City, Nevada 89701-4201 (775) 684-5708 Website: www.nvsos.gov

Articles of Conversion

(PURSUANT TO NRS 92A.205)

Page 1

USE BLACK INK ONLY - DO NOT HIGHLIGHT

ABOVE SPACE IS FOR OFFICE USE ONLY

PLEASE NOTE: The charter document for the resulting entity must be submitted/filed simultaneously with the articles of conversion.

Articles of Conversion (Pursuant to NRS 92A.205)

1. Name and jurisdiction of organization of constituent entity and resulting entity:

RestorGenex Corporation	
Name of constituent entity	
Nevada	Corporation
Jurisdiction	Entity type *
and,	
RestorGenex Corporation	
Name of resulting entity	
Delaware	Corporation
Jurisdiction	Entity type *
law of the jurisdiction governing to 3. Location of plan of conversion: (cl	
law of the jurisdiction governing the same of the jurisdiction governing the same of the s	he constituent entity.
law of the jurisdiction governing the state of the state	he constituent entity. heck one) on is attached to these articles. n of conversion is on file at the registered office or principal
aw of the jurisdiction governing the state of the state o	the constituent entity. The constituent entity. The constituent entity. The conversion is on file at the registered office or principal sulting entity. The conversion for the resulting domestic limited partnership is
aw of the jurisdiction governing the state of the purisdiction governing the state of the state	the constituent entity. The constituent entity. The constituent entity. The conversion is on file at the registered office or principal sulting entity. The conversion for the resulting domestic limited partnership is



BARBARA K. CEGAVSKE Secretary of State 202 North Carson Street Carson City, Nevada 89701-4201 (775) 684-5708 Website: www.nvsos.gov

Articles of Conversion

(PURSUANT TO NRS 92A.205)

Page 2

USE BLACK INK ONLY - DO NOT HIGHLIGHT

ABOVE SPACE IS FOR OFFICE USE ONLY

Attn:	Restor	Genex Corporation			
c/o:	615 Sc	nal Corporate Research outh Dupont Hwy. , Delaware 19901	h, Ltd.		
ffective date and t	ime of f	iling: (optional) (mus	st not be later t	han 90 days after t	the certificate is filed)
D	ate:	June 18, 2015	Time:	5:00 pm ET	
. If constituent entit	ty is a N	evada entity: an office			eneral partners of each
If constituent entit levada limited partr ompany with mana managing partner	ty is a N nership o	evada entity: an office or limited-liability limite	ed partnership; a ire no managers	manager of each h	Nevada limited-liability Nevada business trust;
I. If constituent entity levada limited partrompany with mana managing partner hapter 87).	ty is a Nonership of gers or of a New	evada entity: an office or limited-liability limite one member if there a vada limited-liability pa	ed partnership; a ire no managers artnership (a.k.a	manager of each i a trustee of each i general partnershi	Nevada limited-liability Nevada business trust;
I. If constituent entit Nevada limited partre company with mana managing partner chapter 87). If constituent entit he law governing it.	ty is a Nonership of a New of a New ty is a fo	evada entity: an office or limited-liability limite one member if there a vada limited-liability pa	ed partnership; a ire no managers artnership (a.k.a	manager of each i a trustee of each i general partnershi	Nevada limited-liability Nevada business trust; p governed by NRS
Nevada limited partre company with mana I managing partner chapter 87).	ty is a Nonership of a New of a New ty is a fo	evada entity: an office or limited-liability limite one member if there a vada limited-liability pa reign entity: must be	ed partnership; a ire no managers artnership (a.k.a	manager of each i a trustee of each i general partnershi	Nevada limited-liability Nevada business trust; p governed by NRS
I. If constituent entit Nevada limited partri company with main managing partner chapter 87). 2. If constituent entit he law governing it. RestorGenex Corpo Name of constituent	ty is a Ninership of gers or of a New ty is a for or a new ty is a for or a new ty is a for a new ty is a new ty is a for a new ty is a for a new ty is a new ty is a for a new ty is a new ty is a for a new ty is a new ty is a for a new ty is a new ty is a for a new ty is a for a new ty is a new ty is a new ty is a new ty is a for a new ty is a for a new ty is a new ty i	evada entity: an office or limited-liability limite one member if there a vada limited-liability pa reign entity: must be	ed partnership; a rre no managers artnership (a.k.a signed by the co	n manager of each i ; a trustee of each i general partnershi onstituent entity in th	Nevada limited-liability Nevada business trust; p governed by NRS ne manner provided by
I. If constituent entit Nevada limited partro company with mana managing partner shapter 87). L. If constituent entit he law governing it. RestorGenex Corpo	ty is a Ninership of gers or of a New ty is a for or a new ty is a for or a new ty is a for a new ty is a new ty is a for a new ty is a for a new ty is a new ty is a for a new ty is a new ty is a for a new ty is a new ty is a for a new ty is a new ty is a for a new ty is a for a new ty is a new ty is a new ty is a new ty is a for a new ty is a for a new ty is a new ty i	evada entity: an office or limited-liability limite one member if there a vada limited-liability pa reign entity: must be	ed partnership; a rre no managers artnership (a.k.a signed by the co	manager of each i a trustee of each i general partnershi	Nevada limited-liability Nevada business trust; p governed by NRS

* Pursuant to NRS 92A.205(4) if the conversion takes effect on a later date specified in the articles of conversion pursuant to NRS 92A 240, the constituent document filed with the Secretary of State pursuant to paragraph (b) subsection 1 must state the name and the jurisdiction of the constituent entity and that the existence of the resulting entity does not begin until the later date. This statement must be included within the resulting entity's articles.

FILING FEE: \$350.00

IMPORTANT: Failure to include any of the above information and submit with the proper fees may cause this filing to be rejected.

This form must be accompanied by appropriate fees.

Nevada Secretary of State 92A Conversion Page 2 Revised: 1-5-15

Reset

STATE OF DELAWARE CERTIFICATE OF CONVERSION FROM A NON-DELAWARE CORPORATION TO A DELAWARE CORPORATION PURSUANT TO SECTION 265 OF THE DELAWARE GENERAL CORPORATION LAW

 $\textbf{FIRST:} \ \ \text{The jurisdiction where the Non-Delaware Corporation first formed is Nevada.}$

SECOND: The jurisdiction immediately prior to filing this Certificate is Nevada.

 $\textbf{THIRD}: \ \ \text{The date the Non-Delaware Corporation first formed is January 10, 1995.}$

FOURTH: The name of the Non-Delaware Corporation immediately prior to filing this Certificate is RestorGenex Corporation.

FIFTH: The name of the Corporation as set forth in the Certificate of Incorporation is RestorGenex Corporation.

 $\textbf{SIXTH:} \ \text{The Certificate of Conversion shall be effective as of 5:00 p.m.} \ \text{Eastern Time on June 18, 2015.}$

IN WITNESS WHEREOF, the undersigned being duly authorized to sign on behalf of the converting Non-Delaware Corporation have executed this Certificate on the 17th day of June, A.D. 2015.

RESTORGENEX CORPORATION

a Nevada corporation

/s/ Stephen M. Simes Stephen M. Simes Chief Executive Officer By:

Its:

CERTIFICATE OF INCORPORATION OF RESTORGENEX CORPORATION

ARTICLE I NAME

The name of the corporation is RestorGenex Corporation (the "Corporation").

ARTICLE II REGISTERED OFFICE AND AGENT

The address of the registered office of the Corporation in the State of Delaware is 615 South Dupont Hwy., in the City of Dover, Zip Code of 19901, County of Kent. The name of the registered agent of the Corporation at that address is National Corporate Research, Ltd.

ARTICLE III PURPOSE

The purpose of the Corporation is to engage in any lawful act or activity for which a corporation may be organized under the General Corporation Law of the State of Delaware ("DGCL").

ARTICLE IV CAPITAL STOCK

- A. The total number of shares of common stock which the Corporation shall have authority to issue is 1,000,000,000, at a par value of \$0.001 per share ("Common Stock"), and the total number of shares of preferred stock which the Corporation shall have authority to issue is 5,000,000, at a par value of \$0.001 per share ("Preferred Stock").
 - 1. Common Stock. All preferences, voting powers, relative, participating, optional or other special rights and privileges, and qualifications, limitations, or restrictions of the Common Stock are expressly made subject and subordinate to those that may be fixed with respect to any shares of the Preferred Stock. Except as otherwise required by law or this Certificate of Incorporation, each share of Common Stock shall entitle the holder thereof to one (1) vote, in person or by proxy, on each matter submitted to a vote of stockholders of the Corporation. Subject to the preferential rights of the Preferred Stock, the holders of shares of Common Stock shall be entitled to receive, when and if declared by the Board of Directors, out of the assets of the Corporation which are by law available therefor, dividends payable either in cash, in property or in shares of capital stock. In the event of any dissolution, liquidation or winding up of the affairs of the Corporation, after distribution in full of the preferential amounts, if any, to be distributed to the holders of shares of the Preferred Stock, holders of Common Stock shall be entitled, unless otherwise provided by law or this Certificate of Incorporation, to receive all of the remaining assets of the Corporation of whatever kind available for

distribution to stockholders ratably in proportion to the number of shares of Common Stock held by them respectively.

2. Preferred Stock. The Preferred Stock may be issued from time to time in one or more series, as determined by the Board of Directors of the Corporation (the "Board of Directors"). The Board of Directors is expressly authorized to provide for the issue, in one or more series, of all or any of the remaining shares of Preferred Stock and, in the resolution or resolutions providing for such issue, to establish for each such series the number of its shares, the voting powers, full or limited, of the shares of such series, or that such shares shall have no voting powers, and the designations, preferences and relative, participating, optional or other special rights, if any, of the shares of such series, and any qualifications, limitations or restrictions thereof. The powers, preferences and relative, participating, optional and other special rights of each series of preferred stock and the qualifications, limitations or restrictions thereof, if any, may differ from those of any and all other series at any time outstanding. The Board of Directors is further expressly authorized to increase or decrease (but not below the number of shares of any series then outstanding) the number of shares of any series, the number of which was fixed by it, subsequent to the issuance of shares of such series then outstanding, subject to the powers, preferences and rights, and the qualifications, limitations and restrictions thereof stated in this Certificate of Incorporation or the resolution of the Board of Directors originally fixing the number of shares of such series is so decreased, then the shares constituting such decrease shall resume the status which they had prior to the adoption of the resolution originally fixing the number of shares of such series.

ARTICLE V EXCULPATION AND INDEMNIFICATION

- A. <u>Limitation of Liability.</u> A director of the Corporation shall not be personally liable to the Corporation or its stockholders for monetary damages for breach of fiduciary duty as a director, except to the extent such exemption from liability or limitation thereof is not permitted under the DGCL as it presently exists or may hereafter be amended. Any amendment, modification or repeal of the foregoing sentence shall not adversely affect any right arising prior to the time of such amendment, modification or repeal.
- B. Right of Indemnification. The Corporation shall indemnify and hold harmless, to the fullest extent permitted by applicable law as it presently exists or may hereafter be amended, any person (a "Covered Person") who was or is made or is threatened to be made a party or is otherwise involved in any action, suit or proceeding, whether civil, criminal, administrative or investigative (a "Proceeding"), by reason of the fact that he or she, or a person for whom he or she is the legal representative, is or was a director, officer, employee or agent of the Corporation or, while a director, officer, employee or agent of the Corporation as a director, officer, employee or agent of another corporation or of a partnership, joint venture, trust, enterprise or nonprofit entity, including service with respect to employee benefit plans, against all liability and loss suffered and expenses (including attorneys' fees) reasonably incurred by such Covered Person. Notwithstanding the preceding sentence, except as otherwise provided in section D of this Article V, the Corporation shall not be required

2

to indemnify a Covered Person in connection with a Proceeding (or part thereof) commenced by such Covered Person unless the commencement of such Proceeding (or part thereof) by the Covered Person was authorized in the specific case by the Board of Directors.

- C. <u>Prepayment of Expenses</u>. The Corporation shall to the fullest extent not prohibited by applicable law pay the expenses (including attorneys' fees) incurred by a Covered Person in defending any Proceeding in advance of its final disposition, provided, however, that, to the extent required by law, such payment of expenses in advance of the final disposition of the Proceeding shall be made only upon receipt of an undertaking by the Covered Person to repay all amounts advanced if it should be ultimately determined that the Covered Person is not entitled to be indemnified under this Article V or otherwise.
- D. <u>Claims</u>. If a claim for indemnification (following the final disposition of the Proceeding with respect to which indemnification is sought, including any settlement of such Proceeding) or advancement of expenses under this Article V is not paid in full within thirty (30) days after a written claim therefor by the Covered Person has been received by the Corporation, the Covered Person may file suit to recover the unpaid amount of such claim and, if successful in whole or in part, shall be entitled to be paid the expense of prosecuting such claim to the fullest extent permitted by applicable law. In any such action the Corporation shall have the burden of proving that the Covered Person is not entitled to the requested indemnification or advancement of expenses under this Article V and applicable law.
- E. Non-Exclusivity of Rights. The rights conferred on any Covered Person by this Article V shall not be exclusive of any other rights which such Covered Person may have or hereafter acquire under any statute, any other provision of this Certificate of Incorporation, the Bylaws of the Corporation, or any agreement, vote of stockholders or disinterested directors or otherwise.
- F. Insurance. The Corporation may purchase and maintain insurance on behalf of any person who is or was a director, officer, employee or agent of the Corporation, or is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation or of a partnership, joint venture, trust, enterprise or nonprofit entity, including service with respect to employee benefit plans, against any liability asserted against such person and incurred by such person in any such capacity, or arising out of such person's status as such, whether or not the Corporation would have the power to indemnify such person against such liability under this Article V, the DGCL or otherwise.
- G. <u>Amendment or Repeal</u>. Any right to indemnification or to advancement of expenses of any Covered Person arising hereunder shall not be eliminated or impaired by an amendment to or repeal of this Article V after the occurrence of the act or omission that is the subject of the civil, criminal, administrative or investigative action, suit or proceeding for which indemnification or advancement of expenses is sought.
- H. <u>Other Indemnification and Advancement of Expenses</u>. This Article V shall not limit the right of the Corporation, to the extent and in the manner permitted by law, to indemnify and to advance expenses to persons other than Covered Persons when and as authorized by appropriate corporate action.

3

ARTICLE VI MANAGEMENT

- A. The management of the business and the conduct of the affairs of the Corporation shall be vested in the Board of Directors. The Board of Directors shall fix the number of directors that constitute the whole Board of Directors in the manner provided in the Bylaws of the Corporation, subject to any restrictions that may be set forth in this Certificate of Incorporation.
- B. In furtherance and not in limitation of the powers conferred by statute, the Board of Directors is expressly empowered to adopt, amend or repeal the Bylaws of the Corporation or adopt new Bylaws of the Corporation without any action on the part of the stockholders. Any adoption, amendment or repeal of the Bylaws of the Corporation by the Board of Directors shall require the approval of a majority of the directors then in office. The stockholders of the Corporation shall also have the power to adopt, amend or repeal the Bylaws of the Corporation; provided, however, that, in addition to any vote of the holders of any class or series of stock of the Corporation required by law or by this Certificate of Incorporation, the affirmative vote of the holders of at least sixty-six and two-thirds percent (66-2/3%) of the voting power of all of the then-outstanding shares of the capital stock of the Corporation entitled to vote generally in the election of directors, voting together as a single class, shall be required to adopt, amend or repeal any provision of the Bylaws of the Corporation.
- C. The Corporation reserves the right to amend, alter, change or repeal any provision contained in this Certificate of Incorporation, in the manner now or hereafter prescribed by the DGCL, and all rights conferred upon stockholders herein are granted subject to this reservation.

ARTICLE VII STOCKHOLDER MEETINGS

Meetings of stockholders may be held within or without the State of Delaware, as the Bylaws of the Corporation may provide. Elections of directors need not be by written ballot unless and except to the extent that the Bylaws of the Corporation so provide. Any action required to or which may be taken at a meeting of stockholders of the corporation may be taken without a meeting if authorized by a writing signed by all of the holders of shares who would be entitled to vote upon the action at a meeting for such purpose.

4

ARTICLE VIII INCORPORATOR

The name and mailing address of the incorporator of the Corporation are as follows:

Amy E. Culbert Oppenheimer Wolff & Donnelly LLP Campbell Mithun Tower, Suite 2000 222 South Ninth Street Minneapolis, MN 55402

ARTICLE IX EFFECTIVE TIME

This Certificate of Incorporation shall be effective as of 5:00 p.m. Eastern Time on June 18, 2015.

The undersigned, being the incorporator named above, for the purpose of forming a corporation pursuant to the DGCL, does hereby make this Certificate of Incorporation, hereby acknowledging, declaring and certifying that the foregoing Certificate of Incorporation is the undersigned's act and deed and the facts herein stated are true, and accordingly has hereunto set the undersigned's hand this 17th day of June, 2015.

INCORPORATOR:

/s/ Amy E. Culbert

Amy E. Culbert

BYLAWS OF RESTORGENEX CORPORATION

ARTICLE I

- 1.1 <u>Registered Office</u>. The address of the registered office of RestorGenex Corporation (the "<u>Corporation</u>") in the State of Delaware shall be 615 South Dupont Hwy., Dover, Kent County, Delaware 19901. The name of the registered agent of the Corporation at that address is National Corporate Research, Ltd.
- 1.2 Other Offices. The Corporation may have other offices, both within and without the State of Delaware, as the Board of Directors of the Corporation (the "Board of Directors"), from time to time shall determine or the business of the Corporation may require.

ARTICLE II STOCKHOLDERS' MEETINGS

2.1 Place of Meetings. Meetings of the stockholders of the Corporation may be held at any place, either within or without the State of Delaware, as may be determined from time to time by the Board of Directors. The Board of Directors may, in its sole discretion, determine that a meeting of stockholders hall not be held at any place, but may instead be held solely by means of remote communication as authorized by Section 211(a)(2) of the General Corporation Law of the State of Delaware (as amended from time to time, the "DGCL"). In the absence of any such designation or determination, meetings of the stockholders of the Corporation shall be held at the Corporation's principal executive office.

2.2 <u>Annual Meetings</u>.

- (a) The annual meeting of the stockholders of the Corporation, for the purpose of election of directors and for such other business as may lawfully come before it, shall be held on such date and at such time as may be designated from time to time by the Board of Directors. Nominations of persons for election to the Board of Directors of the Corporation and the proposal of business to be considered by the stockholders may be made at an annual meeting of stockholders: (i) pursuant to the Corporation's notice with respect to such meeting; (ii) by or at the direction of the Board of Directors; or (iii) by any stockholder of the Corporation who was a stockholder of record at the time of giving the stockholder's notice provided for in the following subsection (b), who is entitled to vote at the meeting and who complied with the notice procedures set forth below in this Section 2.2.
- (b) At an annual meeting of the stockholders, only such business shall be conducted as shall have been properly brought before the meeting. For nominations or other business to be properly brought before an annual meeting by a stockholder pursuant to Section 2.2(a)(iii) above, (i) the stockholder must have given timely notice thereof in writing to the Secretary of the Corporation, (ii) such other business must be a proper

matter for stockholder action under the DGCL, and (iii) if the stockholder, or the beneficial owner on whose behalf any such proposal or nomination is made, has provided the Corporation with a Solicitation Notice (as defined below in Section 2.2(d)(iii)(C)(2)), such stockholder or beneficial owner must, in the case of a proposal, have delivered a proxy statement and form of proxy to holders of at least the percentage of the Corporation's voting shares required under applicable law or the Corporation's Certificate of Incorporation (as the same may be amended and/or restated from time to time, the "Certificate of Incorporation") or these Bylaws (as the same may be amended and/or restated from time to time, the "Bylaws") to carry any such proposal, or, in the case of a nomination or nominations, have delivered a proxy statement and form of proxy to holders of a percentage of the Corporation's voting shares reasonably believed by such stockholder or beneficial owner to be sufficient to elect the nominee or nominees proposed to be nominated by such stockholder, and must, in either case, have included in such materials the Solicitation Notice.

- (c) To be timely, a stockholder's notice shall be delivered to the Secretary at the principal executive offices of the Corporation not later than the close of business on the ninetieth (90th) day, nor earlier than the close of business on the one hundred twentieth (120th) day, prior to the first anniversary of the date of the proxy statement delivered to stockholders in connection with the preceding year's annual meeting; provided, however, that in the event (i) the date of the annual meeting is advanced more than thirty (30) days prior to or delayed by more than thirty (30) days after the anniversary of the preceding year's annual meeting, (ii) no proxy statement was delivered to stockholders in connection with the preceding year's annual meeting, or (iii) the Corporation did not hold an annual meeting in the preceding year, notice by the stockholder to be timely must be so delivered not earlier than the close of business on the ninetieth (90th) day prior to such annual meeting and not later than the close of business on the later of the sixtieth (60th) day prior to such annual meeting or the tenth (10th) day following the day on which public announcement of the date of such meeting is first made. In no event shall the public announcement of an adjournment of an annual meeting commence a new time period for the giving of a stockholder's notice as described above.
 - (d) Such stockholder's notice shall set forth:
 - (i) as to each person whom the stockholder proposes to nominate for election or reelection as a director, all information relating to such person that is required to be disclosed in solicitations of proxies for election of directors, or is otherwise required, in each case pursuant to Regulation 14A under the Securities Exchange Act of 1934, as amended (as amended from time to time, the "1934 Act") (including such person's written consent to being named in the proxy statement as a nominee and to serving as a director if elected);
 - (ii) as to any other business that the stockholder proposes to bring before the meeting, a brief description of the business desired to be brought before the meeting, the reasons for conducting such business at the meeting and any material interest in such business of such stockholder and the beneficial owner, if any, on whose behalf the proposal is made; and

2

- (iii) as to the stockholder giving the notice and the beneficial owner, if any, on whose behalf the nomination or proposal is made:
 - (A) the name and address of such stockholder, as they appear on the Corporation's books, and of such beneficial owner;
- (B) (1) the class and number of shares of the Corporation which are owned beneficially and of record by such stockholder and such beneficial owner; (2) any option, warrant, convertible security, stock appreciation right, or similar right with an exercise or conversion privilege or a settlement payment or mechanism at a price related to any class or series of shares of the Corporation or with a value derived in whole or in part from the value of any class or series of shares of the Corporation, whether such instrument or right shall be subject to settlement in the underlying class or series of capital stock of the Corporation or otherwise (a "Derivative Instrument") directly or indirectly owned beneficially by such stockholder and such beneficial owner and any other direct or indirect opportunity to profit or share in any profit derived from any increase or decrease in the value of shares of the Corporation; (3) any proxy, contract, arrangement, understanding, or relationship pursuant to which such stockholder has a right to vote any shares of any security of the Corporation; (4) any short interest in any security of the Corporation (for purposes of this Bylaw a person shall be deemed to have a short interest in a security if such person directly or indirectly, through any contract, arrangement, understanding, relationship or otherwise, has the opportunity to profit or share in any profit derived from any decrease in the value of the subject security) held directly or indirectly by such stockholder and such beneficial owner; (5) any rights to dividends on the shares of the Corporation owned beneficially and of record by such stockholder and such beneficial owner that are separated or separable from the underlying shares of the Corporation; (6) any proportionate interest in shares of the Corporation or Derivative Instruments held, directly or indirectly, by a general or limited partnership in which such stockholder or such beneficial owner is a general partner or, directly or indirectly, beneficially owns an inte

- (C) any other information relating to such stockholder and beneficial owner, if any, that would be required to be disclosed in a proxy statement or other filings required to be made in connection with solicitations of proxies for, as applicable, the proposal and/or for the election of directors in a contested election pursuant to Section 14 of the 1934 Act and the rules and regulations promulgated thereunder, including without limitation:
 - (1) a description of all arrangements or understandings between the stockholder or beneficial owner and each proposed nominee and any other person or persons (including their names) pursuant to which the nomination(s) are to be made by such stockholder; and
 - (2) whether either such stockholder or beneficial owner intends to deliver a proxy statement and form of proxy to holders of, in the case of the proposal, at least the percentage of the Corporation's voting shares required under applicable law or the Certificate of Incorporation or these Bylaws to carry the proposal or, in the case of a nomination or nominations, a sufficient number of holders of the Corporation's voting shares to elect such nominee or nominees (an affirmative statement of such intent, a "Solicitation Notice").
- (e) Notwithstanding anything in Section 2.2(c) of these Bylaws to the contrary, in the event that the number of directors to be elected to the Board of Directors of the Corporation is increased and there is no public announcement naming all of the nominees for director or specifying the size of the increased Board of Directors made by the Corporation at least seventy (70) days prior to the first anniversary of the preceding year's annual meeting (or, if the annual meeting is held more than thirty (30) days before or thirty (30) days after such anniversary date, at least seventy (70) days prior to such annual meeting) a

stockholder's notice required by this Section 2.2 shall also be considered timely, but only with respect to nominees for any new positions created by such increase, if it shall be delivered to the Secretary at the principal executive offices of the Corporation not later than the close of business on the tenth (10th) day following the day on which such public announcement is first made by the Corporation.

(f) Only such persons who are nominated in accordance with the procedures set forth in this Section 2.2 shall be eligible to serve as directors and only such business shall be conducted at a meeting of stockholders as shall have been brought before the meeting in accordance with the procedures set forth in this Section 2.2. Except as otherwise provided by law, the chairman of the meeting shall have the power and duty to determine whether a nomination or any business proposed to be brought before the meeting was made, or proposed, as the case may be, in accordance with the procedures set forth in these Bylaws and, if any proposed nomination or business is not in compliance with these Bylaws, to declare that such defective proposal or nomination shall not be presented for stockholder action at the meeting and shall be disregarded.

_

- (g) Notwithstanding the foregoing provisions of this Section 2.2, in order to include information with respect to a stockholder proposal in the proxy statement and form of proxy for a stockholders' meeting, stockholders must provide notice as required by the regulations promulgated under the 1934 Act. Nothing in these Bylaws shall be deemed to affect any rights of stockholders to request inclusion of proposals in the Corporation's proxy statement pursuant to Rule 14a-8 under the 1934 Act.
- (h) For purposes of these Bylaws, "public announcement" shall mean disclosure in a press release reported by the Dow Jones News Service, Associated Press, PR Newswire, Reuters or comparable national news service or in a document publicly filed by the Corporation with the U.S. Securities and Exchange Commission pursuant to Section 13, 14 or 15(d) of the 1934 Act.

2.3 Special Meetings

- (a) Unless otherwise provided in the Certificate of Incorporation, special meetings of the stockholders of the Corporation may be called, for any purpose or purposes, only by (i) the Chairman of the Board of Directors, (ii) the Chief Executive Officer, or (iii) the Board of Directors pursuant to a resolution adopted by a majority of the directors then in office, and shall be held at such place, on such date and at such time as determined by the Board of Directors.
- (b) If a special meeting is properly called by any person or persons other than the Board of Directors, the request shall be in writing, specifying the general nature of the business proposed to be transacted, and shall be delivered personally or sent by certified or registered mail, return receipt requested, to the Secretary of the Corporation. No business may be transacted at such special meeting otherwise than specified in such notice. The Board of Directors shall determine the time and place of such special meeting, which shall be held not less than thirty-five (35) nor more than one hundred twenty (120) days after the date of the receipt of the request. Upon determination of the time and place of the meeting, the Secretary shall cause notice to be given to the stockholders entitled to vote, in accordance with the provisions of Section 2.4 of these Bylaws. Nothing contained in this subsection (b) shall be construed as limiting, fixing or affecting the time when a meeting of stockholders called by action of the Board of Directors may be held.
- (c) Nominations of persons for election to the Board of Directors may be made at a special meeting of stockholders at which directors are to be elected pursuant to the Corporation's notice of meeting (i) by or at the direction of the Board of Directors or (ii) by any stockholder of the Corporation who is a stockholder of record at the time of giving notice provided for in these Bylaws who shall be entitled to vote at the meeting and who complies with the notice procedures set forth in this Section 2.3(c). In the event the Corporation calls a special meeting of stockholders for the purpose of electing one or more directors to the Board of Directors, any such stockholder may nominate a person or persons (as the case may be), for election to such position(s) as specified in the Corporation's notice of meeting, if the stockholder's notice otherwise required by Section 2.2 of these Bylaws shall be delivered to the Secretary at the principal executive offices

5

of the Corporation not earlier than the close of business on the one hundred twentieth (120th) day prior to such special meeting and not later than the close of business on the later of the ninetieth (90th) day prior to such meeting or the tenth (10th) day following the day on which public announcement is first made of the ate of the special meeting and of the nominees proposed by the Board of Directors to be elected at such meeting. In no event shall the public announcement of an adjournment of a special meeting commence a new time period for the giving of a stockholder's notice as described above.

- (d) Unless the Certificate of Incorporation provides otherwise, any special meeting of the stockholders may be cancelled by resolution duly adopted by a majority of the directors then in office upon public notice given prior to the date previously scheduled for such meeting of stockholders.
- Notice of Meetings. Except as otherwise provided by law, notice, given in writing or by electronic transmission, of each meeting of stockholders shall be given not less than ten (10) nor more than sixty (60) days before the date of the meeting to each stockholder entitled to vote at such meeting as of the record date for determining the stockholders entitled to notice of the meeting. Such notice shall specify the place, date and hour of the meeting, the means of remote communication(s), if any, by which stockholders and proxyholders may be deemed to be present in person and vote at such meeting (as authorized by the Board of Directors entitled to notice of the meeting, and, in the case of a special meeting, the purpose of the meeting. Notice of any meeting of stockholders, if mailed, is given when deposited in the United States mail, postage prepaid, directed to the stockholder at such stockholder's address as it appears on the records of the Corporation and otherwise is given when delivered. Notice of the time, place, and purpose of any meeting of stockholders may be waived in writing, signed by the person entitled to notice thereof, or by electronic transmission by such person, either before or after such meeting, and will be waived by any stockholder by his or her attendance thereat in person or by proxy, except when the stockholder at meeting is not lawfully called or convened. Any stockholder so waiving notice of such meeting is hot be unascated at, nor the purpose of, any annual or special meeting of the tockholders need be specified in any written waiver of notice or any waiver by electronic transmission.

2.5 <u>Determination of Stockholders of Record.</u>

(a) In order that the Corporation may determine the stockholders entitled to notice of any meeting of stockholders or any adjournment thereof, the Board of Directors may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted by the Board of Directors, and which record date shall not be more than sixty (60) nor less than ten (10) days before the date of such meeting. If the Board of Directors so fixes a date, such date shall also be the record date for determining the stockholders entitled to vote at such meeting unless the Board of Directors determines, at the time it fixes such record date, that a later date on or before

6

the date of the meeting shall be the date for making such determination. If no record date is fixed by the Board of Directors, the record date for determining stockholders entitled to notice of or to vote at a meeting of stockholders shall be at the close of business on the day next preceding the day on which notice is given, or, if notice is waived, at the close of business on the day next preceding the day on which the meeting is held. A determination of stockholders of record entitled to notice of or to vote at a meeting of stockholders shall apply to any adjournment of the meeting; provided, however, that the Board of Directors may fix a new record date for determination of stockholders entitled to vote at the adjourned meeting the same or an earlier date as that fixed for determination of stockholders entitled to vote at the adjourned meeting.

- (b) In order that the Corporation may determine the stockholders entitled to consent to corporate action in writing without a meeting, the Board of Directors may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted by the Board of Directors, and which date shall not be more than ten (10) days after the date upon which the resolution fixing the record date is adopted by the Board of Directors. If no record date has been fixed by the Board of Directors, the record date for determining stockholders entitled to consent to corporate action in writing without a meeting, when no prior action by the Board of Directors is required by the DGCL, shall be the first date on which a signed written consent setting forth the action taken or proposed to be taken is delivered to the Corporation by delivery to its registered office in the State of Delaware, its principal place of business or an officer or agent of the Corporation having custody of the book in which proceedings of meetings of stockholders are recorded. Delivery made to the Corporation's registered office shall be by hand or by certified or registered mail, return receipt requested. If no record date has been fixed by the Board of Directors and prior action by the Board of Directors is required by the DGCL, the record date for determining stockholders entitled to consent to corporate action in writing without a meeting shall be at the close of business on the day on which the Board of Directors adopts the resolution taking such prior action.
- 2.6 Quorum. At all meetings of the stockholders, except where otherwise provided by law, the Certificate of Incorporation or these Bylaws, the presence, in person or by proxy duly authorized, of the holders of a majority of the outstanding shares of stock entitled to vote shall constitute a quorum for the transaction of business. In the absence of a quorum, any meeting of stockholders may be adjourned, from time to time, either by the chairman of the meeting or by vote of the holders of a majority of the shares represented thereat, but no other business shall be transacted at such meeting. The stockholders present at a duly called or convened meeting, at which a quorum is present, may continue to transact business until adjournment, notwithstanding the withdrawal of enough stockholders to leave less than a quorum. Where a separate vote by a class or series or classes or series is required, except where otherwise provided by law or by the Certificate of Incorporation or these Bylaws, a majority of the outstanding shares of such class or series or classes or series, present in person or represented by proxy duly authorized, shall constitute a quorum entitled to take action with respect to that vote on that matter.

7

Adjournment and Notice of Adjourned Meetings. Any meeting of stockholders, whether annual or special, may be adjourned from time to time either by the chairman of the meeting or by the vote of a majority of the shares present in person or represented by proxy at the meeting. When a meeting is adjourned to another time or place, notice need not be given of the adjourned meeting if the time and place thereof, and the means of remote communication(s), if any, by which stockholders and proxyholders may be deemed to be present in person and vote at such adjourned meeting (as authorized by the Board of Directors in its sole discretion pursuant to Section 211(a)(2) of the DGCL), are announced at the meeting at which the adjournment is taken. At the adjourned meeting, the Corporation may transact any business which might have been transacted at the original meeting. If the adjournment is for more than thirty (30) days, a notice of the adjourned meeting shall be given to each stockholder of record entitled to vote at the meeting. If after the adjournment a new record date for stockholders entitled to vote is fixed for the adjourned meeting to each stockholder of record entitled to vote at such adjourned meeting as of the record date fixed for notice of such adjourned meeting.

- 2.8 <u>Voting.</u> Except as otherwise provided by law or by the Certificate of Incorporation or these Bylaws, in all matters other than the election of directors, the affirmative vote of the majority of shares present in person or represented by proxy at the meeting and entitled to vote generally on the subject matter shall be the act of the stockholders. Except as otherwise provided by law, the Certificate of Incorporation or these Bylaws, directors shall be elected by a plurality of the votes of the shares present in person or represented by proxy at the meeting and entitled to vote generally on the election of directors. Where a separate vote by a class or series or classes or series is required, except where otherwise provided by law or by the Certificate of Incorporation or these Bylaws, the affirmative vote of the majority (plurality, in the case of the election of directors) of shares of such class or series or classes or series present in person or represented by proxy at the meeting shall be the act of such class or series or classes or series.
- 2.9 <u>Voting Rights</u>. For the purpose of determining those stockholders entitled to vote at any meeting of the stockholders, except as otherwise provided by law, only persons in whose names shares stand on the stock records of the Corporation on the record date, as provided in Section 6.5 of these Bylaws, shall be entitled to vote at any meeting of stockholders. Every person entitled to vote shall have the right to do so either in person or by an agent or agents authorized by a proxy granted in accordance with the DGCL. An agent so appointed need not be a stockholder. No proxy shall be voted after three (3) years from its date of creation unless the proxy provides for a longer period.
- 2.10 <u>Joint Owners of Stock</u>. If shares or other securities having voting power stand of record in the names of two (2) or more persons, whether fiduciaries, members of a partnership, joint tenants, in common, tenants by the entirety, or otherwise, or if two (2) or more persons have the same fiduciary relationship respecting the same shares, unless the Secretary is given written notice to the contrary and is furnished with a copy of the instrument or order appointing them or creating the relationship wherein it is so provided, their acts with respect to voting shall have the following effect: (a) if only one (1) votes, his or her act binds all; (b) if more than one (1) votes, the act of the majority so voting binds all; (c) if more than one (1) votes,

but the vote is evenly split on any particular matter, each faction may vote the securities in question proportionally, or may apply to the Delaware Court of Chancery for relief as provided in Section 217(b) of the DGCL. If the instrument filed with the Secretary shows that any such tenancy is held in unequal interests, a majority or even-split for the purpose of clauses (b) and (c) shall be a majority or even-split in interest.

List of Stockholders. The officer who has charge of the stock ledger of the Corporation shall prepare and make, at least ten (10) days before every meeting of stockholders, a complete list of the stockholders entitled to vote at said meeting (provided, however, that if the record date for determining the stockholders entitled to vote is less than ten (10) days before the date of the meeting, the list shall reflect the stockholders entitled to vote as of the tenth (10th) day before the meeting, date), arranged in alphabetical order, showing the address of each stockholder and the number of shares registered in the name of each stockholder. Nothing contained in this Section 2.11 shall require the Corporation to include electronic mail addresses or other electronic contact information on such list. Such list shall be open to the examination of any stockholder, for any purpose germane to the meeting for a period of at least ten (10) days prior to the meeting, (a) on a reasonably accessible electronic network, provided that the information required to gain access to such list is provided with the notice of the meeting, or (b) during ordinary business hours, at the principal place of business of the Corporation. In the event the Corporation determines to make the list available on an electronic network, the Corporation may take reasonable steps to ensure that such information is available only to stockholders of the Corporation. If the meeting is to be held at a place, then a list of stockholders entitled to vote at the meeting shall be produced and kept at the time and place of the meeting during the whole time thereof and may be examined by any stockholder of the Corporation who is present. If the meeting is to be held solely by means of remote communication, then such list shall also be open to the examination of any stockholder of the Corporation during the whole time of the meeting on a reasonably accessible electronic network, and the information required to access such list shall be provided with the notice of

2.12 Action Without Meeting

- (a) Any action to be taken at any annual or special meeting of stockholders may be taken without a meeting, without prior notice and without a vote, if a consent or consents in writing, setting forth the action to be so taken, shall be signed by the holders of outstanding stock having not less than the minimum number of votes that would be necessary to authorize or take such action at a meeting at which all shares entitled to vote thereon were present and voted and shall be delivered (by hand or by certified or registered mail, return receipt requested) to the Corporation by delivery to its registered office in the State of Delaware, its principal place of business or an officer or agent of the Corporation having custody of the book in which proceedings of meetings of the stockholders are recorded.
- (b) Every written consent shall bear the date of signature of each stockholder who signs the consent, and no written consent shall be effective to take the corporate action referred to therein unless, within sixty (60) days of the earliest dated consent delivered in the manner required by this Section 2.11, written consents signed by a sufficient number of holders to take action are delivered to the Corporation as aforesaid.

9

Delivery made to the Corporation's registered office shall be by hand or by certified or registered mail, return receipt requested. Any person executing a consent may provide, whether through instruction to an agent or otherwise, that such a consent will be effective at a future time (including a time determined upon the happening of an event), no later than 60 days after such instruction is given or such provision is made, and, if evidence of such instruction or provision is provided to the Corporation, such later effective time shall serve as the date of signature. Unless otherwise provided, any such consent shall be revocable prior to its becoming effective.

- (c) Any copy, facsimile or other reliable reproduction of a consent in writing may be substituted or used in lieu of the original writing for any and all purposes for which the original writing could be used, provided that such copy, facsimile or other reproduction shall be a complete reproduction of the entire original writing.
- (d) Prompt notice of the taking of the corporate action without a meeting by less than unanimous written consent shall, to the extent required by applicable law, be given to those stockholders who have not consented in writing, and who, if the action had been taken at a meeting, would have been entitled to notice of the meeting if the record date for notice of such meeting had been the date that written consents signed by a sufficient number of holders to take the action were delivered to the Corporation.

2.13 Organization.

- (a) At every meeting of stockholders, (i) the Chairman of the Board of Directors or, if a Chairman of the Board of Directors has not been appointed or is absent, (ii) the Vice Chairman of the Board of Directors, if any, or, if the Vice Chairman of the Board of Directors is absent or there is no Vice Chairman of the Board of Directors, (iii) the Chief Executive Officer or, if the Chief Executive Officer is absent, (iv) such person as the Chairman of the Board of Directors shall appoint or, if such Chairman has not been appointed, (v) any officer of the Corporation chosen by the Board of Directors, shall act as chairman of the meeting. The Secretary, or in the absence of the Secretary an Assistant Secretary the chairman of the meeting may appoint any person to act as secretary of the meeting.
- (b) The Board of Directors, in advance of any meeting of stockholders, may, and shall if required by law, appoint one (1) or more inspector(s), who may include individual(s) who serve the Corporation in other capacities, including without limitation as officers, employees or agents, to act at the meeting of stockholders and make a written report thereof. The Board of Directors may designate one (1) or more persons as alternate inspector(s) to replace any inspector who fails to act. If no inspector alternate has been appointed or is able to act at a meeting of stockholders, the chairman of the meeting shall appoint one (1) or more inspector(s) to act at the meeting. Each inspector, before discharging his or her duties, shall take and sign an oath to faithfully execute the duties of inspector with strict impartiality and according to the best of his or her ability. The inspector(s) or alternate(s) shall have the duties prescribed pursuant to Section 231 of the DGCL or other applicable law.

10

- (c) The Board of Directors of the Corporation shall be entitled to make such rules or regulations for the conduct of meetings of the stockholders as it shall deem necessary, appropriate or convenient. Subject to such rules and regulations of the Board of Directors, if any, the chairman of the meeting shall have the right and authority to prescribe such rules, regulations and procedures and to do all such acts as, in the judgment of such chairman, are necessary, appropriate or convenient for the proper conduct of the meeting, including, without limitation, establishing an agenda or order of business for the meeting, rules and procedures for maintaining order at the meeting and the safety of those present, limitations on participation in such meeting to stockholders of record of the Corporation and their duly authorized and constituted proxies and such other persons as the chairman shall permit, restrictions on entry to the meeting after the time fixed for the commencement thereof, limitations on the time allotted to questions or comments by participants and regulation of the opening and closing of the polls for each matter upon which the stockholders will vote at the meeting shall be announced at the meeting. Unless and to the extent determined by the Board of Directors or the chairman of the meeting, meetings of the stockholders shall not be required to be held in accordance with rules of parliamentary procedure.
- 2.14 <u>Postponement and Cancellation of Meeting</u>. Any previously scheduled annual or special meeting of stockholders may be postponed, and any previously scheduled annual or special meeting of stockholders may be cancelled, by resolution of the Board of Directors upon public notice given prior to the time previously scheduled for such meeting.

ARTICLE III DIRECTORS

- 3.1 General Powers. The business and affairs of the Corporation shall be managed by or under the direction of the Board of Directors, except as may be otherwise provided in the DGCL or in the Certificate of Incorporation. The Board of Directors may adopt such rules and procedures, not inconsistent with the Certificate of Incorporation, these Bylaws or applicable law, as it may deem proper for the conduct of its meetings and the management of the Corporation, except as may be otherwise provided by the DGCL or by the Certificate of Incorporation.
- 3.2 Number, Term of Office and Qualifications. The Board of Directors shall consist of one or more members, the number thereof to be determined from time to time by resolution of the Board of Directors. Each director shall hold office until a successor is duly elected and qualified or until the director's earlier death, resignation, disqualification or removal. Directors shall be natural persons, but need not be stockholders of the Corporation unless otherwise required by the Certificate of Incorporation.
- 3.3 <u>Vacancies</u>. Unless otherwise provided in the Certificate of Incorporation and subject to the rights of the holders of any series of preferred stock then outstanding, any vacancies on the Board of Directors resulting from death, resignation, disqualification, removal or other causes and any newly created directorships resulting from any increase in the number of directors shall, unless the Board of Directors determines by

newly created directorships shall be filled by stockholders, be filled only by the affirmative vote of a majority of the directors then in office, even if less than a quorum of the Board of Directors. Any director elected in accordance with the preceding sentence shall hold office for the remainder of the full term of the director for which the vacancy was created or occurred and until such director's successor shall have been elected and qualified.

- 3.4 Resignation. Any director may resign at any time by delivering his or her notice in writing or by electronic transmission to the Secretary, such resignation to specify whether it will be effective at a particular time, upon receipt by the Secretary or at the pleasure of the Board of Directors. If no such specification is made, it shall be deemed effective at the pleasure of the Board of Directors. Verbal resignation shall not be deemed effective until confirmed by the director in writing or by electronic transmission to the Secretary. When one or more directors shall resign from the Board of Directors effective at a future date, a majority of the directors then in office, including those who have so resigned, shall have the power to fill such vacancy or vacancies, the vote thereon to take effect when such resignation or resignations shall become effective, and each director so chosen shall hold office for the unexpired portion of the term of the director whose place shall be vacated and until his or her successor shall have been duly elected and qualified.
- 3.5 Removal. Subject to the Certificate of Incorporation, any director or the entire Board of Directors may be removed, with or without cause, by the affirmative vote of the holders of a majority of the voting power of all of then outstanding shares of capital stock of the Corporation then entitled to vote in the election of directors, voting together as a single class. No reduction of the authorized number of directors shall have the effect of removing any director prior to the expiration of such director's term of office.

3.6 Meetings

- (a) Unless otherwise restricted by the Certificate of Incorporation, regular meetings of the Board of Directors may be held at any time or date and at any place within or without the State of Delaware which has been designated by the Board of Directors and publicized among all directors, either orally or in writing, by telephone, including a voice-messaging system or other system designed to record and communicate messages, facsimile, telegraph or telex, or by electronic mail or other electronic means. No further notice shall be required for regular meetings of the Board of Directors.
- (b) Unless otherwise restricted by the Certificate of Incorporation, special meetings of the Board of Directors may be held at any time and place within or without the State of Delaware whenever called by the Chairman of the Board of Directors, the Chief Executive Officer, or a majority of the directors then in office.
- (c) Any member of the Board of Directors, or of any committee thereof, may participate in a meeting by means of conference telephone or other communications equipment pursuant to which all persons participating in the meeting can hear each other, and participation in a meeting by such means shall constitute presence in person at such meeting.

12

- (d) Notice of the time and place of all special meetings of the Board of Directors shall be given to each director (i) by giving notice to such director in person or by telephone, including a voice messaging system or other system designed to record and communicate messages, during normal business hours, at least twenty-four (24) hours before the meeting, (ii) by sending a telegram or delivering notice by facsimile transmission, by electronic mail or by hand, to such director at his or her last known business or home address, during normal business hours, at least twenty-four (24) hours before the meeting, or (iii) by mailing notice, via first class United States mail, to such director at his or her last known business or home address at least three (3) days in advance of the meeting. Notice of any meeting may be waived in writing, or by electronic transmission, at any time before or after the meeting and will be waived by any director by attendance thereat, except when the director attends the meeting for the express purpose of objecting, at the beginning of the meeting, to the transaction of any business because the meeting is not lawfully called or convened. Notice of a special meeting of the Board of Directors need not specify the purpose of the meeting.
- (e) The transaction of all business at any meeting of the Board of Directors, or any committee thereof, however called or noticed, or wherever held, shall be as valid as though had at a meeting duly held after regular call and notice, if a quorum be present and if, either before or after the meeting, each of the directors not present who did not receive notice shall sign a written waiver of notice or shall waive notice by electronic transmission. All such waivers shall be filed with the corporate records or made a part of the minutes of the meeting. Neither the business to be transacted at, nor the purpose of, any regular or special meeting of the Board of Directors need be specified in any written waiver of notice or any waiver by electronic transmission.

3.7 Quorum and Voting

- (a) Unless the Certificate of Incorporation requires a greater number, a quorum of the Board of Directors shall consist of a majority of the directors then in office. At any meeting whether a quorum be present or otherwise, a majority of the directors present may adjourn from time to time until the time fixed for the next regular meeting of the Board of Directors, without notice other than by announcement at the meeting.
- (b) At each meeting of the Board of Directors at which a quorum is present, all questions and business shall be determined by the affirmative vote of a majority of the directors present at the meeting, unless a different vote be required by law, the Certificate of Incorporation or these Bylaws.
- 3.8 Action Without Meeting. Unless otherwise restricted by the Certificate of Incorporation or these Bylaws, any action required or permitted to be taken at any meeting of the Board of Directors, or of any committee thereof, may be taken without a meeting, if all members of the Board of Directors or committee, as the case may be, consent thereto in writing or by electronic transmission, and such writing or writings or transmissions are filed with the minutes of proceedings of the Board of Directors or committee. Such filing shall be in paper form if the minutes are maintained in paper form and shall be in electronic form if the

13

minutes are maintained in electronic form. Any person (whether or not then a director) may provide, whether through instruction to an agent or otherwise, that a consent to action will be effective at a future time (including a time determined upon the happening of an event), no later than 60 days after such instruction is given or such provision is made and such consent shall be deemed to have been given at such effective time so long as such person is then a director and did not revoke the consent prior to such time. Any such consent shall be revocable prior to its becoming effective.

3.9 <u>Fees and Compensation</u>. Directors shall be entitled to such compensation for their services as may be approved by the Board of Directors, or any committee thereof, including, if so approved by resolution of the Board of Directors or such committee, a fixed sum and expenses of attendance, if any, for attendance at each regular or special meeting of the Board of Directors and at any meeting of a committee of the Board of Directors. Nothing herein contained shall be construed to preclude any director from serving the Corporation in any other capacity as an officer, agent, employee, or otherwise and receiving compensation therefor.

3.10 <u>Committees</u>

- (a) The Board of Directors may, from time to time, appoint such committees as may be permitted by law. Such committees appointed by the Board of Directors shall consist of one (1) or more members of the Board of Directors and shall have such powers and perform such duties as may be prescribed by the resolution or resolutions creating such committees, but no committee shall have the power or authority in reference to (i) approving or adopting, or recommending to the stockholders, any action or matter expressly required by the DGCL to be submitted to stockholders for approval, or (ii) adopting, amending or repealing any provision of these Bylaws.
- (b) The Board of Directors, subject to any requirements of any outstanding series of preferred stock and the provisions of subsections (a) and (b) of this Section 3.10, may at any time increase or decrease the number of members of a committee or terminate the existence of a committee. The membership of a committee member shall terminate on the date of his or her death or voluntary resignation from the committee or from the Board of Directors. The Board of Directors may at any time for any reason remove any individual committee member and the Board of Directors may fill any committee vacancy created by death, resignation, removal or increase in the number of members of the committee. The Board of Directors may designate one or more directors as alternate members of any committee, who may replace any absent or disqualified member at any meeting of the committee, and, in addition, in the absence or disqualification of any member of a committee, the member or members thereof present at any meeting and not disqualified from voting, whether or not he, she or they constitute a quorum, may unanimously appoint another member of the Board of Directors to act at the meeting in the place of any such absent or disqualified member.
- (c) Unless the Board of Directors shall otherwise provide, regular meetings of any committee appointed pursuant to this Section 3.10 shall be held at such times and places as are determined by the Board of Directors, or by any such committee, and when notice thereof has been given to each member of such committee, no further notice of

3.11 Organization. At every meeting of the directors, the Chairman of the Board of Directors, or, if a Chairman of the Board of Directors has not been appointed or is absent, the Vice Chairman of the Board of Directors, or if a Vice Chairman of the Board of Directors has not been appointed or is absent, the Chief Executive Officer (if a director), or if the Chief Executive Officer is absent, or, in the absence of any such person, a chairman of the meeting chosen by a majority of the directors present, shall preside over the meeting, or in the absence of the Secretary and Assistant Secretary, shall act as secretary of the meeting, but in the absence of the Secretary and any Assistant Secretary the chairman of the meeting may appoint any person to act as secretary of the meeting.

ARTICLE IV

- 4.1 <u>Positions; Election and Qualification.</u> The officers of the Corporation shall be elected annually by the Board of Directors and shall include a President or Chief Executive Officer ("<u>Chief Executive Officer"</u>), a Chief Financial Officer, a Treasurer and a Secretary. The Board of Directors, in its discretion, may also elect a Chairman (who must be a director), one or more Vice Chairman (who must be directors) and one or more Vice Presidents, Assistant Treasurers, Assistant Secretaries and other officers. The Board of Directors may assign such additional titles to one or more of the officers as it shall deem appropriate. Any two or more offices may be held by the same person. No officer need be a stockholder.
- 4.2 Term. Each officer of the Corporation shall hold office at the pleasure of the Board of Directors and until such officer's successor is elected and qualified or until such officer's earlier death, resignation or removal, subject to the rights, if any, of an officer under contract of employment. If the office of any officer becomes vacant for any reason, the vacancy may be filled by the Board of Directors. The election or appointment of an officer shall not of itself create contract rights.

15

4.3 Duties

- (a) The Chairman of the Board of Directors shall, if present, preside at meetings of the Board of Directors and stockholders and exercise and perform such other powers and duties as may from time to time be assigned to him or her by the Board of Directors or as may be prescribed by these Bylaws.
- (b) Subject to such supervisory powers, if any, as the Board of Directors may give to the Chairman of the Board of Directors, the Chief Executive Officer shall, subject to the control of the Board of Directors, have general supervision, direction and control of the business and affairs of the Corporation and shall report directly to the Board of Directors. All other officers, officials, employees and agents shall report directly or indirectly to the Chief Executive Officer. The Chief Executive Officer shall see that all orders and resolutions of the Board of Directors are carried into effect.
- (c) In the absence or disability of the Chief Executive Officer, the Vice President(s), if any, in order of their rank as fixed by the Board of Directors or, if not ranked, a Vice President designated by the Board of Directors, shall perform all the duties of the Chief Executive Officer and, when so acting, shall have all the powers of, and be subject to all the restrictions upon, the Chief Executive Officer. The Vice President(s) shall have such other powers and perform such other duties as form time to time may be prescribed for them respectively by the Board of Directors, these Bylaws, the Chairman of the Board of Directors, the Chief Executive Officer. The Board of Directors may designate one or more Executive Vice Presidents or Senior Vice Presidents or may otherwise specify the order of seniority of the Vice Presidents
- (d) The Chief Financial Officer shall keep and maintain, or cause to be kept and maintained, adequate and correct books and records of accounts of the properties and business transactions of the Corporation, including accounts of its assets, liabilities, receipts, disbursements, gains, losses, capital and retained earnings. The Chief Financial Officer shall deposit all money and other valuables in the name and to the credit of the Corporation with such depositaries as may be designated by the Board of Directors or Chief Executive Officer. The Chief Financial Officer shall disburse the funds of the Corporation as may be ordered by the Board of Directors, shall render to the Board of Directors and Chief Executive Officer, whenever they request, an account of all of his or her transactions as Chief Financial Officer and of the financial condition of the Corporation, and shall have such other powers and perform such other duties as may be prescribed by the Board of Directors or these Bylaws. In lieu of any contrary resolution duly adopted by the Board of Directors, the Chief Financial Officer shall also be the Treasurer of the Corporation.
- (e) The Secretary shall keep or cause to be kept, at the principal executive office of the Corporation, or such other place as the Board of Directors may direct, a book of minutes of all meetings and actions of directors, committees of directors, and stockholders. The minutes shall show the time and place of each meeting, whether regular or special (and, if special, how authorized and the notice given), the names of those present at directors' meetings or committee meetings, the number of shares present or represented at stockholders' meetings, and the proceedings thereof.

16

The Secretary shall keep, or cause to be kept, at the principal executive office of the Corporation or at the office of the Corporation's transfer agent or registrar, as determined by resolution of the Board of Directors, a share register, or a duplicate share register, showing the names of all stockholders and their addresses, the number and classes of shares held by each, the number and date of certificates evidencing such shares, and the number and date of cancellation of every certificate surrendered for cancellation.

The Secretary shall give, or cause to be given, notice of all meetings of the stockholders, the Board of Directors and any committee(s) of the Board of Directors, required to be given by law or by these Bylaws.

The Secretary shall keep the seal of the Corporation, if one be adopted, in safe custody and shall have such other powers and perform such other duties as may be prescribed by the Board of Directors or by these Bylaws.

- (f) The Vice Chairman of the Board, if any, shall, in the absence of the Chairman of the Board, or in the event of his or her inability or refusal to act, perform the duties and exercise the powers of the Chairman of the Board and shall perform such other duties and have such other powers as the Board of Directors may from time to time prescribe.
- (g) The Assistant Secretary(ies), if any, in the order determined by the Board of Directors (or if there be no such determination, then in the order of their election) shall, in the absence of the Secretary or in the event of his or her inability or refusal to act, perform the duties and exercise the powers of the Secretary and shall perform such other duties and have such other powers as the Board of Directors may from time to time prescribe.
- (h) The Assistant Treasurer(s), if any, in the order determined by the Board of Directors (or if there be no such determination, then in the order of their election), shall, in the absence of the Chief Financial Officer or in the event of his or her inability or refusal to act, perform the duties and exercise the powers of the Chief Financial Officer and shall perform such other duties and have such other powers as the Board of Directors may from time to time prescribe.
- 4.4 <u>Delegation of Authority.</u> The Board of Directors may from time to time delegate the powers or duties of any officer to any other officer or agent, notwithstanding any provision hereof.
- 4.5 Resignations. Any officer may resign at any time by giving notice in writing or by electronic transmission to the Chairman of the Board, or to the Chief Executive Officer or to the Secretary. Verbal resignation shall not be deemed effective until confirmed by the officer in writing or by electronic transmission to the Chairman of the Board, Chief Executive Officer or Secretary. Any such resignation shall be effective when received by the person or persons to whom such notice is given, unless a later time is specified therein, in which event the resignation shall become effective at such later time. Unless otherwise specified in such notice, the acceptance of any such resignation shall not be necessary to make it effective. Any resignation shall be without prejudice to the rights, if any, of the Corporation under any contract with the resigning officer.

17

4.6 Removal. Subject to the rights, if any, of an officer under contract of employment, any officer may be removed from office at any time, either with or without cause, by the affirmative vote of a majority of the directors in office at the time, or by any committee or superior officers upon whom such power of removal may have been conferred by the Board of Directors.

ARTICLE V EXECUTION OF CORPORATE INSTRUMENTS AND VOTING OF SECURITIES OWNED BY THE CORPORATION

5.1 <u>Execution of Corporate Instruments.</u> The Board of Directors may, in its discretion, determine the method and designate the signatory officer or officers, or other persons, to execute on behalf of the Corporation any corporate instrument or document, or to sign on behalf of the Corporation the corporate name without limitation, or to enter into contracts on behalf of the Corporation, except where otherwise provided by law or these Bylaws, and such execution or signature shall be binding upon the Corporation.

All checks and drafts drawn on banks or other depositaries on funds to the credit of the Corporation or in special accounts of the Corporation shall be signed by such person or persons as the Board of Directors shall authorize so to do.

Unless authorized or ratified by the Board of Directors or within the agency power of an officer, no officer, agent or employee shall have any power or authority to bind the Corporation by any contract or engagement or to pledge its credit or to render it liable for any purpose or for any amount.

5.2 <u>Voting of Securities Owned by the Corporation</u>. All stock and other securities of other Corporations owned or held by the Corporation for itself, or for other parties in any capacity, shall be voted, and all proxies with respect thereto shall be executed, by the person authorized so to do by resolution of the Board of Directors, or, in the absence of such authorization, by the Chairman of the Board of Directors, the Chief Executive Officer or any Vice President.

6.1 <u>Form and Execution of Certificates</u>. Shares of stock of the Corporation shall be represented by certificates, or shall be uncertificated. Certificates for the shares of stock of the Corporation, if any, shall be in such form as is consistent with the Certificate of Incorporation and applicable law. Every holder of stock represented by certificate shall be entitled to have a certificate signed by or in the name of the Corporation by the Chairman of the Board of Directors, or the Chief Executive Officer or any Vice President and by the Treasurer or Assistant Treasurer or the Secretary or Assistant Secretary, certifying the number of shares owned by him or her in the Corporation. Any or all of the signatures on the certificate may be facsimiles. In case any officer, transfer agent, or registrar who has signed or whose facsimile signature has been placed upon a certificate shall have ceased to be such officer, transfer agent, or registrar

18

before such certificate is issued, it may be issued with the same effect as if he or she were such officer, transfer agent, or registrar at the date of issue.

- 6.2 Transfers of Stock. Transfers of record of shares of stock of the Corporation shall be made only upon its books by the holders thereof, in person or by attorney duly authorized, and, in the case of stock represented by certificate, upon the surrender of a properly endorsed certificate or certificates for a like number of shares. To the extent designated by the Chief Executive Officer or any Vice President or the Treasurer of the Corporation, the Corporation may recognize the transfer of fractional uncertificated shares, but shall not otherwise be required to recognize the transfer of fractional shares. The Corporation shall have power to enter into and perform any agreement with any number of stockholders of any one or more classes of stock of the Corporation to restrict the transfer of shares of stock of the Corporation of any one or more classes owned by such stockholders in any manner not prohibited by the DGCL.
 - 6.3 Transfer Agents and Registrars. The Board of Directors may appoint, or authorize any officer or officers to appoint, one or more transfer agents and one or more registrars.
- 6.4 Lost, Stolen or Destroyed Certificates. A new certificates or uncertificates or uncertificated shares shall be issued in place of any certificate or certificates theretofore issued by the Corporation alleged to have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the person claiming the certificate of stock to be lost, stolen or destroyed. The Corporation may require, as a condition precedent to the issuance of a new certificate or certificates or uncertificated shares, the owner of such lost, stolen or destroyed certificates, or the owner's legal representative, to agree to indemnify the Corporation in such manner as it shall require or to give the Corporation a surety bond in such form and amount as it may direct as indemnity against any claim that may be made against the Corporation with respect to the certificate alleged to have been lost, stolen or destroyed.

6.5 Fixing Record Dates

(a) In order that the Corporation may determine the stockholders entitled to notice of or to vote at any meeting of stockholders or any adjournment thereof, the Board of Directors may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted by the Board of Directors, and which record date shall not be more than sixty (60) nor less than ten (10) days before the date of such meeting. If the Board of Directors so fixes a date, such date shall also be the record date for determining the stockholders entitled to vote at such meeting unless the Board of Directors determines, at the time it fixes such record date, that a later date on or before the date of the meeting shall be the date for making such determination. If no record date is fixed by the Board of Directors, the record date for determining stockholders entitled to notice of or to vote at a meeting of stockholders shall be at the close of business on the day next preceding the day on which notice is given, or, if notice is waived, at the close of business on the day next preceding the day on which notice is given, or, if notice is waived, at the close of business on the day next preceding the day on which notice is given, or, if notice is waived, at the close of business on the day next preceding the day on which notice is given, or, if notice is waived, at the close of business on the day next preceding the day on which notice is given, or, if notice is waived, at the close of business on the day next preceding the day on which notice is given, or, if notice is waived, at the close of business on the day next preceding the day on which notice is given, or, if notice is waived, at the close of business on the day next preceding the day on which notice is given, or, if notice is waived, at the close of business on the day next preceding the day on which notice is given, or, if notice is waived, at the close of business on the day next preceding the day on which notice is given, or, if notice i

19

stockholders entitled to vote at the adjourned meeting and in such case shall also fix as the record date for stockholders entitled to notice of such adjourned meeting the same or an earlier date as that fixed for the determination of stockholders entitled to vote therewith at the adjourned meeting.

- (b) In order that the Corporation may determine the stockholders entitled to consent to corporate action in writing without a meeting, the Board of Directors may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted by the Board of Directors, and which record date shall not be more than ten (10) days after the date upon which the resolution fixing the record date is adopted by the Board of Directors. If no record date has been fixed by the Board of Directors, the record date for determining stockholders entitled to consent to corporate action in writing without a meeting: (i) when no prior action by the Board of Directors is required by law, the record date for such purpose shall be the first date on which a signed written consent setting forth the action taken or proposed to be taken is delivered to the Corporation by delivery (by hand, or by certified or registered mail, return receipt requested) to its registered office in the State of Delaware, its principal place of business, or an officer or agent of the Corporation having custody of the book in which proceedings of meetings of the stockholders are recorded and (ii) if prior action by the Board of Directors is required by law, the record date for such purpose shall be at the close of business on the day on which the Board of Directors adopts the resolution taking such prior action.
- (c) In order that the Corporation may determine the stockholders entitled to receive payment of any dividend or other distribution or allotment of any rights or the stockholders entitled to exercise any rights in respect of any change, conversion or exchange of stock, or for the purpose of any other lawful action, the Board of Directors may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted, and which record date shall be not more than 60 days prior to such action. If no record date is fixed, the record date for determining stockholders for any such purpose shall be at the close of business on the day on which the Board of Directors adopts the resolution relating thereto.
- Registered Stockholders. The Corporation shall be entitled to recognize the exclusive right of a person registered on its books as the owner of shares to receive dividends, and to vote as such owner, and shall not be bound to recognize any equitable or other claim to or interest in such share or shares on the part of any other person whether or not it shall have express or other notice thereof, except as otherwise provided by applicable law.

ARTICLE VII OTHER SECURITIES OF THE CORPORATION

7.1 <u>Execution of Other Securities</u>. All bonds, debentures and other corporate securities of the Corporation, other than stock certificates (covered in Section 6.1), may be signed by the Chairman of the Board of Directors, the Chief Executive Officer, the President or any Vice President, or such other person as may be authorized by the Board of Directors, and the corporate seal, if any, may be impressed thereon or a facsimile of such seal imprinted thereon

20

and attested by the signature of the Secretary or an Assistant Secretary, or the Chief Financial Officer or Treasurer or an Assistant Treasurer; provided, however, that where any such bond, debenture or other corporate security shall be authenticated by the manual signature, or where permissible facisimile signature, of a trustee under an indenture pursuant to which such bond, debenture or other corporate security shall be issued, the signatures of the persons signing and, if applicable, attesting the corporate seal on such bond, debenture or other corporate security may be the imprinted facsimile of the signatures of such persons. Interest coupons appertaining to any such bond, debenture or other corporate security, authenticated by a trustee as aforesaid, shall be signed by the Treasurer or an Assistant Treasurer of the Corporation or such other person as may be authorized by the Board of Directors, or bear imprinted thereon the facsimile signature of such person. In case any officer who shall have signed or attested any bond, debenture or other corporate security, or whose facsimile signature shall appear thereon or on any such interest coupon, shall have ceased to be such officer before the bond, debenture or other corporate security so signed or attested shall have been delivered, such bond, debenture or other corporate security so signed or attested shall have been used thereon had not ceased to be such officer of the Corporation.

ARTICLE VIII DIVIDENDS

- 8.1 <u>Declaration of Dividends.</u> Dividends upon the capital stock of the Corporation, subject to the provisions of the Certificate of Incorporation and applicable law, if any, may be declared by the Board of Directors pursuant to law at any regular or special meeting. Dividends may be paid in cash, in property, or in shares of the capital stock, subject to the provisions of the Certificate of Incorporation and applicable law.
- 8.2 <u>Dividend Reserve</u>. Before payment of any dividend, there may be set aside out of any funds of the Corporation available for dividends such sum or sums as the Board of Directors from time to time, in their absolute discretion, think proper as a reserve or reserves to meet contingencies, or for equalizing dividends, or for repairing or maintaining any property of the Corporation, or for such other purpose as the Board of Directors shall think conducive to the interests of the Corporation, and the Board of Directors may modify or abolish any such reserve in the manner in which it was created.

ARTICLE IX GENERAL PROVISIONS

- 9.1 Fiscal Year. The fiscal year of the Corporation shall be fixed from time to time by resolution of the Board of Directors.
- 9.2 Corporate Seal. The Corporation may, but need not, have a corporate seal. In the event the Corporation has a seal, the seal need not be affixed for any contract, resolution or other document executed by or on behalf of the Corporation to be valid and duly authorized.

- (a) Written notice to stockholders of stockholder meetings shall be given as provided in Section 2.4 herein. Without limiting the manner by which notice may otherwise be given effectively to stockholders under any agreement or contract with such stockholder, and except as otherwise required by law, written notice to stockholders for purposes other than stockholder meetings may be sent by United States mail or nationally recognized overnight courier, or by facsimile, telegraph or telex or by electronic mail or other electronic means.
- (b) Notice to directors of special meetings shall be given as provided in Section 3.7(d) herein. Subject to the preceding sentence and except as expressly stated otherwise herein, notice may otherwise be given by the methods stated in subsection (a) above.
- (c) An affidavit of mailing, executed by a duly authorized and competent employee of the Corporation or its transfer agent appointed with respect to the class of stock affected, or other agent, specifying the name and address or the names and addresses of the stockholder or stockholders, or director or directors, to whom any such notice or notices was or were given, and the time and method of giving the same, shall in the absence of fraud, be prima facie evidence of the facts therein contained.
- (d) It shall not be necessary that the same method of giving notice be employed in respect of all recipients of notice, but one permissible method may be employed in respect of any one or more recipients, and any other permissible method or methods may be employed in respect of any other or others.
- (e) Whenever notice is required to be given, under any provision of the DGCL, the Certificate of Incorporation or these Bylaws, to any person with whom communication is unlawful, the giving of such notice to such person shall not be required and there shall be no duty to apply to any governmental authority or agency for a license or permit to give such notice to such person. Any action or meeting which shall be taken or held without notice to any such person with whom communication is unlawful shall have the same force and effect as if such notice had been duly given. In the event the action taken by the Corporation is such as to require the filing of a certificate under any provision of the DGCL, the certificate shall state, if such is the fact and if notice is required, that notice was given to all persons entitled to receive notice except such persons with whom communication is unlawful.
- (f) Whenever notice is required to be given, under any provision of the DGCL, the Certificate of Incorporation or these Bylaws, to any stockholder to whom (i) notice of two (2) consecutive annual meetings, or (ii) all, and at least two (2), payments (if sent by first-class mail) of dividends or interest on securities during a twelve (12) month period, have been mailed addressed to such person at such person's address as shown on the records of the Corporation and have been returned undeliverable, the giving of such notice to such person shall not be required. Any actions or meeting which shall be taken or held without notice to such person shall have the same force and effect as if such notice had been duly given. If any such person shall deliver to the Corporation a written notice setting forth such person's then current address, the requirement that

notice be given to such person shall be reinstated. In the event that the action taken by the Corporation is such as to require the filing of a certificate under any provision of the DGCL, the certificate need not state that the Corporation did not give notice to persons not required to be given notice pursuant to Section 230(b) of the DGCL. The exception in clause (i) above to the requirement that notice be given shall not be applicable to any notice returned as undeliverable if the notice was given by electronic transmission.

- (g) Except as otherwise prohibited under the DGCL, any notice given under the provisions of the DGCL, the Certificate of Incorporation or these Bylaws shall be effective if given by a single written notice to stockholders who share an address if consented to by the stockholders at that address to whom such notice is given. Such consent shall be deemed to have been given if such stockholder fails to object in writing to the Corporation within sixty (60) days of having been given notice by the Corporation of its intention to send the single notice. Any consent shall be revocable by the stockholder by written notice to the Corporation.
- (h) Without limiting the manner by which notice otherwise may be given effectively to stockholders, any notice to stockholders given by the Corporation under any provision of the DGCL, the Certificate of Incorporation or these Bylaws shall be effective if given by a form of electronic transmission previously consented to by the stockholder to whom the notice is given. Any such consent shall be revokable by the stockholder by written notice to the Corporation. Any such consent shall be deemed revoked if (i) the Corporation is unable to deliver by electronic transmission two (2) consecutive notices given by the Corporation in accordance with such consent, and (ii) such inability becomes known to the Secretary or an Assistant Secretary of the Corporation, the transfer agent or other person responsible for the giving of notice; provided, however, the inadvertent failure to treat such inability as a revocation shall not invalidate any meeting or other action.

Notice given pursuant to the above paragraph shall be deemed given (i) if by facsimile telecommunication, when directed to a number at which the stockholder has consented to receive notice, (ii) if by a posting on an electronic network together with a separate notice to the stockholder of such specific posting, upon the later of (A) such posting and (B) the giving of such separate notice, and (iv) if by any other form of electronic transmission, when directed to the stockholder. An affidavit of the Secretary or Assistant Secretary, the transfer agent or other agent of the Corporation that the notice has been given by a form of electronic transmission shall in the absence of fraud, be prima facie evidence of the facts stated therein.

For purposes of these Bylaws, "electronic transmission" means any form of communication, not directly involving the physical transmission of paper, that creates a record that may be retained, retrieved and reviewed by a recipient thereof, and that may be directly reproduced in paper form by such a recipient through an automated process. This Section 9.3 shall not apply to Section 164 (failure to pay for stock; remedies), Section 296 (adjudication of claims; appeal), Section 311 (revocation of voluntary

23

dissolution), Section 312 (renewal, revival, extension and restoration of certificate of incorporation) or Section 324 (attachment of shares of stock) of the DGCL.

- 9.4 <u>Construction.</u> Unless the context requires otherwise, the general provisions, rules of construction and definitions in the DGCL shall govern the construction of these Bylaws. The singular number includes the plural, and the plural number includes the singular. All pronouns used in these Bylaws shall be deemed to refer to the masculine, feminine and/or neuter, as the identity of the person or persons so designated may require.
- 9.5 <u>Conflict With Applicable Law or Certificate of Incorporation</u>. These Bylaws are adopted subject to any applicable law and the Certificate of Incorporation. Whenever these Bylaws may conflict with any applicable law or the Certificate of Incorporation, such conflict shall be resolved in favor of such law or the Certificate of Incorporation.

ARTICLE X INDEMNIFICATION AND ADVANCEMENT OF EXPENSES

- 10.1 Right to Indemnification. The Corporation shall indemnify and hold harmless, to the fullest extent permitted by applicable law as it presently exists or may hereafter be amended, any person (an "Indemnitee") who was or is made or is threatened to be made a party or is otherwise involved in any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative (a "Proceeding"), by reason of the fact that he or she, or a person for whom he or she is the legal representative, is or was a director, officer, employee or agent of another corporation or of a partnership, joint venture, trust or other enterprise, including service with respect to employee benefit plans (any such entity, an "Other Entity"), against all liability and loss suffered (including, but not limited to expenses (including but not limited to, attorneys' fees and expenses), judgments, fines and amounts paid in settlement actually and reasonably incurred by such Indemnitee in connection with any such Proceeding) provided, that the Indemnitee acted in good faith and in a manner such Indemnitee reasonably believed to be in or not opposed to the best interests of the Corporation of any Proceeding by judgment, order, settlement, conviction or upon a plea of nolo contendere or its equivalent, does not, of itself, create a presumption that the Indemnitee did not act in good faith and in a manner in which he or she reasonably believed to be in or not opposed to the best interests of the Corporation, or that, with respect to any criminal proceeding he or she had reasonable cause to believe that his or her conduct was unlawful. The Corporation shall not indemnify an Indemnitee for any claim, issue or matter as to which the Indemnitee has been adjudged by a court of competent jurisdiction, after exhaustion of all appeals therefrom, to be liable to the Corporation or for any amounts paid in settlement to the Corporation, unless and only to the extent that the court in which the Proceeding was brought or
 - 10.2 Advancement of Expenses. The Corporation shall to the fullest extent not prohibited by applicable law pay the expenses (including attorneys' fees) incurred by an

2/

Indemnitee in defending any Proceeding in advance of its final disposition, provided, however, that, to the extent required by law, such payment of expenses in advance of the final disposition of the Proceeding shall be made only upon receipt of an undertaking by the Indemnitee to repay all amounts advanced if it should be ultimately determined that the Indemnitee is not entitled to be indemnified under this Article X or otherwise.

- 10.3 <u>Former Directors, Officers, Employees and Agents.</u> Indemnification pursuant to this Article X shall continue as to an Indemnitee who has ceased to be a director, officer, employee or agent of the Corporation or member, manager or managing member of a predecessor limited liability company or affiliate of such limited liability company or a director, officer, employee, agent, partner, member, manager or fiduciary of, or to serve in any other capacity for, another corporation or any partnership, joint venture, limited liability company, trust, or other enterprise and shall inure to the benefit of his or her heirs, executors and administrators.
- 10.4 Claims. If a claim for indemnification (following the final disposition of the Proceeding with respect to which indemnification is sought, including any settlement of such Proceeding) or advancement of expenses under this Article X is not paid in full within sixty (60) days after a written claim therefor by the Indemnitee has been received by the Corporation, the Indemnitee may file suit to recover the unpaid amount of such claim and, if successful in whole or in part, shall be entitled to be paid the expense of prosecuting such claim to the fullest extent permitted by applicable law. In any such action the Corporation shall have the burden of proving that the Indemnitee is not entitled to the requested indemnification or advancement of expenses under this Article X and applicable law.
- 10.5 Non-Exclusivity of Rights. The rights conferred on any Indemnitee by this Article X shall not be exclusive of any other rights which such Indemnitee may have or hereafter acquire under any statute, the Certificate of Incorporation, these Bylaws, or any agreement, vote of stockholders or disinterested directors or otherwise.

- 10.6 <u>Insurance</u>. The Corporation may purchase and maintain insurance on behalf of any person who is or was a director, officer, employee or agent of the Corporation, or is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation or of a partnership, joint venture, trust or other enterprise, including service with respect to employee benefit plans, against any liability asserted against such person and incurred by such person in any such capacity, or arising out of such person's status as such, whether or not the Corporation would have the power to indemnify such person against such liability under this Article X, the DGCL or otherwise.
- 10.7 <u>Amendment or Repeal.</u> Any right to indemnification or to advancement of expenses of any Indemnitee arising hereunder shall not be eliminated or impaired by an amendment to or repeal of this Article X after the occurrence of the act or omission that is the subject of the Proceeding for which indemnification or advancement of expenses is sought.
- 10.8 Saving Clause. If this Article X or any portion hereof shall be invalidated on any ground by any court of competent jurisdiction, then the Corporation shall nevertheless indemnify each director, officer, employee and agent to the fullest extent not prohibited by any applicable

portion of this Article X that shall not have been invalidated, or by any other applicable law. If this Article X shall be invalid due to the application of the indemnification provisions of another jurisdiction, then the Corporation shall indemnify each director, officer, employee and agent to the fullest extent under any other applicable law.

ARTICLE XI EXCLUSIVE JURISDICTION FOR CERTAIN ACTIONS

11.1 <u>Exclusive Jurisdiction for Certain Actions.</u> Unless the Corporation consents in writing to the selection of an alternative forum, the Court of Chancery of the State of Delaware shall be the sole and exclusive forum for (i) any derivative action or proceeding brought on behalf of the Corporation, (ii) any action asserting a claim for breach of a fiduciary duty owed by any director, officer, employee or agent of the Corporation to the Corporation or the Corporation or stockholders, (iii) any action asserting a claim governed by the internal affairs doctrine, in each case subject to said Court of Chancery having personal jurisdiction over the indispensable parties named as defendants therein.

ARTICLE XII AMENDMENTS

12.1 Amendments. The Board of Directors is expressly empowered to adopt, amend or repeal these Bylaws or adopt new Bylaws of the Corporation. Any adoption, amendment or repeal of these Bylaws or adoption of new Bylaws of the Corporation by the Board of Directors shall require the approval of a majority of the directors then in office. Notwithstanding the foregoing, the stockholders of the Corporation shall also have power to adopt, amend or repeal the Bylaws of the Corporation; provided, however, that, in addition to any vote of the holders of any class or series of stock of the Corporation required by law or by the Certificate of Incorporation, the affirmative vote of the holders of at least sixty-six and two-thirds percent (66-2/3%) of the voting power of all of the then-outstanding shares of the capital stock of the Corporation entitled to vote generally in the election of directors, voting together as a single class, shall be required to adopt, amend or repeal any provision of the Bylaws of the Corporation.

Adopted as of June 18, 2015.





RESTORGENEX CORPORATION
THE CORPORATION WILL FURNISH WITHOUT CHARGE TO EACH STOCKHOLDER WHO SO REQUESTS, A SUMMARY OF THE POWERS, DESIGNATIONS, PREFERENCES AND RELATIVE, PARTICIPATING, OPTIONAL OR OTHER SPECIAL RIGHTS OF EACH CLASS OF STOCK OF THE CORPORATION AND THE QUALIFICATIONS, LIMITATIONS OR RESTRECTIONS OF SUCH PREFERENCES AND RIGHTS, AND THE VARIATIONS IN RIGHTS, PREFERENCES AND LIMITATIONS DETERMINED FOR EACH SERIES, WHICH ARE FIXED BY THE CERTIFICATE OF INCORPORATION OF THE CORPORATION, AS AMENDED, AND THE RESOLUTIONS OF THE BOARD OF DIRECTORS TO DETERMINE VARIATIONS FOR FUTURE SERIES, SUCH REQUEST MAY BE MADE TO THE OFFICE OF THE SECRETARY OF THE CORPORATION OR TO THE TRANSFER AGENT. THE BOARD OF DIRECTORS MAY REQUIRE THE OWNER OF A LOST OR DESTROYED STOCK CERTIFICATE, OR HIS OR HER LEGAL REPRESENTATIVES, TO GIVE THE CORPORATION A BOND TO INDENNIFY IT AND ITS TRANSFER AGENTS AND REGISTRARS AGAINST ANY CLAIM THAT MAY BE MADE AGAINST THEM ON ACCOUNT OF THE ALLEGED LOSS OR DESTRUCTION OF ANY SUCH CERTIFICATE.

The following abbreviations, when used in the inscription on the face according to applicable laws or regulations:	of this certificate, shall be construed as though they were written out in full
TEN COM - as tenants in common UNIF 6	SIFT MIN ACTCustodian
TEN ENT - as tenants by the entireties	(Coat) (Miner) under Uniform Gifts to Minors Act. (State)
JT TEN - as joint tenants with right of survivorship UNIF T and not as tenants in common	RF MIN ACT
Additional abbreviations may also be used though not in the above	(Minor) (State)
For value received,hereby sell, assign	and transfer unto
PLEASE PRINT OR TYPEWRITE NAME AND ADDRESS, INCLUDING POSTAL ZIP CODE, OF ASSIGNEE)	
of the Capital Stock represented by the within Certificate, and do hereby	Share
,	Attorne
to transfer the said Stock on the books of the within-named Corporation	with full power or substitution in the premises.
Dated:20	Signature(s) Guaranteed: Medailien Guarantee Stamp The ECHATURES SHOULD BE GUARANTEED BY AN LIQUIED GUARANTEE BY AN AND SIGNATURE HISTORIAN BUSTUTION (Basks, Stackholer, Savigs and Loan Associations and Chedi Usens) WITH MEMBERSHIP AN AN APPROVED SIGNATURE GUARANTEE MEMBLICH PROGRAM, PURSUNT 10 SEC. OR LET 1744-15.
Signature:	
Signaturo:	
Notice: The signature to this assignment must correspond with as written upon the face of the certificate, in every without alteration or enlargement, or any change whate	particular,

SECURI TY I NSTRUCTI ONS



INDEMNIFICATION AGREEMENT

This Indemnification Agreement (this "Agreement") is made as of , 20 by and between RestorGenex Corporation, a Delaware corporation (the "Company"), and ("In This Agreement supersedes and replaces any and all previous agreements between the Company and Indemnitee covering the subject matter of this Agreement except as otherwise provided herein.

("Indemnitee").

RECITALS

WHEREAS, highly competent persons have become more reluctant to serve publicly-held corporations as directors or officers or in other capacities unless they are provided with adequate protection through insurance or adequate indemnification against inordinate risks of claims and actions against them arising out of their service to and activities on behalf of the corporation;

WHEREAS, the Board of Directors of the Company (the "Board") has determined that, in order to attract and retain qualified individuals, the Company will attempt to maintain on an ongoing basis, at its sole expense, liability insurance to protect persons serving the Company and its subsidiaries from certain liabilities. Although the furnishing of such insurance has been a customary and widespread practice among United States-based corporations and other business enterprises, the Company believes that, given current market conditions and trends, such insurance may be available to it in the future only at higher premiums and with more exclusions. At the same time, directors, officers and other persons in service to corporations or business enterprises are being increasingly subjected to expensive and time-consuming litigation relating to, among other things, matters that traditionally would have been brought only against the Company or business enterprise itself. The Certificate of Incorporation of the Company (the "Certificate of Incorporation") and Bylaws of the Company (the "Bylaws") require indemnification of the directors and officers of the Company, Indemnitee also may be entitled to indemnification pursuant to the General Corporation Law of the State of Delaware (the "DGCL"). The Certificate of Incorporation, Bylaws and the DGCL expressly provide that contracts may be entered into between the Company and members of the board of directors, officers and other persons with respect to indemnification;

WHEREAS, the uncertainties relating to such insurance and to indemnification have increased the difficulty of attracting and retaining such persons;

WHEREAS, the Board has determined that the increased difficulty in attracting and retaining such persons is detrimental to the best interests of the Company and its stockholders and that the Company should act to assure such persons that there will be increased certainty of such protection in the future;

WHEREAS, it is reasonable, prudent and necessary for the Company contractually to obligate itself to indemnify, and to advance expenses on behalf of, such persons to the fullest extent permitted by applicable law so that they will serve or continue to serve the Company free from undue concern that they will not be so indemnified;

WHEREAS, this Agreement is a supplement to and in furtherance of the Certificate of Incorporation and Bylaws and any resolutions adopted pursuant thereto, and shall not be deemed a substitute therefor, nor to diminish or abrogate any rights of Indemnitee thereunder; and

WHEREAS, Indemnitee does not regard the protection available under the Certificate of Incorporation and Bylaws and insurance as adequate in the present circumstances, and may not be willing to serve as a director or officer without adequate protection, and the Company desires Indemnitee to serve in such capacity. Indemnitee is willing to serve, continue to serve and to take on additional service for or on behalf of the Company on the condition that he or she be so indemnified.

NOW, THEREFORE, in consideration of the premises and the covenants contained herein, the Company and Indemnitee do hereby covenant and agree as follows:

Section 1. Services to the Company. Indemnitee agrees to serve as a director or officer of the Company, as applicable, or, at the request of the Company, as a director, officer, employee, agent or fiduciary of another corporation, partnership, joint venture, trust or other enterprise, as applicable. Indemnitee may at any time and for any reason resign from such position (subject to any other contractual obligation or any obligation imposed by operation of law), in which event the Company shall have no obligation under this Agreement to continue Indemnitee in such position. This Agreement shall not be deemed an employment contract between the Company (or any of its subsidiaries or any Enterprise) and Indemnitee. Indemnitee specifically acknowledges that Indemnitee's employment with the Company (or any of its subsidiaries or any Enterprise), if any, is at will, and the Indemnitee may be discharged at any time for any reason, with or without cause, except as may be otherwise provided in any written employment contract between Indemnitee and the Company (or any of its subsidiaries or any Enterprise), other applicable formal severance policies duly adopted by the Board, or, with respect to service as a director or officer of the Company, by the Certificate of Incorporation, the Bylaws and the DGCL. The foregoing notwithstanding, this Agreement shall continue in force after Indemnitee has ceased to serve as a director or officer of the Company, as applicable, or, at the request of the Company, as a director, officer, employee, agent or fiduciary of another corporation, partnership, joint venture, trust or other enterprise, as applicable, as provided in Section 16 hereof.

Section 2. <u>Definitions</u>. As used in this Agreement:

- (a) References to "agent" shall mean any person who is or was a director, officer or employee of the Company or a subsidiary of the Company or other person authorized by the Company to act for the Company, to include such person serving in such capacity as a director, officer, employee, fiduciary or other official of another corporation, partnership, limited liability company, joint venture, trust or other enterprise at the request of, for the convenience of, or to represent the interests of the Company or a subsidiary of the Company.
 - (b) A "Change in Control" shall be deemed to occur upon the earliest to occur after the date of this Agreement of any of the following events:

2

- i. <u>Acquisition of Stock by Third Party.</u> Any Person (as defined below) is or becomes the Beneficial Owner (as defined below), directly or indirectly, of securities of the Company representing 50.1% or more of the combined voting power of the Company's then outstanding securities unless the change in relative Beneficial Ownership of the Company's securities by any Person results solely from a reduction in the aggregate number of outstanding shares of securities entitled to vote generally in the election of directors;
- ii. Change in Board of Directors. During any period of twelve (12) consecutive months, individuals who at the beginning of such period constitute the Board, and any new director (other than a director designated by a person who has entered into an agreement with the Company to effect a transaction described in Sections 2(b)(i), 2(b)(iii) or 2(b)(iii) or 2(b)(iii) or a 2(b)(iii) or 2(b)(iiii) or 2(b)(iii) or 2(b)(iii) or 2(b)(iiii) or 2(b)(iii) or 2(b)(iii) or 2(b)(iii)
- iii. <u>Corporate Transactions</u>. The effective date of a merger or consolidation of the Company with any other entity, other than a merger or consolidation which would result in the voting securities of the Company outstanding immediately prior to such merger or consolidation continuing to represent (either by remaining outstanding or by being converted into voting securities of the surviving entity) more than 50.1% of the combined voting power of the voting securities of the surviving entity outstanding immediately after such merger or consolidation and with the power to elect at least a majority of the board of directors or other governing body of such surviving entity;
- iv. <u>Liquidation</u>. The approval by the stockholders of the Company of a complete liquidation of the Company or an agreement for the sale or disposition by the Company of all or substantially all of the Company's assets; and
- v. Other Events. There occurs any other event of a nature that would be required to be reported in response to Item 6(e) of Schedule 14A of Regulation 14A (or a response to any similar item on any similar schedule or form) promulgated under the Exchange Act (as defined below), whether or not the Company is then subject to such reporting requirement.

For purposes of this Section 2(b), the following terms shall have the following meanings:

- (A) "Exchange Act" shall mean the Securities Exchange Act of 1934, as amended from time to time.
- (B) "Person" shall have the meaning as set forth in Sections 13(d) and 14(d) of the Exchange Act; <u>provided</u>, <u>however</u>, that Person shall exclude (i) the Company, (ii) any trustee or other fiduciary holding securities under an employee benefit plan of the Company, and (iii) any corporation owned, directly or indirectly, by the stockholders of the Company in substantially the same proportions as their ownership of stock of the Company.

- (C) "Beneficial Owner" shall have the meaning given to such term in Rule 13d-3 under the Exchange Act; <u>provided</u>, <u>however</u>, that Beneficial Owner shall exclude any Person otherwise becoming a Beneficial Owner by reason of the stockholders of the Company approving a merger of the Company with another entity.
- (c) "Corporate Status" describes the status of a person who is or was a director, trustee, partner, managing member, officer, employee, agent or fiduciary of the Company or of any other corporation, limited liability company, partnership or joint venture, trust or other enterprise which such person is or was serving at the request of the Company.
 - (d) "Disinterested Director" shall mean a director of the Company who is not and was not a party to the Proceeding in respect of which indemnification is sought by Indemnitee.
- (e) "Enterprise" shall mean the Company and any other corporation, limited liability company, partnership, joint venture, trust or other enterprise of which Indemnitee is or was serving at the request of the Company as a director, officer, trustee, partner, managing member, employee, agent or fiduciary.

- (f) "Expenses" shall include all reasonable attorneys' fees, retainers, court costs, transcript costs, fees of experts and other professionals, witness fees, travel expenses, duplicating costs, printing and binding costs, telephone charges, postage, delivery service fees, fax transmission charges, secretarial services, any federal, state, local or foreign taxes imposed on Indemnitee as a result of the actual or deemed receipt of any generating to prosecute or defend, investigating, being or preparing to be a deponent or witness in, or otherwise participating in, a Proceeding. Expenses also shall include (i) Expenses incurred in connection with any appeal resulting from any Proceeding, including without limitation the premium, security for, and other costs relating to any cost bond, supersedeas bond or other appeal bond or its equivalent, and (ii) Expenses incurred in connection with recovery under any directors' and officers' liability insurance policies maintained by the Company, regardless of whether the Indemnitee is ultimately determined to be entitled to such indemnification, advancement or Expenses or insurance recovery, as the case may be, and (iii) for purposes of Section 14(d) only, Expenses incurred by or on behalf of Indemnitee in connection with the interpretation, enforcement or defense of Indemnitee's rights under this Agreement, by litigation or otherwise. The parties agree that for the purposes of any advancement of Expenses for which Indemnitee has made written demand to the Company in accordance with this Agreement, all Expenses included in such demand that are certified by affidavit of Indemnitee's counsel as being reasonable shall be presumed conclusively to be reasonable. Expenses, however, shall not include amounts paid in settlement by Indemnitee or the amount of judgments or fines against Indemnitee.
- (g) "Independent Counsel" shall mean a law firm, or a member of a law firm, that is experienced in matters of corporation law and neither presently is, nor in the past five (5) years has been, retained to represent: (i) the Company or Indemnitee in any matter material to either such party (other than with respect to matters concerning the Indemnitee under this Agreement, or of other indemnitees under similar indemnification agreements), or (ii) any other party to the

Proceeding giving rise to a claim for indemnification hereunder. Notwithstanding the foregoing, the term "Independent Counsel" shall not include any person who, under the applicable standards of professional conduct then prevailing, would have a conflict of interest in representing either the Company or Indemnitee in an action to determine Indemnitee's rights under this Agreement. The Company agrees to pay the reasonable fees and expenses of the Independent Counsel referred to above and to fully indemnify such counsel against any and all Expenses, claims, liabilities and damages arising out of or relating to this Agreement or its engagement pursuant hereto.

- (h) The term "Proceeding" shall include any threatened, pending or completed action, suit, claim, counterclaim, cross claim, arbitration, mediation, alternate dispute resolution mechanism, investigation, inquiry, administrative hearing or any other actual, threatened or completed proceeding, whether brought in the right of the Company or otherwise and whether of a civil, criminal, administrative, regulatory, legislative or investigative (formal or informal) nature, including any appeal therefrom, in which Indemnitee was, is or will be involved as a party, potential party, non-party witness or otherwise by reason of the fact that Indemnitee is or was a director or officer of the Company, by reason of any action taken by him or her (or a failure to take action by him or her) or of any action (or failure to act) on his or her part while acting pursuant to his or her Corporate Status, in each case whether or not serving in such capacity at the time any liability or Expense is incurred for which indemnification, reimbursement or advancement of Expenses can be provided under this Agreement. If the Indemnitee believes in good faith that a given situation may lead to or culminate in the institution of a Proceeding, this shall be considered a Proceeding under this paragraph.
- (i) Reference to "other enterprise" shall include employee benefit plans; references to "fines" shall include any excise tax assessed with respect to any employee benefit plan; references to "serving at the request of the Company" shall include any service as a director, officer, employee or agent with respect to an employee benefit plan, its participants or beneficiaries; and a person who acted in good faith and in a manner he or she reasonably believed to be in the best interests of the participants and beneficiaries of an employee benefit plan shall be deemed to have acted in manner "not opposed to the best interests of the Company" as referred to in this Agreement.
- Section 3. <u>Indemnity in Third-Party Proceedings.</u> The Company shall indemnify Indemnitee in accordance with the provisions of this Section 3 if Indemnitee is, or is threatened to be made, a party to or a participant in any Proceeding, other than a Proceeding by or in the right of the Company to procure a judgment in its favor. Pursuant to this Section 3, Indemnitee shall be indemnified to the fullest extent permitted by applicable law against all Expenses, judgments, fines and amounts paid in settlement) actually and reasonably incurred by Indemnitee or on his or her behalf in connection with such Proceeding or any claim, issue or matter therein, if Indemnitee acted in good faith and in a manner he or she reasonably believed to be in or not opposed to the best interests of the Company and, in the case of a criminal Proceeding had no reasonable cause to believe that his or her conduct was unlawful. The parties hereto intend that this Agreement shall provide to the fullest extent permitted by law for indemnification in excess of that expressly

5

permitted by statute, including, without limitation, any indemnification provided by the Certificate of Incorporation, the Bylaws, vote of its stockholders or disinterested directors or applicable law.

- Section 4. Indemnity in Proceedings by or in the Right of the Company. The Company shall indemnify Indemnitee in accordance with the provisions of this Section 4 if Indemnitee is, or is threatened to be made, a party to or a participant in any Proceeding by or in the right of the Company to procure a judgment in its favor. Pursuant to this Section 4, Indemnitee shall be indemnified to the fullest extent permitted by applicable law against all Expenses actually and reasonably incurred by him or her or on his or her behalf in connection with such Proceeding or any claim, issue or matter therein, if Indemnitee acted in good faith and in a manner he or she reasonably believed to be in or not opposed to the best interests of the Company. No indemnification for Expenses shall be made under this Section 4 in respect of any claim, issue or matter as to which Indemnitee shall have been finally adjudged by a court to be liable to the Company, unless and only to the extent that the Delaware Court of Chancery or any court in which the Proceeding was brought shall determine upon application that, despite the adjudication of liability but in view of all the circumstances of the case, Indemnitee is fairly and reasonably entitled to indemnification.
- Section 5. <u>Indemnification for Expenses of a Party Who is Wholly or Partly Successful</u>. Notwithstanding any other provisions of this Agreement, to the fullest extent permitted by applicable law and to the extent that Indemnitee is a party to (or a participant in) and is successful, on the merits or otherwise, in any Proceeding or in defense of any claim, issue or matter therein, in whole or in part, the Company shall indemnify Indemnitee against all Expenses actually and reasonably incurred by him or her in connection therewith. If Indemnitee is not wholly successful in such Proceeding but is successful, on the merits or otherwise, as to one or more but less than all claims, issues or matters in such Proceeding, the Company shall indemnity Indemnitee against all Expenses actually and reasonably incurred by him or her or on his or her behalf in connection with or related to each successfully resolved claim, issue or matter to the fullest extent permitted by law. For purposes of this Section and without limitation, the termination of any claim, issue or matter in such a Proceeding by dismissal, with or without prejudice, shall be deemed to be a successful result as to such claim, issue or matter.
- Section 6. <u>Indemnification For Expenses of a Witness</u>. Notwithstanding any other provision of this Agreement, to the fullest extent permitted by applicable law and to the extent that Indemnitee is, by reason of his or her Corporate Status, a witness or otherwise asked to participate in any aspect of a Proceeding to which Indemnitee is not a party, he or she shall be indemnified against all Expenses actually and reasonably incurred by him or her on his or her behalf in connection therewith.
- Section 7. Partial Indemnification. If Indemnitee is entitled under any provision of this Agreement to indemnification by the Company for some or a portion of Expenses, but not, however, for the total amount thereof, the Company shall nevertheless indemnify Indemnitee for the portion thereof to which Indemnitee is entitled.

6

Section 8. <u>Additional Indemnification</u>.

- (a) Notwithstanding any limitation in Sections 3, 4 or 5, the Company shall indemnify Indemnitee to the fullest extent permitted by applicable law if Indemnitee is a party to or threatened to be made a party to or a participant in any Proceeding (including a Proceeding by or in the right of the Company to procure a judgment in its favor) against all Expenses, judgments, fines and amounts paid in settlement (including all interest, assessments and other charges paid or payable in connection with or in respect of such Expenses, judgments, fines and amounts paid in settlement) actually and reasonably incurred by or on behalf of Indemnitee in connection with the Proceeding.
 - (b) For purposes of Section 8(a), the meaning of the phrase "to the fullest extent permitted by applicable law" shall include, but not be limited to:
- i. to the fullest extent permitted by the provision of the DGCL that authorizes or contemplates additional indemnification by agreement, or the corresponding provision of any amendment to or replacement of the DGCL, and
- ii. to the fullest extent authorized or permitted by any amendments to or replacements of the DGCL adopted after the date of this Agreement that increase the extent to which a corporation may indemnify its officers and directors.
- Section 9. Exclusions. Notwithstanding any provision in this Agreement, the Company shall not be obligated under this Agreement to make any indemnification payment in connection with any claim made against Indemnitee:
- (a) for which payment has actually been made to or on behalf of Indemnitee under any insurance policy or other indemnity provision, except with respect to any excess beyond the amount paid under any insurance policy or other indemnity provision; or
- (b) for (i) an accounting of profits made from the purchase and sale (or sale and purchase) by Indemnitee of securities of the Company within the meaning of Section 16(b) of the Exchange Act (as defined in Section 2(b) hereof) or similar provisions of state statutory law or common law, or (ii) any reimbursement of the Company by the Indemnitee of any bonus or other incentive-based or equity-based compensation or of any porfits realized by the Indemnitee from the sale of securities of the Company, as required in each case under the Exchange Act (including any such reimbursements that arise from an accounting restatement of the Company pursuant to Section 304 of the Sarbanes-Oxley Act of 2002 (the "Sarbanes-Oxley Act"), or the payment to the Company of profits arising from the purchase and sale by Indemnitee of securities in violation of Section 306 of the Sarbanes-Oxley Act); or
- (c) except as provided in Section 14(d) of this Agreement, in connection with any Proceeding (or any part of any Proceeding) initiated by Indemnitee, including any Proceeding (or any part of any Proceeding) initiated by Indemnitee against the Company or its directors, officers, employees or other indemnitees, unless (i) the Board authorized the Proceeding (or any part of any Proceeding) prior to its initiation, (ii) such payment arises in connection with any mandatory counterclaim or cross-claim or affirmative defense brought or raised by Indemnitee in any Proceeding (or any part of any Proceeding), or (iii) the Company provides the

indemnification, in its sole discretion, pursuant to the powers vested in the Company under applicable law.

Section 10. Advances of Expenses. Notwithstanding any provision of this Agreement to the contrary (other than Section 14(d)), the Company shall advance, to the extent not prohibited by law, the Expenses incurred by or on behalf of Indemnitee in connection with any Proceeding (or any part of any Proceeding) not initiated by Indemnitee, and such advancement shall be made within thirty (30) days after the receipt by the Company of a statement or statements requesting such advances from time to time (which shall include invoices received by the Indemnitee in connection with such Expenses but, in the case of invoices in connection with legal services, any references to legal work performed or to expenditures made that would cause Indemnitee to waive any privilege accorded by applicable law shall not be so included), whether prior to or after final disposition of any Proceeding. Advances shall be unsecured and interest free. Advances shall be made without regard to Indemnitee's ability to repay the Expenses and without regard to Indemnitee's ultimate entitlement to indemnification under the other provisions of this Agreement. In accordance with Section 14(d), advances shall include any and all reasonable Expenses incurred pursuing an action to enforce this right of advancement, including Expenses incurred preparing and forwarding statements to the Company to support the advances claimed. The Indemnitee shall qualify for advances upon the execution and delivery to the Company of this Agreement, which shall constitute an undertaking providing that the Indemnitee undertakes to repay the amounts advanced (without interest) to the extent that it is ultimately determined that Indemnitee is not entitled to be indemnified by the Company. No other form of undertaking shall be required other than the execution of this Agreement. This Section 10 shall not apply to any claim made by Indemnitee for which indemnity is excluded pursuant to Section 9.

Section 11. <u>Procedure for Notification and Defense of Claim.</u>

- (a) Indemnitee shall notify the Company in writing of any matter with respect to which Indemnitee intends to seek indemnification or advancement of Expenses hereunder as soon as reasonably practicable following the receipt by Indemnitee of written notice thereof or Indemnitee's becoming aware thereof. The written notification to the Company shall include a description of the nature of the Proceeding and the facts underlying the Proceeding, in each case to the extent known to Indemnitee. To obtain indemnification under this Agreement, Indemnitee shall submit to the Company a written request, including therein or therewith such documentation and information as is reasonably available to Indemnitee and is reasonably necessary to determine whether and to what extent Indemnitee is entitled to indemnification following the final disposition of such Proceeding. The failure by Indemnitee to notify the Company hereunder will not relieve the Company from any liability which it may have to Indemnitee hereunder or otherwise than under this Agreement, and any delay in so notifying the Company shall not constitute a waiver by Indemnitee of any rights under this Agreement, except to the extent (solely with respect to the indemnity hereunder) that such failure or delay materially prejudices the Company.
 - (b) The Company will be entitled to participate in the Proceeding at its own expense.

8

(c) The Company shall not settle any Proceeding (in whole or in part) if such settlement would impose any Expense, judgment, liability, fine, penalty or limitation on Indemnitee which Indemnitee is not entitled to be indemnified hereunder without the Indemnitee's prior written consent.

Section 12. <u>Procedure Upon Application for Indemnification</u>

- (a) Upon written request by Indemnitee for indemnification pursuant to Section 11(a), a determination, if required by applicable law, with respect to Indemnitee's entitlement thereto shall be made in the specific case: (i) if a Change in Control shall have occurred, by Independent Counsel in a written opinion to the Board, a copy of which shall be delivered to Indemnitee; or (ii) if a Change in Control shall not have occurred, (A) by a majority vote of the Disinterested Directors, even though less than a quorum of the Board, (B) by a committee of Disinterested Directors designated by a majority vote of the Disinterested Directors, even though less than a quorum of the Board, (C) if there are no such Disinterested Directors so direct, by Independent Counsel in a written opinion to the Board, a copy of which shall be delivered to Indemnitee, or (D) if so directed by the Board, by the stockholders of the Company; and, if it is so determined that Indemnitee is entitled to indemnification, payment to Indemnitee shall be made within ten (10) days after such determination. Indemnitee shall cooperate with the person, persons or entity making such determination with respect to Indemnitee's entitlement to indemnification, including providing to such person, persons or entity upon reasonable advance request any documentation or information which is not privileged or otherwise protected from disclosure and which is reasonably available to Indemnitee and reasonably necessary to such determination. Any costs or Expenses (including attorneys' fees and disbursements) incurred by or on behalf of Indemnitee in so cooperating with the person, persons or entity making such determination shall be borne by the Company (irrespective of the determination as to Indemnitee's entitlement to indemnification, including a description of any reason or basis for which indemnification has been denied.
- (b) In the event the determination of entitlement to indemnification is to be made by Independent Counsel pursuant to Section 12(a) hereof, the Independent Counsel shall be selected as provided in this Section 12(b). If a Change in Control shall not have occurred, the Independent Counsel shall be selected by the Board, and the Company shall give written notice to Indemnitee advising him or her of the identity of the Independent Counsel so selected. If a Change in Control shall have occurred, the Independent Counsel shall be selected by Indemnitee (unless Indemnitee shall request that such selection be made by the Board, in which event the preceding sentence shall apply), and Indemnitee shall give written notice to the Company advising it of the identity of the Independent Counsel so selected. In either event, Indemnitee or the Company, as the case may be, may, within ten (10) days after such written notice of selection shall have been given, deliver to the Company or to Indemnitee, as the case may be, a written objection to such selection; provided, however, that such objection may be asserted only on the ground that the Independent Counsel so selected does not meet the requirements of "Independent Counsel" as defined in Section 2 of this Agreement, and the objection shall set forth with particularity the factual basis of such assertion. Absent a proper and timely objection, the person

9

so selected shall act as Independent Counsel. If such written objection is so made and substantiated, the Independent Counsel so selected may not serve as Independent Counsel unless and until such objection is withdrawn or the Delaware Court has determined that such objection is without merit. If, within twenty (20) days after the later of submission by Indemnitee of a written request for indemnification pursuant to Section 11(a) hereof and the final disposition of the Proceeding, no Independent Counsel shall have been selected and not objected to, either the Company or Indemnitee may petition the Delaware Court for resolution of any objection which shall have been made by the Company or Indemnitee to the other's selection of Independent Counsel and/or for the appointment as Independent Counsel of a person selected by such court or by such other person as such court shall designate, and the person with respect to whom all objections are so resolved or the person so appointed shall act as Independent Counsel under Section 12(a) hereof. Upon the due commencement of any judicial proceeding or arbitration pursuant to Section 14(a) of this Agreement, Independent Counsel shall be discharged and relieved of any further responsibility in such capacity (subject to the applicable standards of professional conduct then prevailing).

(c) If the Company disputes a portion of the amounts for which indemnification is requested, the undisputed portion shall be paid and only the disputed portion withheld pending resolution of any such dispute.

Section 13. <u>Presumptions and Effect of Certain Proceedings.</u>

- (a) In making a determination with respect to entitlement to indemnification hereunder, the person or persons or entity making such determination shall, to the fullest extent not prohibited by law, presume that Indemnitee is entitled to indemnification under this Agreement if Indemnitee has submitted a request for indemnification in accordance with Section 11(a) of this Agreement, and the Company shall, to the fullest extent not prohibited by law, have the burden of proof to overcome that presumption in connection with the making by any persons, persons or entity of any determination contrary to that presumption. Neither the failure of the Company (including by its directors or Independent Counsel) to have made a determination by the Company (including by its directors or Independent Counsel) that Indemnitee has not met such applicable standard of conduct, shall be a defense to the action or create a presumption that Indemnitee has not met the applicable standard of conduct.
- (b) Subject to Section 14(e), if the person, persons or entity empowered or selected under Section 12 of this Agreement to determine whether Indemnitee is entitled to indemnification shall not have made a determination within sixty (60) days after receipt by the Company of the request therefor, the requisite determination of entitlement to indemnification shall, to the fullest extent not prohibition been made and Indemnitee shall be entitled to such indemnification, absent (i) a misstatement by Indemnitee of a material fact, or an omission of a material fact necessary to make Indemnitee's statement not materially misleading, in connection with the request for indemnification, or (ii) a prohibition of such indemnification under applicable law; provided. however, that such sixty (60)-day period may be extended for a reasonable time, not to exceed an additional thirty (30) days, if the person, persons or entity making the determination with respect to entitlement to indemnification in good

10

faith requires such additional time for the obtaining or evaluating of documentation and/or information relating thereto; and provided, further, that the foregoing provisions of this Section 13(b) shall not apply (i) if the determination of entitlement to indemnification is to be made by the stockholders pursuant to Section 12(a) of this Agreement and if (A) within fifteen (15) days after receipt by the Company of the request for such determination the Board has resolved to submit such determination to the stockholders for their consideration at an annual meeting thereof to be held within seventy-five (75) days after such receipt and such determination is made thereat, or (B) a special meeting of stockholders is called within fifteen (15) days after such receipt for the purpose of making such determination, such meeting is held for such purpose within sixty (60) days after having been so called and such determination is made thereat, or (ii) if the determination of entitlement to indemnification is to be made by Independent Counsel pursuant to Section 12(a) of this Agreement.

- (c) The termination of any Proceeding or of any claim, issue or matter therein, by judgment, order, settlement or conviction, or upon a plea of nolo contendere or its equivalent, shall not (except as otherwise expressly provided in this Agreement) of itself adversely affect the right of Indemnitee to indemnification or create a presumption that Indemnitee did not act in good faith and in a manner which he or she reasonably believed to be in or not opposed to the best interests of the Company or, with respect to any criminal Proceeding, that Indemnitee had reasonable cause to believe that his or her conduct was unlawful.
- (d) For purposes of any determination of good faith, Indemnitee shall be deemed to have acted in good faith if Indemnitee's action is based on the records or books of account of the Enterprise, including financial statements, or on information supplied to Indemnitee by the directors or officers of the Enterprise in the course of their duties, or on the advice of legal counsel for the Enterprise or on information or records

given or reports made to the Enterprise by an independent certified public accountant or by an appraiser or other expert selected with reasonable care by the Enterprise. The provisions of this Section 13(d) shall not be deemed to be exclusive or to limit in any way the other circumstances in which Indemnitee may be deemed to have met the applicable standard of conduct set forth in this Agreement. Whether or not the foregoing provisions of this Section 13(d) are satisfied, it shall in any event be presumed that Indemnitee has at all times acted in good faith and in a manner reasonably believed to be in or not opposed to the best interests of the Company.

(e) The knowledge and/or actions, or failure to act, of any director, officer, trustee, partner, managing member, fiduciary, agent or employee of the Enterprise shall not be imputed to Indemnitee for purposes of determining the right to indemnification under this Agreement.

Section 14. Remedies of Indemnitee

(a) Subject to Section 14(e), in the event that (i) a determination is made pursuant to Section 12 of this Agreement that Indemnitee is not entitled to indemnification under this Agreement, (ii) advancement of Expenses is not timely made pursuant to Section 10 of this Agreement, (iii) no determination of entitlement to indemnification shall have been made pursuant to Section 12(a) of this Agreement within ninety (90) days after receipt by the Company of the request for indemnification, (iv) payment of indemnification is not made pursuant to Section 5, 6 or 7 or the last sentence of Section 12(a) of this Agreement within ten

11

(10) days after receipt by the Company of a written request therefor, (v) payment of indemnification pursuant to Section 3, 4 or 8 of this Agreement is not made within ten (10) days after a determination has been made that Indemnitee is entitled to indemnification, or (vi) in the event that the Company or any other person takes or threatens to take any action to declare this Agreement void or unenforceable, or institutes any litigation or other action or Proceeding designed to deny, or to recover from, the Indemnitee the benefits provided or intended to be provided to the Indemnitee hereunder, Indemnitee shall be entitled to an adjudication by a court of his or her entitlement to such indemnification or advancement of Expenses. Alternatively, Indemnitee, at his or her option, may seek an award in arbitration to be conducted by a single arbitrator pursuant to the Commercial Arbitration Rules of the American Arbitration Association. Indemnitee such proceeding seeking an adjudication or an award in arbitration within one hundred eighty (180) days following the date on which Indemnitee first has the right to commence such proceeding pursuant to this Section 14(a); provided, however, that the foregoing clause shall not apply in respect of a proceeding brought by Indemnitee to enforce his or her rights under Section 5 of this Agreement. The Company shall not oppose Indemnitee's right to seek any such adjudication or award in arbitration.

- (b) In the event that a determination shall have been made pursuant to Section 12(a) of this Agreement that Indemnitee is not entitled to indemnification, any judicial proceeding or arbitration commenced pursuant to this Section 14 shall be conducted in all respects as a de novo trial, or arbitration, on the merits and Indemnitee shall not be prejudiced by reason of that adverse determination. In any judicial proceeding or arbitration commenced pursuant to this Section 14 the Company shall have the burden of proving Indemnitee is not entitled to indemnification or advancement of Expenses, as the case may be.
- (c) If a determination shall have been made pursuant to Section 12(a) of this Agreement that Indemnitee is entitled to indemnification, the Company shall be bound by such determination in any judicial proceeding or arbitration commenced pursuant to this Section 14, absent (i) a misstatement by Indemnitee of a material fact, or an omission of a material fact necessary to make Indemnitee's statement not materially misleading, in connection with the request for indemnification, or (ii) a prohibition of such indemnification under applicable law.
- (d) The Company shall, to the fullest extent not prohibited by law, be precluded from asserting in any judicial proceeding or arbitration commenced pursuant to this Section 14 that the procedures and presumptions of this Agreement are not valid, binding and enforceable and shall stipulate in any such court or before any such arbitrator that the Company is bound by all the provisions of this Agreement. It is the intent of the Company that, to the fullest extent permitted by law, the Indemnitee not be required to incur legal fees or other Expenses associated with the interpretation, enforcement or defense of Indemnitee's rights under this Agreement by litigation or otherwise because the cost and expense thereof would substantially detract from the benefits intended to be extended to the Indemnitee. The Company shall, to the fullest extent permitted by law, indemnify Indemnitee against any and all Expenses and, if requested by Indemnitee, shall (within ten (10) days after receipt by the Company of a written request therefor) advance, to the extent not prohibited by law, such Expenses to Indemnitee, which are incurred by or on behalf of Indemnitee in connection with any action brought by Indemnitee for indemnification or advancement of Expenses from the Company under this Agreement or under any directors' and officers' liability insurance policies maintained by the Company if, in the case

12

of indemnification, Indemnitee is wholly successful on the underlying claims; if Indemnitee is not wholly successful on the underlying claims, then such indemnification shall be only to the extent Indemnitee is successful on such underlying claims or otherwise as permitted by law, whichever is greater.

(e) Notwithstanding anything in this Agreement to the contrary, no determination as to entitlement of Indemnitee to indemnification under this Agreement shall be required to be made prior to the final disposition of the Proceeding.

Section 15. <u>Non-exclusivity; Survival of Rights; Insurance; Subrogation.</u>

- (a) The rights of indemnification and to receive advancement of Expenses as provided by this Agreement (i) shall not be deemed exclusive of any other rights to which Indemnitee may at any time be entitled under applicable law, the Certificate of Incorporation, the Bylaws, any agreement, a vote of stockholders or a resolution of directors or otherwise, and (ii) shall be interpreted independently of, and without reference to, any other such rights to which Indemnitee may at any time be entitled. No amendment, alteration or repeal of this Agreement or of any provision hereof shall limit or restrict any right of Indemnitee under this Agreement in respect of any action taken or omitted by such Indemnitee in his or her Corporate Status prior to such amendment, alteration or repeal. To the extent that a change in Delaware law, whether by statute or judicial decision, permits greater indemnification or advancement of Expenses than would be afforded currently under the Bylaws, the Certificate of Incorporation and this Agreement, it is the intent of the parties hereto that Indemnitee shall enjoy by this Agreement the greater benefits so afforded by such change. No right or remedy herein conferred is intended to be exclusive of any other right or remedy, and every other right and remedy given hereunder or now or hereafter existing at law or in equity or otherwise. The assertion or employment of any right or remedy hereunder, or otherwise, shall not prevent the concurrent assertion or employment of any other right or remedy.
- (b) For the duration of Indemnitee's service as a director and/or officer of the Company and for a reasonable period of time thereafter, which such period shall be determined by the Company in its sole discretion, the Company shall use commercially reasonable efforts (taking into account the scope and amount of coverage available relative to the cost thereof) to cause to be maintained in effect policies of directors' and officers' liability insurance providing coverage for directors and/or officers of the Company, and, if applicable, that is substantially comparable in scope and amount to that provided by the Company's current policies of directors' and officers' liability insurance. Upon reasonable request, the Company shall provide Indemnitee or his or her counsel with a copy of all directors' and officers' liability insurance applications, binders, policies, declarations, endorsements and other related materials. In all policies of directors' and officers' liability insurance obtained by the Company, Indemnitee shall be named as an insured in such a manner as to provide Indemnitee the same rights and benefits, subject to the same limitations, as are accorded to the Company's directors and officers most favorably insured by such policy. Notwithstanding the foregoing, (a) the Company may, but shall not be required to, create a trust fund, grant a security interest or use other means, including, without limitation, a letter of credit, to ensure the payment of such amounts as may be necessary to satisfy its obligations to indemnify and advance expenses pursuant to this Agreement and (b) in renewing or seeking to renew any insurance hereunder, the Company will

13

not be required to expend more than two (2) times the premium amount of the immediately preceding policy period (equitably adjusted if necessary to reflect differences in policy periods).

- (c) To the extent that the Company maintains an insurance policy or policies providing liability insurance for directors, officers, employees or agents of the Enterprise, Indemnitee shall be covered by such policy or policies in accordance with its or their terms to the maximum extent of the coverage available for any such director, officer, employee or agent under such policy or policies. If, at the time of the receipt of a notice of a claim pursuant to the terms hereof, the Company has director and officer liability insurance in effect, the Company shall give prompt notice of such claim or of the commencement of a Proceeding, as the case may be, to the insurers in accordance with the procedures set forth in the respective policies. The Company shall thereafter take all necessary or desirable action to cause such insurers to pay, on behalf of the Indemnitee, all amounts payable as a result of such Proceeding in accordance with the terms of such policies.
- (d) In the event of any payment under this Agreement, the Company shall be subrogated to the extent of such payment to all of the rights of recovery of Indemnitee, who shall execute all papers required and take all action necessary to secure such rights, including execution of such documents as are necessary to enable the Company to bring suit to enforce such rights.
- (e) The Company shall not be liable under this Agreement to make any payment of amounts otherwise indemnifiable (or for which advancement is provided hereunder) hereunder if and to the extent that Indemnitee has otherwise actually received such payment under any insurance policy, contract, agreement or otherwise.
- Section 16. <u>Duration of Agreement</u>. This Agreement shall continue until and terminate upon the later of: (a) ten (10) years after the date that Indemnitee shall have ceased to serve as a director or officer of the Company, as applicable, or, at the request of the Company, as a director, officer, employee, agent or fiduciary of another corporation, partnership, joint venture, trust or other enterprise, as applicable, or (b) one (1) year after the final termination of any Proceeding then pending in respect of which Indemnitee is granted rights of indemnification or advancement of Expenses hereunder and of any proceeding commenced (including any appeal thereof) by Indemnitee pursuant to Section 14 of this Agreement relating thereto. The indemnification and advancement of expenses rights provided by or granted pursuant to this Agreement shall be binding upon and be enforceable by the parties hereto and their respective successors and assigns (including any direct or indirect successor by purchase, merger, consolidation or otherwise to all or substantially all of the business or assets of the Company), shall continue as to an Indemnitee who has ceased to be a director, officer, employee or agent of the Company or of any other Enterprise, and shall inure to the benefit of Indemnitee and his or her spouse, assigns, heirs, devisees, executors and administrators and other legal representatives. The Company shall require and shall cause any successor (whether direct or indirect by purchase, merger, consolidation or otherwise) to all or substantially all of the business or assets of the Company to, by written agreement, expressly assume and agree to perform this Agreement in the same manner and to the same extent that the Company would be required to perform if no such succession had taken place.

Section 17. Severability. If any provision or provisions of this Agreement shall be held to be invaremaining provisions of this Agreement (including without limitation, each portion of any Section of this Agreement illegal or unenforceable) shall not in any way be affected or impaired thereby and shall remain enforceable to the necessary to conform to applicable law and to give the maximum effect to the intent of the parties hereto; and (c) of any Section of this Agreement containing any such provision held to be invalid, illegal or unenforceable, that is thereby.	fullest extent permitted by law; (b) such provision or provisions shall be deemed reformed to the extent to the fullest extent possible, the provisions of this Agreement (including, without limitation, each portion				
Section 18. <u>Enforcement</u> .					
(a) The Company expressly confirms and agrees that it has entered into this Agreement and assumed the obligations imposed on it hereby in order to induce Indemnitee to serve as a director or officer of the Company, and the Company acknowledges that Indemnitee is relying upon this Agreement in serving as a director or officer of the Company.					
(b) This Agreement constitutes the entire agreement between the parties hereto with respect to the between the parties hereto with respect to the subject matter hereof; provided, however, that this Agreement is a officers' insurance maintained by the Company and applicable law, and shall not be deemed a substitute therefor,					
Section 19. <u>Modification and Waiver</u> . No supplement, modification or amendment of this Agreement shall be deemed or shall constitute a waiver of any other provisions of this Agreement nor shall	nent shall be binding unless executed in writing by the parties hereto. No waiver of any of the provisions l any waiver constitute a continuing waiver.				
Section 20. <u>Notice by Indemnitee</u> . Indemnitee agrees promptly to notify the Company in writing document relating to any Proceeding or matter which may be subject to indemnification or advancement of Exper of any obligation which it may have to the Indemnitee under this Agreement or otherwise.	upon being served with any summons, citation, subpoena, complaint, indictment, information or other uses covered hereunder. The failure of Indemnitee to so notify the Company shall not relieve the Company				
Section 21. <u>Notices.</u> All notices, requests, demands and other communications under this Agreen for by the party to whom said notice or other communication shall have been directed, (b) mailed by certified or r (c) mailed by reputable overnight courier and receipted for by the party to whom said notice or other communications.					
15					
or (d) sent by facsimile transmission, with receipt of oral confirmation that such transmission has been received:					
(a) If to Indemnitee, at the address indicated on the signature page of this Agreement, or such other	r address as Indemnitee shall provide to the Company.				
(b) If to the Company to:					
RestorGenex Corporation					
2150 East Lake Cook Road, Suite 750 Buffalo Grove, Illinois 60089 Attention: Chief Executive Officer					
or to any other address as may have been furnished to Indemnitee by the Company.					
Section 22. <u>Contribution</u> . To the fullest extent permissible under applicable law, if the indemnificin lieu of indemnifying Indemnitee, shall contribute to the amount incurred by or on behalf of Indemnitee, whethe Expenses, in connection with any claim relating to an indemnifiable event under this Agreement, in such proporti reflect (i) the relative benefits received by the Company and Indemnitee as a result of the event(s) and/or transaction officers, employees and agents) and Indemnitee in connection with such event(s) and/or transaction(s).	on as is deemed fair and reasonable in light of all of the circumstances of such Proceeding in order to				
Section 23. <u>Applicable Law and Consent to Jurisdiction</u> . This Agreement and the legal relations a State of Delaware, without regard to its conflict of laws rules. Except with respect to any arbitration commenced irrevocably and unconditionally (i) agree that any action or proceeding arising out of or in connection with this A not in any other state or federal court in the United States of America or any court in any other country, (ii) conservaining out of or in connection with this Agreement, (iii) appoint, to the extent such party is not otherwise subject Dupont Hwy, in the City of Dover, County of Kent, 19901 as its agent in the State of Delaware for acceptance of and validity as if served upon such party personally within the State of Delaware, (iv) waive any objection to the plead or to make, any claim that any such action or proceeding brought in the Delaware Court has been brought in	greement shall be brought only in the Chancery Court of the State of Delaware (the "Delaware Court"), and nt to submit to the exclusive jurisdiction of the Delaware Court for purposes of any action or proceeding to service of process in the State of Delaware, irrevocably National Corporate Research, Ltd., 615 South legal process in connection with any such action or proceeding against such party with the same legal force laying of venue of any such action or proceeding in the Delaware Court, and (v) waive, and agree not to				
16					
Section 24. <u>Identical Counterparts</u> . This Agreement may be executed in one or more counterparts one and the same Agreement. Only one such counterpart signed by the party against whom enforceability is sough	s, each of which shall for all purposes be deemed to be an original but all of which together shall constitute the three to be produced to evidence the existence of this Agreement.				
Section 25. <u>Miscellaneous</u> . The headings of this Agreement are inserted for convenience only and shall not be deemed to constitute part of this Agreement or to affect the construction thereof.					
[Remainder of page intentionally left blank]					
17					
IN WITNESS WHEREOF, the parties have caused this Agreement to be signed as of the day and year fi	rst above written.				
RESTORGENEX CORPORATION	INDEMNITEE				
Ву:					
Name: Title:	Name: Address:				

RESTORGENEX CORPORATION 2015 EQUITY INCENTIVE PLAN

Purpose of Plan.

The purpose of the RestorGenex Corporation 2015 Equity Incentive Plan (the "Plan") is to advance the interests of RestorGenex Corporation (the "Company") and its stockholders by enabling the Company and its Subsidiaries to attract and retain qualified individuals to perform services for the Company and its Subsidiaries, providing incentive compensation for such individuals that is linked to the growth and profitability of the Company and increases in stockholder value, and aligning the interests of such individuals with the interests of its stockholders through opportunities for equity participation in the Company.

Definitions.

The following terms will have the meanings set forth below, unless the context clearly otherwise requires. Terms defined elsewhere in the Plan will have the same meaning throughout the Plan.

- 2.1 "Adverse Action" means any action by a Participant that the Committee, in its sole discretion, determines to be injurious, detrimental, prejudicial or adverse to the interests of the Company or any Subsidiary, including: (a) disclosing confidential information of the Company or any Subsidiary to any person not authorized by the Company or Subsidiary to receive it, (b) engaging, directly or indirectly, in any commercial activity that in the judgment of the Committee competes with the business of the Company or any Subsidiary or (c) interfering with the relationships of the Company or any Subsidiary and their respective employees, independent contractors, customers, prospective customers and vendors.
 - 2.2 "Board" means the Board of Directors of the Company.
- 2.3 "Broker Exercise Notice" means a written notice pursuant to which a Participant, upon exercise of an Option, irrevocably instructs a broker or dealer to sell a sufficient number of shares of Common Stock or loan a sufficient amount of money to pay all or a portion of the exercise price of the Option and/or any related withholding tax obligations and remit such sums to the Company and directs the Company to deliver shares of Common Stock to be issued upon such exercise directly to such broker or dealer or their nominee.
- 2.4 "Cause" means "cause" as defined in any employment or other agreement or policy applicable to the Participant, or if no such agreement or policy exists, will mean (a) dishonesty, fraud, misrepresentation, embezzlement or deliberate injury or attempted injury, in each case related to the Company or any Subsidiary, (b) any unlawful or criminal activity of a serious nature, (c) any intentional and deliberate breach of a duty or duties that, individually or in the aggregate, are material in relation to the Participant's overall duties, or (d) any material breach by a Participant of any employment, service, consulting, confidentiality, assignment of inventions, non-compete or non-solicitation agreement entered into with the Company or any Subsidiary.
- 2.5 "Change in Control" means an event described in Section 15.1 of the Plan; provided, however, if payment under an Incentive Award that is subject to Section 409A of the Code is triggered by a Change in Control, the term Change in Control will mean a change in the ownership or effective control of the Company, or in the ownership of a substantial portion of the assets of the Company, as such term is defined in Section 409A of the Code.
- 2.6 "Code" means the Internal Revenue Code of 1986, as amended. Any reference to a section of the Code in the Plan will be deemed to include a reference to any applicable rules and regulations thereunder and any successor or amended section of the Code.
 - 2.7 "Committee" means the group of individuals administering the Plan, as provided in Section 3 of the Plan.
- 2.8 "Common Stock" means the common stock of the Company, par value \$0.001 per share, or the number and kind of shares of stock or other securities into which such Common Stock may be changed in accordance with Section 4.3 of the Plan.
 - 2.9 "Company" means RestorGenex Corporation, a Nevada corporation, and any successor thereto as provided in Section 21.7 of the Plan.
- 2.10 "Consultant" means a person engaged to provide consulting or advisory services (other than as an Employee or a Director) to the Company or any Subsidiary that: (a) are not in connection with the offer and sale of the Company's securities in a capital raising transaction and (b) do not directly or indirectly promote or maintain a market for the Company's securities.
 - 2.11 "<u>Director</u>" means a member of the Board.
- 2.12 "Disability" means the disability of the Participant such as would entitle the Participant to receive disability income benefits pursuant to the long-term disability plan of the Company or Subsidiary then covering the Participant or, if no such plan exists or is applicable to the Participant, the permanent and total disability of the Participant within the meaning of Section 22(e)(3) of the Code; provided, however, that if distribution of an Incentive Award subject to Section 409A of the Code is triggered by an Eligible Recipient's Disability, such term will mean that the Eligible Recipient is disabled as defined by Section 409A of the Code and the regulations and rulings issued thereunder.
 - 2.13 "Effective Date" means June 17, 2015 or such later date as the Plan is initially approved by the Company's stockholders.
- 2.14 "Eligible Recipients" means (a) for the purposes of granting Incentive Stock Options, all Employees and (b) for the purposes of granting Non-Statutory Stock Options and other Incentive Awards, means all Employees, Directors and Consultants.
- 2.15 "Employee" means any individual performing services for the Company or a Subsidiary and designated as an employee of the Company or a Subsidiary on the payroll records thereof. An Employee will not include any individual during any period he or she is classified or treated by the Company or Subsidiary as an independent contractor, a consultant, or any employee of an employment, consulting or temporary agency or any other entity other than the Company or Subsidiary, without regard to whether such individual is subsequently determined to have been, or is subsequently retroactively reclassified as a common-law employee of the Company or Subsidiary during such period. An individual will not cease to be an Employee in the case of: (a) any leave of absence approved by the Company, or (b) transfers between locations of the Company or between the Company or any Subsidiaries. For purposes of Incentive Stock Options, no such leave may exceed ninety (90) days, unless reemployment upon expiration of such leave is guaranteed by statute or contract. If reemployment upon expiration of a leave of absence approved by the Company or a Subsidiary, as applicable, is not so guaranteed, then three (3) months following the ninety-first (91st) day of such leave, any Incentive Stock Option held by a Participant will cease to be treated as an Incentive Stock Option and will be treated for tax purposes as a Non-Statutory

2

Stock Option. Neither service as a Director nor payment of a Director's fee by the Company will be sufficient to constitute "employment" by the Company.

- 2.16 "Exchange Act" means the Securities Exchange Act of 1934, as amended. Any reference to a section of the Exchange Act in the Plan will be deemed to include a reference to any applicable rules and regulations thereunder and any successor or amended section of the Exchange Act.
- 2.17 "Fair Market Value" means, with respect to the Common Stock, as of any date: (a) the closing sale price of the Common Stock as of such date at the end of the regular trading session, as reported by The NASDAQ Stock Market, The New York Stock Exchange, NYSE MKT LLC or any national securities exchange on which the Common Stock is then listed (or, if no shares were traded on such date, as of the next the end of the regular trading session, as reported by OTC Bulletin Board or the OTC Markets Group Inc. or other comparable service (or, if no shares were traded or quoted on such date, as of the next preceding date on which there was such a trade or quote); or (c) if the Common Stock is not so listed or reported, such price as the Committee determines in good faith, in the exercise of its reasonable discretion, and consistent with the definition of "fair market value" under Section 409A of the Code and in conformity with generally accepted accounting principles in the United States. If determined by the Committee, such determination will be final, conclusive and binding for all purposes and on all persons, including the Company, the Stockholders of the Common Stock that is made in good faith.
 - 2.18 "Full Value Award" means an Incentive Award other than in the form of an Option or Stock Appreciation Right, and which is settled by the issuance of shares of Common Stock.
 - 2.19 "Grant Date" means the date an Incentive Award is granted to a Participant pursuant to the Plan and as determined pursuant to Section 5 of the Plan.
- 2.20 "Incentive Award" means an Option, Stock Appreciation Right, Restricted Stock Award, Restricted Stock Unit, Performance Award, Stock Bonus or Other Stock-Based Award granted to an Eligible Recipient pursuant to the Plan.
- 2.21 "Incentive Award Agreement" means either: (a) a written or electronic (as provided in Section 21.9) agreement entered into by the Company and a Participant setting forth the terms and provisions applicable to an Incentive Award granted under the Plan, including any amendment or modification thereof, or (b) a written or electronic (as provided in Section 21.9) statement issued by the Company to a Participant describing the terms and provisions of such an Incentive Award, including any amendment or modification thereof. The Committee may provide for the use of electronic, internet or other non-paper Incentive Award Agreements, and the use of electronic, internet or other non-paper means for the acceptance thereof and actions thereunder by a Participant.
- 2.22 "Incentive Stock Option" means a right to purchase shares of Common Stock granted to an Employee pursuant to Section 6 of the Plan that is designated as and intended to meet the requirements of an "incentive stock option" within the meaning of Section 422 of the Code.

2.23 "Non-Statutory Stock Option" means a right to purchase shares of Common Stock granted to an Eligible Recipient pursuant to Section 6 of the Plan that is not intended to meet the requirements of or does not qualify as an Incentive Stock Option.

- 2.24 "Option" means an Incentive Stock Option or a Non-Statutory Stock Option.
- 2.25 "Other Stock-Based Award" means an equity-based or equity-related Incentive Award not otherwise described by the terms of the Plan, granted pursuant to Section 11 of the Plan.
- 2.26 "Participant" means an Eligible Recipient who receives one or more Incentive Awards under the Plan.
- 2.27 "Performance Award" means a right granted to an Eligible Recipient pursuant to Section 9 of the Plan to receive an amount of cash, a number of shares of Common Stock, or a combination of both, contingent upon and the value of which at the time it is payable is determined as a function of the extent of the achievement of specified performance objectives during a specified period. A Performance Award is also commonly referred to as a "performance unit" or "performance share."
- 2.28 "Previously Acquired Shares" means shares of Common Stock that are already owned by the Participant or, with respect to any Incentive Award, that are to be issued to the Participant upon the grant, exercise, vesting or settlement of such Incentive Award.
- 2.29 "Restricted Stock Award" means an award of shares of Common Stock granted to an Eligible Recipient pursuant to Section 8 of the Plan that is subject to restrictions on transferability and a risk of forfeiture.
- 2.30 "Restricted Stock Unit" means an award denominated in shares of Common Stock granted to an Eligible Recipient pursuant to Section 8 of the Plan to receive an amount of cash, a number of shares of Common Stock, or a combination of both, contingent upon the achievement of specified performance objectives or that the Participant remain in the continuous employment or service with the Company for a certain period or other conditions.
- 2.31 "Retirement" means unless otherwise defined in the Incentive Award Agreement or in a written employment, services or other agreement between the Participant and the Company or a Subsidiary, means "Retirement" as defined from time to time for purposes of the Plan by the Committee or by the Company's chief human resources officer or other person performing that function or, if not so defined, means voluntary termination of employment or service by the Participant on or after the date the Participant reaches age fifty-five (55) with the present intention to leave the Company's industry or to leave the general workforce and completion of at least ten (10) years of continuous service with the Company or a Subsidiary.
- 2.32 "Securities Act" means the Securities Act of 1933, as amended. Any reference to a section of the Securities Act in the Plan will be deemed to include a reference to any applicable rules and regulations thereunder and any successor or amended section of the Securities Act.
- 2.33 "Stock Appreciation Right" means a right granted to an Eligible Recipient pursuant to Section 7 of the Plan to receive a payment from the Company, in the form of shares of Common Stock, cash or a combination of both, equal to the difference between the Fair Market Value of one or more shares of Common Stock and a specified exercise price of such shares.
 - 2.34 "Stock Bonus" means an award of shares of Common Stock granted to an Eligible Recipient pursuant to Section 10 of the Plan.
 - 2.35 "Subsidiary" means any entity that is directly or indirectly controlled by the Company or any entity in which the Company has a significant equity interest, as determined by the Committee,

4

provided the Company has a "controlling interest" in the Subsidiary as defined in Treas. Reg. Sec. 1.409A-1(b)(5)(iii)(E)(1).

2.36 "Tax Date" means the date any withholding tax obligation arises under the Code for a Participant with respect to an Incentive Award.

Plan Administration.

- 3.1 The Committee. The Plan will be administered by the Board or by a committee of the Board. So long as the Company has a class of its equity securities registered under Section 12 of the Exchange Act, any committee administering the Plan will consist solely of two or more members of the Board who are "non-employee directors" within the meaning of Rule 16b-3 under the Exchange Act and who are "independent directors" as required by the Listing Rules of The NASDAQ Stock Market (or other applicable exchange or market on which the Company's Common Stock may be traded or quoted). Such a committee, if established, will act by majority approval of the members (but may also take action by the written consent of all of the members of such committee), and a majority of the members of such a committee will constitute a quorum. As used in the Plan, "Committee" will refer to the Board or to such a committee, if established.
- 3.2 Authority of the Committee. In accordance with and subject to the provisions of the Plan, the Committee will have the authority to determine all provisions of Incentive Awards as the Committee may deem necessary or desirable and as consistent with the terms of the Plan, including the following: (a) the Eligible Recipients to be selected as Participants; (b) the nature and extent of the Incentive Awards to be made to each Participant (including the number of shares of Common Stock to be subject to each Incentive Award, any exercise price, the manner in which Incentive Awards will vest or become exercisable and whether Incentive Awards will be granted in tandem with other Incentive Awards) and the form of Incentive Award Agreement; (c) the time or times when Incentive Awards will be granted; (d) the duration of each Incentive Award; and (e) the restrictions and other conditions to which the payment or vesting of Incentive Awards may be subject. In addition, the Committee will have the authority to: (i) interpret, administer, reconcile any inconsistency in, correct any defect in and/or supply any omission in the Plan and any instrument or agreement relating to, or Incentive Award granted under, the Plan; (ii) establish, amend, suspend or waive any rules and regulations and appoint such agents as the Committee may deem appropriate for the proper administration of the Plan; and (iii) make any other determination and take any other action that the Committee may deem necessary or desirable for the administration of the Plan. In addition, the Committee will have the authority under the Plan in its sole discretion to pay the economic value of any Incentive Award in the form of cash, Common Stock or any combination of both. The Committee may exercise its duties, power and authority under the Plan in its sole discretion without the consent of any Participant or other party, unless the Plan specifically provides otherwise. The Committee will not be obligated to treat Participants or Eligible Recipients uniformly, and d
- 3.3 <u>Delegation</u>. Except to the extent prohibited by applicable law or the applicable rules and regulations of any securities exchange or inter-dealer quotation system on which the securities of the Company are listed or traded, the Committee may allocate all or any portion of its responsibilities and powers to any one or more of its members and may delegate all or any part of its responsibilities and

5

powers to any person or persons selected by it. Any such allocation or delegation may be revoked by the Committee at any time. Without limiting the generality of the foregoing and to the extent consistent with the applicable corporate law of the Company's jurisdiction of incorporation, the Committee may, by resolution, authorize one or more Directors or one or more officers of the Company to do one or both of the following on the same basis as can the Committee: (a) designate Eligible Recipients to be recipients of Incentive Awards pursuant to the Plan; and (b) determine the size of any such Incentive Awards; provided, however, that (x) the Committee will not delegate such responsibilities to any such Director(s) or officer(s) for any Incentive Awards granted to an Eligible Recipient who is subject to the reporting and liability provisions of Section 16 under the Exchange Act; (y) the resolution providing such authorization will set forth the type of Incentive Awards and total number of each type of Incentive Awards such Director(s) or officer(s) may grant; and (z) such Director(s) or officer(s) will report periodically to the Committee regarding the nature and scope of the Incentive Awards granted pursuant to the authority delegated.

- 3.4 No Re-Pricing. Notwithstanding any other provision of the Plan other than Section 4.3, the Committee may not, without prior approval of the Company's stockholders, seek to effect any re-pricing of any previously granted, "underwater" Option or Stock Appreciation Right by: (a) amending or modifying the terms of the Option or Stock Appreciation Right to lower the exercise price; (b) canceling the underwater Option or Stock Appreciation Right in exchange for (i) cash; (ii) replacement Options or Stock Appreciation Rights having a lower exercise price; or (iii) other Incentive Awards; or (c) repurchasing the underwater Options or Stock Appreciation Rights and granting new Incentive Awards under the Plan. For purposes of this Section 3.4, Options and Stock Appreciation Rights will be deemed to be "underwater" at any time when the Fair Market Value of the Common Stock is less than the exercise price of the Option or Stock Appreciation Right.
- 3.5 Participants Based Outside the United States. In addition to the authority of the Committee under Section 3.2 of the Plan and notwithstanding any other provision of the Plan, the Committee may, in its sole discretion, amend the terms of the Plan or Incentive Awards with respect to Participants resident outside of the United States or employed by a non-U.S. Subsidiary in order to comply with local legal requirements, to otherwise protect the Company's or Subsidiary's interests, or to meet objectives of the Plan, and may, where appropriate, establish one or more sub-plans (including the adoption of any required rules and regulations) for the purposes of qualifying for preferred tax treatment under foreign tax laws. The Committee will have no authority, however, to take action pursuant to this Section 3.5 of the Plan: (a) to reserve shares or grant Incentive Awards in excess of the limitations provided in Section 4.1 of the Plan; (b) to effect any re-pricing in violation of Section 3.4 of the Plan; (c) to grant Options or Stock Appreciation Rights having an exercise price less than 100% of the Fair Market Value of one share of Common Stock on the Grant Date in violation of Section 6.3 or 7.3 of the Plan, as the case may be; or (d) for which stockholder approval would then be required pursuant to Section 422 of the Code or the Listing Rules of The NASDAQ Stock Market (or other applicable exchange or market on which the Company's Common Stock may be traded or quoted).

4. <u>Shares Available for Issuance</u>.

4.1 <u>Maximum Number of Shares Available; Certain Restrictions on Awards</u>. Subject to adjustment as provided in Section 4.3 of the Plan, the maximum number of shares of Common Stock that will be available for issuance under the Plan will be the sum of:

(b) the number of shares issued or Incentive Awards granted under the Plan in connection with the settlement, assumption or substitution of outstanding awards or obligations to

6

grant future awards as a condition of the Company and/or any Subsidiary(ies) acquiring, merging or consolidating with another entity; and

(c) the number of shares that are unallocated and available for grant under a stock plan assumed by the Company or any Subsidiary(ies) in connection with the merger, consolidation, or acquisition of another entity by the Company and/or any of its Subsidiaries, based on the applicable exchange ratio and other transaction terms, but only to the extent that such shares may be utilized by the Company or its Subsidiaries following the transaction pursuant to the rules and regulations of The NASDAQ Stock Market (or other applicable exchange or market on which the Company's Common Stock may be traded or quoted).

The shares available for issuance under the Plan may, at the election of the Committee, be either treasury shares or shares authorized but unissued, and, if treasury shares are used, all references in the Plan to the issuance of shares will, for corporate law purposes, be deemed to mean the transfer of shares from treasury.

Notwithstanding any other provisions of the Plan to the contrary, (i) no more than 2,500,000 shares of Common Stock may be issued pursuant to the exercise of Incentive Stock Options granted under the Plan; and (ii) no more than 1,000,000 shares of Common Stock may be issued or issuable under the Plan in connection with the grant of Full Value Awards. All of the foregoing share limits are subject, in each case, to adjustment as provided in Section 4.3 of the Plan. Incentive Stock Options issued as a result of the Company's assumption or substitution of like awards issued by any acquired, merged or consolidated entity pursuant to applicable provisions of the Code will not count towards the limit in clause (i).

The maximum aggregate number of shares of Common Stock subject to Incentive Awards to any one Director who is not an Employee in any one calendar year may not exceed 275,000; provided, that such limit shall not apply to any election of a Director to receive shares of Common Stock in lieu of all or a portion of any annual Board, chair and other retainers and any meeting fees otherwise payable in cash.

4.2 Accounting for Incentive Awards. Shares of Common Stock that are issued under the Plan or that are subject to outstanding Incentive Awards will be applied to reduce the maximum number of shares of Common Stock remaining available for issuance under the Plan. All shares so subtracted from the amount available under the Plan with respect to an Incentive Award that lapses, expires, is forfeited (including issued shares forfeited under a Restricted Stock Award) or for any reason is terminated unexercised or unvested or is settled or paid in cash or any form other than shares of Common Stock will automatically again become available for issuance under the Plan; provided, however, that the full number of shares of Common Stock subject to a Stock Appreciation Right granted that is settled by the issuance of Store Common Stock will be counted against the shares of Common Stock authorized for issuance under the Plan, regardless of the number of shares actually issued upon settlement of such Stock Appreciation Right. Furthermore, any shares of Common Stock withheld to satisfy tax withholding obligations on Incentive Awards issued under the Plan, any shares of Common Stock withheld to pay the exercise price of Incentive Awards under the Plan and any shares of Common Stock not issued or delivered as a result of the "net exercise" of an outstanding Option pursuant to Section 6.5 of the Plan or settlement of a Stock Appreciation Right in shares of Common Stock pursuant to Section 7.7 of the Plan will be counted against the shares of Common Stock authorized for issuance under the Plan and will not be available again for grant under the Plan. Any shares of Common Stock pursuant to Section 7.7 of the Plan will be counted against the shares of Common Stock authorized for issuance under the Plan and will not increase the number of shares available for future grant of Incentive Awards. Any shares of Common Stock repurchased by the Company on the open market using the proceeds from the exercise of an Incentive Award will not in

7

issuance of shares of Common Stock, or are settled in cash in lieu of shares, or are exchanged with the Committee's permission, prior to the issuance of shares, for Incentive Awards not involving shares, will be available again for grant under the Plan.

4.3 Adjustments to Shares and Incentive Awards.

- (a) In the event of any reorganization, merger, consolidation, recapitalization, liquidation, reclassification, stock dividend, stock split, combination of shares, rights offering, divestiture or extraordinary dividend (including a spin-off) or any other similar change in the corporate structure or shares of the Company, the Committee (or, if the Company is not the surviving corporation in any such transaction, the board of directors of the surviving corporation) will make appropriate adjustment (which determination will be conclusive) as to the number and kind of securities or other property (including cash) available for issuance or payment under the Plan, including the sub-limits set forth in Section 4.1 of the Plan and, in order to prevent dilution or enlargement of the rights of Participants, (a) the number and kind of securities or other property (including cash) subject to outstanding Incentive Awards, and (b) the exercise price of outstanding Incentive Awards. The determination of the Committee as to the foregoing adjustments, if any, will be final, conclusive and binding on Participants under the Plan.
- (b) Notwithstanding anything else in the Plan to the contrary, without affecting the number of shares of Common Stock reserved or available under the Plan, including the sub-limits in Section 4.1 of the Plan, the Committee may authorize the issuance or assumption of benefits under the Plan in connection with any merger, consolidation, acquisition of property or stock or reorganization upon such terms and conditions as it may deem appropriate, subject to compliance with the rules under Sections 422 and 424 of the Code, as and where applicable.

Participation

Participants in the Plan will be those Eligible Recipients who, in the judgment of the Committee, have contributed, are contributing or are expected to contribute to the achievement of objectives of the Company or its Subsidiaries. Eligible Recipients may be granted from time to time one or more Incentive Awards, singly or in combination or in tandem with other Incentive Awards, as may be determined by the Committee in its sole discretion. Incentive Awards will be deemed to be granted as of the date specified in the grant resolution of the Committee, which date will be the Grant Date of any related Incentive Award Agreement with the Participant.

6. Options

6.1 Grant. An Eligible Recipient may be granted one or more Options under the Plan, and such Options will be subject to such terms and conditions, consistent with the other provisions of the Plan, as may be determined by the Committee in its sole discretion. The Committee may designate whether an Option is to be considered an Incentive Stock Option or a Non-Statutory Stock Option. An Option will be an Incentive Stock Option only if the Eligible Recipient receiving the Option is an Employee. To the extent that any Incentive Stock Option (or portion thereof) granted under the Plan ceases for any reason to qualify as an "incentive stock option" for purposes of Section 422 of the Code, such Incentive Stock Option (or portion thereof) will continue to be outstanding for purposes of the Plan but will thereafter be deemed to be a Non-Statutory Stock Option. Options may be granted to an Eligible Recipient for services provided to a Subsidiary only if, with respect to such Eligible Recipient, the underlying shares of Common Stock constitute "service recipient stock" within the meaning of Treas. Reg. Section 1.409A-1(b)(5)(iii).

8

- 6.2 <u>Incentive Award Agreement.</u> Each Option grant will be evidenced by an Incentive Award Agreement that will specify the exercise price of the Option, the maximum duration of the Option, the number of shares of Common Stock to which the Option pertains, the conditions upon which an Option will become vested and exercisable, and such other provisions as the Committee will determine which are not inconsistent with the terms of the Plan. The Incentive Award Agreement also will specify whether the Option is intended to be an Incentive Stock Option or a Non-Statutory Stock Option.
- 6.3 Exercise Price. The per share price to be paid by a Participant upon exercise of an Option granted pursuant to this Section 6 will be determined by the Committee in its sole discretion at the time of the Option grant; provided, however, that such price will not be less than 100% of the Fair Market Value of one share of Common Stock on the date of grant (or 110% of the Fair Market Value of one share of Common Stock on the date of grant of an Incentive Stock Option if, at the time the Incentive Stock Option is granted, the Participant owns, directly or indirectly, more than 10% of the total combined voting power of all classes of stock of the Company or any parent or subsidiary corporation of the Company). Notwithstanding the foregoing, to the extent that Options are granted under the Plan as a result of the Company's assumption or substitution of options issued by any acquired, merged or consolidated entity, the exercise price for such Options will be the price determined by the Committee pursuant to the conversion terms applicable to the transaction.
- Exercisability and Duration. An Option will become exercisable at such times and in such installments and upon such terms and conditions as may be determined by the Committee in its sole discretion at the time of grant, including (a) the achievement of one or more specified performance objectives; and/or that (b) the Participant remain in the continuous employment or service with the Company or a Subsidiary for a certain period; provided, however, that no Option may be exercisable after ten (10) years from the Grant Date (five years from the Grant Date in the case of an Incentive Stock Option if, at the time the Incentive Stock Option is granted, the Participant owns, directly or indirectly, more than 10% of the total combined voting power of all classes of stock of the Company or any parent or subsidiary corporation of the Company). Notwithstanding the foregoing, if the Option (other than in the case of an Incentive Stock Option) would expire at a time when trading in the shares of Common Stock is prohibited by the Company's insider trading policy (or Company-imposed "blackout period"), then the term of the Option will be automatically extended until the thirtieth (30th) day following the expiration of such prohibition or if the exercise of an Option that is exercisable in accordance with its terms is otherwise prevented by the provisions of Section 17 of the Plan, then the Option will remain exercisable until thirty (30) days after the date such exercise first would no longer be prevented by such provisions, but in any event no later than the expiration date of such Option.

6.5 <u>Payment of Exercise Price</u>.

- (a) The total purchase price of the shares of Common Stock to be purchased upon exercise of an Option must be paid entirely in cash (including check, bank draft or money order); provided, however, that the Committee, in its sole discretion and upon terms and conditions established by the Committee, may allow such payments to be made, in whole or in part, by (i) tender of a Broker Exercise Notice; (ii) tender, either by actual delivery or attestation as to ownership, of Previously Acquired Shares that are acceptable to the Committee; (iii) a "net exercise" of the Option (as further described in paragraph (b), below); (iv) a combination of such methods; or (v) any other method approved or accepted by the Committee in its sole discretion.
- (b) In the case of a "net exercise" of an Option, the Company will not require a payment of the exercise price of the Option from the Participant but will reduce the number of shares of Common Stock issued upon the exercise by the largest number of whole shares that has a

Fair Market Value on the exercise date that does not exceed the aggregate exercise price for the shares exercised under this method. Shares of Common Stock will no longer be outstanding under an Option (and will therefore not thereafter be exerciseable) following the exercise of such Option to the extent of (i) shares used to pay the exercise price of an Option under the "net exercise," (ii) shares actually delivered to the Participant as a result of such exercise and (iii) any shares withheld for purposes of tax withholding pursuant to Section 14.1 of the Plan.

- (c) For purposes of such payment, Previously Acquired Shares tendered or covered by an attestation will be valued at their Fair Market Value on the exercise date of the Option.
- 6.6 Manner of Exercise. An Option may be exercised by a Participant in whole or in part from time to time, subject to the conditions contained in the Plan and in the Incentive Award Agreement evidencing such Option, by delivery in person, by facsimile or electronic transmission or through the mail of written notice of exercise to the Company at its principal executive office in Buffalo Grove, Illinois (or to the Company's designee as may be established from time to time by the Company and communicated to Participants) and by paying in full the total exercise price for the shares of Common Stock to be purchased in accordance with Section 6.5 of the Plan.

7. Stock Appreciation Rights.

- 7.1 Grant. An Eligible Recipient may be granted one or more Stock Appreciation Rights under the Plan, and such Stock Appreciation Rights will be subject to such terms and conditions, consistent with the other provisions of the Plan, as may be determined by the Committee in its sole discretion. The Committee will have the sole discretion to determine the form in which payment of the economic value of Stock Appreciation Rights will be made to a Participant (i.e., cash, shares of Common Stock or any combination thereof) or to consent to or disapprove the election by a Participant of the form of such payment. Stock Appreciation Rights may be granted to an Eligible Recipient for services provided to a Subsidiary only if, with respect to such Eligible Recipient, the underlying shares of Common Stock constitute "service recipient stock" within the meaning of Treas. Reg. Section 1.409A-1(b)(5)(iii).
- 7.2 Incentive Award Agreement. Each Stock Appreciation Right will be evidenced by an Incentive Award Agreement that will specify the exercise price of the Stock Appreciation Right, the term of the Stock Appreciation Right, and such other provisions as the Committee will determine which are not inconsistent with the terms of the Plan.
- 7.3 Exercise Price. The exercise price of a Stock Appreciation Right will be determined by the Committee, in its sole discretion, at the Grant Date; provided, however, that such price may not be less than 100% of the Fair Market Value of one share of Common Stock on the Grant Date. Notwithstanding the foregoing, to the extent that Stock Appreciation Rights are granted under the Plan as a result of the Company's assumption or substitution of stock appreciation rights issued by any acquired, merged or consolidated entity, the exercise price for such Stock Appreciation Rights will be the price determined by the Committee pursuant to the conversion terms applicable to the transaction.
- 7.4 <u>Exercisability and Duration.</u> A Stock Appreciation Right will become exercisable at such times and in such installments as may be determined by the Committee in its sole discretion at the time of grant; provided, however, that no Stock Appreciation Right may be exercisable after ten (10) years from its Grant Date. Notwithstanding the foregoing, if the Stock Appreciation Right would expire at a time when trading in the shares of Common Stock is prohibited by the Company's insider trading policy (or Company-imposed "blackout period"), then the term of the Stock Appreciation Right will be automatically extended until the thirtieth (30th) day following the expiration of such prohibition or if the exercise of a

10

Stock Appreciation Right that is exercisable in accordance with its terms is otherwise prevented by the provisions of Section 17 of the Plan, the Stock Appreciation Right will remain exercisable until thirty (30) days after the date such exercise first would no longer be prevented by such provisions, but in any event no later than the expiration date of such Stock Appreciation Right.

- 7.5 Manner of Exercise. A Stock Appreciation Right will be exercised by giving notice in the same manner as for Options, as set forth in Section 6.6 of the Plan, subject to any other terms and conditions consistent with the other provisions of the Plan as may be determined by the Committee in its sole discretion.
 - 7.6 Settlement. Upon the exercise of a Stock Appreciation Right, a Participant will be entitled to receive payment from the Company in an amount determined by multiplying:
 - (a) The excess of the Fair Market Value of a share of Common Stock on the date of exercise over the per share exercise price; by
 - (b) The number of shares of Common Stock with respect to which the Stock Appreciation Right is exercised.
- 7.7 Form of Payment. Payment, if any, with respect to a Stock Appreciation Right settled in accordance with Section 7.6 of the Plan will be made in accordance with the terms of the applicable Incentive Award Agreement, in cash, shares of Common Stock or a combination thereof, as the Committee determines in its sole discretion.

8. Restricted Stock Awards and Restricted Stock Units.

- 8.1 <u>Grant</u>. An Eligible Recipient may be granted one or more Restricted Stock Awards or Restricted Stock Units under the Plan, and such awards will be subject to such terms and conditions, consistent with the other provisions of the Plan, as may be determined by the Committee in its sole discretion. Restricted Stock Units will be similar to Restricted Stock Awards except that no shares of Common Stock are actually awarded to the Participant on the Grant Date of the Restricted Stock Units. Restricted Stock Units will be denominated in shares of Common Stock but paid in cash, shares of Common Stock or a combination of cash and shares of Common Stock as the Committee, in its sole discretion, will determine, and as provided in the Incentive Award Agreement.
- 8.2 Incentive Award Agreement. Each Restricted Stock Award or Restricted Stock Unit will be evidenced by an Incentive Award Agreement that will specify the type of Incentive Award, the period(s) of restriction, the number of shares of restricted Common Stock, or the number of Restricted Stock Units granted, and such other provisions as the Committee will determine which are not inconsistent with the terms of the Plan.
- 8.3 <u>Vesting Requirements and Restrictions</u>. The Committee will impose such restrictions or conditions, not inconsistent with the provisions of the Plan, to the vesting of such Restricted Stock Awards or Restricted Stock Units as it deems appropriate, including (a) the achievement of one or more specified performance objectives; or that (b) the Participant remain in the continuous employment or service with the Company or a Subsidiary for a certain period. If any vesting requirements of a Restricted Stock Award or Restricted Stock Unit are not satisfied, the Restricted Stock Award or Restricted Stock Unit will be forfeited and the shares of Common Stock subject to the Restricted Stock Award will be returned to the Company and no shares of Common Stock or other consideration will be issued with respect to the forfeited Restricted Stock Unit. If the Participant paid any purchase price with respect to such forfeited

11

shares, unless otherwise provided by the Committee in the Incentive Award Agreement evidencing the Restricted Stock Award, the Company will refund to the Participant the lesser of (x) such purchase price and (y) the Fair Market Value of such shares on the date of forfeiture. Notwithstanding the foregoing, Restricted Stock Awards may in the sole discretion of the Committee be granted without any restrictions or conditions.

8.4 Rights as a Shareholder. Except as provided in Sections 8.1, 8.5 and 8.6 of the Plan, upon a Participant becoming the holder of record of shares of Common Stock issued under a Restricted Stock Award pursuant to this Section 8, the Participant will have all voting, dividend, liquidation and other rights with respect to such shares (other than the right to sell or transfer such shares) as if such Participant were a holder of record of shares of unrestricted Common Stock. A Participant will have no voting, dividend, liquidation and other rights with respect to any shares of Common Stock underlying any Restricted Stock Units granted hereunder unless and until the shares of Common Stock are issued under the terms thereof and the Participant becomes the holder of record of such shares.

8.5 Dividends and Distributions.

- (a) Unless the Committee determines otherwise in its sole discretion (either in the Incentive Award Agreement evidencing the Restricted Stock Award at the time of grant or at any time after the grant of the Restricted Stock Award), any dividends or distributions paid with respect to shares of Common Stock subject to the unvested portion of a Restricted Stock Award will be subject to the same restrictions as the shares to which such dividends or distributions relate. The Committee will determine in its sole discretion whether any interest will be paid on such dividends or distributions.
- (b) Unless the Committee determines otherwise in its sole discretion (either in the Incentive Award Agreement evidencing the Restricted Stock Unit at the time of grant or at any time after the grant of the Restricted Sock Unit), any Restricted Stock Unit shall carry with it a right to "dividend equivalents" (as defined in Section 12 of the Plan).
- 8.6 Enforcement of Restrictions. To enforce the restrictions referred to in this Section 8, the Committee may place a legend on the stock certificates or book-entry notation representing Restricted Stock Awards referring to such restrictions and may require the Participant, until the restrictions have lapsed, to keep any stock certificates, together with duly endorsed stock powers, in the custody of the Company or its transfer agent, or to maintain evidence of stock ownership, together with duly endorsed stock powers, in a certificateless book-entry stock account with the Company's transfer agent. Alternatively, Restricted Stock Awards may be held in non-certificated form pursuant to such terms and conditions as the Company may establish with its registrar and transfer agent or any third-party administrator designated by the Company to hold Restricted Stock Awards on behalf of Participants.
- 8.7 <u>Lapse of Restrictions; Settlement.</u> Except as otherwise provided in this Section 8, shares of Common Stock underlying a Restricted Stock Award will become freely transferable by the Participant after all conditions and restrictions applicable to such shares have been satisfied or lapse (including satisfaction of any applicable tax withholding obligations). Upon the vesting of a Restricted Stock Unit, the Restricted Stock Unit will be settled, subject to the terms and conditions of the applicable Incentive Award Agreement, (a) in cash, based upon the Fair Market Value of the vested underlying shares of Common Stock, (b) in shares of Common Stock or (c) a combination thereof, as provided in the Incentive Award Agreement, except to the extent that a Participant has properly elected to defer income that may be attributable to a Restricted Stock Unit under a Company deferred compensation plan or arrangement.

8.8 Section 83(b) Election for Restricted Stock Award. If a Participant makes an election pursuant to Section 83(b) of the Code with respect to a Restricted Stock Award, the Participant must file, within thirty (30) days following the Grant Date of the Restricted Stock Award, a copy of such election with the Company and with the Internal Revenue Service, in accordance with the regulations under Section 83 of the Code.

The Committee may provide in the Incentive Award Agreement that the Restricted Stock Award is conditioned upon the Participant's making or refraining from making an election with respect to the award under Section 83(b) of the Code.

Performance Awards

- 9.1 Grant. An Eligible Recipient may be granted one or more Performance Awards under the Plan, and such awards will be subject to such terms and conditions, consistent with the other provisions of the Plan, as may be determined by the Committee in its sole discretion, including the achievement of one or more specified performance objectives.
- 9.2 Incentive Award Agreement. Each Performance Award will be evidenced by an Incentive Award Agreement that will specify the amount of cash, shares of Common Stock or combination of both to be received by the Participant upon payout of the Performance Award, any performance objectives upon which the Performance Award is subject, any performance period during which any such performance objectives must be achieved and such other provisions as the Committee will determine which are not inconsistent with the terms of the Plan.
- 9.3 <u>Vesting</u>. The Committee may impose such restrictions or conditions, not inconsistent with the provisions of the Plan, to the vesting of such Performance Awards as it deems appropriate, including the achievement of one or more specified performance objectives.
- 9.4 Form and Timing of Performance Award Payment. Subject to the terms of the Plan, after the applicable performance period has ended, the holder of Performance Awards will be entitled to receive payment on the value and number of Performance Awards earned by the Participant over the performance period, to be determined as a function of the extent to which the corresponding performance objectives have been achieved. Payment of earned Performance Awards will be as determined by the Committee and as evidenced in the Incentive Award Agreement. Subject to the terms of the Plan, the Committee, in its sole discretion, may pay earned Performance Awards in the form of cash or in shares of Common Stock (or in a combination thereof) equal to the value of the earned Performance Awards at the close of the applicable performance period. Payment of any Performance Award will be made as soon as practicable after the Committee has determined the extent to which the applicable performance objectives have been achieved and not later than the March 15th immediately following the end of the performance period, or earlier than the January 1st preceding such March 15, except to the extent that a Participant has properly elected to defer payment that may be attributable to a Performance Award under a Company deferred compensation plan or arrangement. The determination of the Committee with respect to the form of payment of Performance Awards will be set forth in the Incentive Award Agreement pertaining to the grant of the award. Any shares of Common Stock issued in payment of earned Performance Awards may be granted subject to any restrictions deemed appropriate by the Committee, including that the Participant remain in the continuous employment or service with the Company or a Subsidiary for a certain period.

Stock Bonuses

An Eligible Recipient may be granted one or more Stock Bonuses under the Plan, and such Stock Bonuses will be subject to such terms and conditions, if any, consistent with the other provisions of the

13

Plan, as may be determined by the Committee in its sole discretion, including the achievement of one or more specified performance objectives.

Other Stock-Based Awards

- 11.1 Other Stock-Based Awards. Subject to such terms and conditions, consistent with the other provisions of the Plan, as may be determined by the Committee in its sole discretion, the Committee may grant Other Stock-Based Awards not otherwise described by the terms of the Plan (including the grant or offer for sale of unrestricted shares of Common Stock) in such amounts and subject to such terms and conditions as the Committee will determine. Such Incentive Awards may involve the transfer of actual shares of Common Stock to Participants or payment in cash or otherwise of amounts based on the value of shares of Common Stock, and may include Incentive Awards designed to comply with or take advantage of the applicable local laws of jurisdictions other than the United States.
- 11.2 <u>Value of Other Stock-Based Awards</u>. Each Other Stock-Based Award will be expressed in terms of shares of Common Stock or units based on shares of Common Stock, as determined by the Committee. The Committee may establish performance objectives in its sole discretion for any Other Stock-Based Award. If the Committee exercises its discretion to establish performance objectives for any such Incentive Awards, the number or value of Other Stock-Based Awards that will be paid out to the Participant will depend on the extent to which the specified performance objectives are met.
- 11.3 Payment of Other Stock-Based Awards. Payment, if any, with respect to an Other Stock-Based Award will be made in accordance with the terms of the Incentive Award and in cash or shares of Common Stock, as the Committee determines, except to the extent that a Participant has properly elected to defer payment that may be attributable to an Other Stock-Based Award under a Company deferred compensation plan or arrangement.

Dividend Equivalents.

Any Participant selected by the Committee may be granted dividend equivalents based on the dividends declared on shares of Common Stock that are subject to any Incentive Award, to be credited as of dividend payment dates, during the period between the Grant Date of the Incentive Award and the date the Incentive Award is exercised, vests, is settled or expires, as determined by the Committee. Such dividend equivalents will be converted to cash or additional shares of Common Stock by such formula and at such time and subject to such limitations as may be determined by the Committee. Notwithstanding the foregoing, the Committee may not grant dividend equivalents based on the dividends declared on shares of Common Stock that are subject to any unvested Incentive Awards, the vesting of which is based on the achievement of performance objectives.

13. Effect of Termination of Employment or Other Service.

- 13.1 <u>Termination Due to Death, Disability or Retirement.</u> Unless otherwise expressly provided by the Committee in its sole discretion in an Incentive Award Agreement, and subject to Sections 13.3, 13.4, 13.5 and 13.6 of the Plan, in the event a Participant's employment or other service with the Company and all Subsidiaries is terminated by reason of death, Disability or Retirement:
 - (a) All outstanding Options and Stock Appreciation Rights then held by the Participant will, to the extent exercisable as of such termination, remain exercisable in full for a period of one year after such termination (but in no event after the expiration date of any such

14

Option or Stock Appreciation Right). Options and Stock Appreciation Rights not exercisable as of such termination will be forfeited and terminate.

- (b) All Restricted Stock Awards then held by the Participant that have not vested as of such termination will be terminated and forfeited; and
- (c) All outstanding but unpaid Restricted Stock Units, Performance Awards, Stock Bonuses and Other Stock-Based Awards then held by the Participant will be terminated and forfeited; provided, however, that with respect to any such Incentive Awards the vesting of which is based on the achievement of specified performance objectives, if a Participant's employment or other service with the Company or any Subsidiary, as the case may be, is terminated by reason of death or Disability prior to the end of the performance period of such Incentive Award, but after the conclusion of a portion of the performance period (but in no event less than one year), the Committee may, in its sole discretion, cause shares of Common Stock to be delivered or payment made with respect to the Participant's Incentive Award, but only if otherwise earned for the entire performance period and only with respect to the portion of the applicable performance period completed at the date of such event, with proration based on full fiscal years only and no shares to be delivered for partial fiscal years. The Committee will consider the provisions of Sections 13.6 and 13.7 of the Plan and will have the discretion to consider any other fact or circumstance in making its decision as to whether to deliver such shares of Common Stock or other payment, including whether the Participant again becomes employed.
- 13.2 <u>Termination for Reasons Other than Death, Disability or Retirement.</u> Unless otherwise expressly provided by the Committee in its sole discretion in an Incentive Award Agreement, and subject to Sections 13.3, 13.4, 13.5 and 13.6 of the Plan, in the event a Participant's employment or other service with the Company and all Subsidiaries is terminated for any reason other than death, Disability or Retirement, or a Participant is in the employment or service with a Subsidiary and the Subsidiary ceases to be a Subsidiary of the Company (unless the Participant continues in the employment or service with the Company or another Subsidiary):
 - (a) All outstanding Options and Stock Appreciation Rights then held by the Participant will, to the extent exercisable as of such termination, remain exercisable in full for a period of three months after such termination (but in no event after the expiration date of any such Option or Stock Appreciation Right). Options and Stock Appreciation Rights not exercisable as of such termination will be forfeited and terminate;
 - (b) All Restricted Stock Awards then held by the Participant that have not vested as of such termination will be terminated and forfeited; and
 - (c) All outstanding but unpaid Restricted Stock Unit, Performance Awards, Stock Bonuses and Other Stock-Based Awards then held by the Participant will be terminated and forfeited.
- 13.3 <u>Modification of Rights Upon Termination</u>. Notwithstanding the other provisions of this Section 13, upon a Participant's termination of employment or other service with the Company and all Subsidiaries, the Committee may, in its sole discretion (which may be exercised at any time on or after the Grant Date, including following such termination), except as provided below, cause Options or Stock

Based Awards then held by such Participant to terminate, vest, settle or become free of restrictions and conditions to payment, as the case may be, following such termination of employment or service, in each case in the manner determined by the Committee; provided, however, that no Option or Stock Appreciation Right may remain exercisable beyond its expiration date and any such action adversely affecting any outstanding Incentive Award will not be effective without the consent of the affected Participant (subject to the right of the Committee to take whatever action it deems appropriate under Sections 4.3, 13.5, 13.6 and 15 of the Plan).

13.4 <u>Determination of Termination of Employment or Other Service</u>

- (a) The change in a Participant's status from that of an Employee to that of a Consultant will, for purposes of the Plan, be deemed to result in a termination of such Participant's employment with the Company and its Subsidiaries, unless the Committee otherwise determines in its sole discretion.
- (b) The change in a Participant's status from that of a Consultant to that of an Employee will not, for purposes of the Plan, be deemed to result in a termination of such Participant's service as a Consultant, and such Participant will thereafter be deemed to be an Employee, in which event such Participant will be governed by the provisions of the Plan relating to termination of employment or service (subject to paragraph (a) above).
- (c) Unless the Committee otherwise determines in its sole discretion, a Participant's employment or other service will, for purposes of the Plan, be deemed to have terminated on the date recorded on the personnel or other records of the Company or the Subsidiary for which the Participant provides employment or other service, as determined by the Committee in its sole discretion based upon such records.
- (d) Notwithstanding the foregoing, if payment of an Incentive Award that is subject to Section 409A of the Code is triggered by a termination of a Participant's employment or other service, such termination must also constitute a "separation from service" within the meaning of Section 409A of the Code, and any change in employment status that constitutes a "separation from service" under Section 409A of the Code will be treated as a termination of employment or service, as the case may be.
- 13.5 Effect of Actions Constituting Cause or Adverse Action. Notwithstanding anything in the Plan to the contrary and in addition to the other rights of the Committee under this Section 13, if a Participant is determined by the Committee, acting in its sole discretion, to have taken any action that would constitute Cause or an Adverse Action during or after the termination of employment or other service with the Company or a Subsidiary, irrespective of whether such action or the Committee's determination occurs before or after termination of such Participant's employment or other service with the Company or any Subsidiary and irrespective of whether or not the Participant was terminated as a result of such Cause or Adverse Action, (a) all rights of the Participant under the Plan and any Incentive Award Agreements evidencing an Incentive Award then held by the Participant will terminate and be forfeited without notice of any kind, and (b) the Committee in its sole discretion will have the authority to rescind the exercise, vesting, settlement or issuance of, or payment in respect of, any Incentive Awards of the Participant that were exercised, vested, settled or issued, or as to which such payment was made, and to require the Participant to pay to the Company, within ten (10) days of receipt from the Company of notice of such rescission, any amount received or the amount of any gain realized as a result of such rescinded exercise, vesting, settlement, issuance or payment (including any dividends paid or other distributions made with respect to any shares of Common Stock subject to any Incentive Award). The Company may

16

defer the exercise of any Option or Stock Appreciation Right for a period of up to six (6) months after receipt of the Participant's written notice of exercise or the issuance of share certificates upon the vesting of any Incentive Award for a period of up to six (6) months after the date of such vesting in order for the Committee to make any determination as to the existence of Cause or an Adverse Action. The Company will be entitled to withhold and deduct from future wages of the Participant (or from other amounts that may be due and owing to the Participant from the Company or a Subsidiary) or make other arrangements for the collection of all amounts necessary to satisfy such payment obligations. Unless otherwise provided by the Committee in an applicable Incentive Award Agreement, this Section 13.5 will not apply to any Participant following a Change in Control.

13.6 Clawback/Forfeiture of Incentive Awards. If the Company is required to prepare an accounting restatement due to the material noncompliance of the Company, as a result of misconduct, with any financial reporting requirement under the securities laws, then any Participant who is one of the individuals subject to automatic forfeiture under Section 304 of the Sarbanes-Oxley Act of 2002 will reimburse the Company for the amount of any Incentive Award received by such individual under the Plan during the 12-month period following the first public issuance or filing with the Securities and Exchange Commission, as the case may be, of the financial document embodying such financial reporting requirement. All Incentive Awards under the Plan will be subject to forfeiture or other penalties pursuant to the Dodd-Frank Wall Street Reform and Consumer Protection Act or any other clawback, forfeiture or recoupment provision required by applicable law or under the requirements of any stock exchange or market upon which the Common Stock is then listed or traded. In addition, all Incentive Awards under the Plan will be subject to any compensation "clawback," forfeiture or recoupment policy that the Committee may adopt from time to time that is applicable by its terms to the Participant and such forfeiture and/or penalty conditions or provisions as determined by the Committee and set forth in the applicable Incentive Award Agreement.

14. Payment of Withholding Taxes.

- General Rules. The Company is entitled to (a) withhold and deduct from future wages of the Participant (or from other amounts that may be due and owing to the Participant from the Company or a Subsidiary), or make other arrangements for the collection of, an amount the Company reasonably determines to be the minimum statutory amount necessary to satisfy any and all federal, foreign, state and local withholding and employment-related tax requirements attributable to an Incentive Award or a disqualifying disposition of stock received upon exercise of an Incentive Stock Option; (b) withhold cash paid or payable or shares of Common Stock from the shares of Common Stock issued or otherwise issuable to the Participant in connection with an Incentive Award; or (c) require the Participant promptly to remit the amount of such withholding to the Company before taking any action, including issuing any shares of Common Stock or paying any cash amounts, with respect to an Incentive Award. Shares of Common Stock issued or otherwise issuable to the Participant in connection with an Incentive Award that gives rise to the tax withholding or bulgation that are withheld for purposes of satisfying the Participant's withholding or employment-related tax obligation, will be valued at their Fair Market Value on the Tax Date. When withholding for taxes is effected under the Plan, it will be withheld only up to the minimum required tax withholding rates or such other rate that will not trigger a negative accounting impact on the Company.
- 14.2 Special Rules. The Committee may, in its sole discretion and upon terms and conditions established by the Committee, permit or require a Participant to satisfy, in whole or in part, any withholding or employment-related tax obligation described in Section 14.1 of the Plan by withholding shares of Common Stock underlying an Incentive Award, by electing to tender, or by attestation as to

17

ownership of, Previously Acquired Shares, by delivery of a Broker Exercise Notice or a combination of such methods. For purposes of satisfying a Participant's withholding or employment-related tax obligation, shares of Common Stock withheld by the Company or Previously Acquired Shares tendered or covered by an attestation will be valued at their Fair Market Value on the Tax Date.

15. Change in Control.

- 15.1 A "<u>Change in Control</u>" will mean the occurrence of any one or more of the following: (i) the accumulation (if over time, in any consecutive twelve (12) month period), whether directly, indirectly, beneficially or of record, by any individual, entity or group (within the meaning of Section 13(d)(3) or 14(d)(2) of the Exchange Act) of 50.1% or more of the shares of the outstanding Common Stock of the Company, whether by merger, consolidation, sale or other transfer of shares of Common Stock (other than a merger or consolidation where the stockholders of the Company prior to the merger or consolidation are the holders of a majority of the voting securities of the entity that survives such merger or consolidation), (ii) a sale of all or substantially all of the assets of the Company or (iii) during any period of twelve (12) consecutive months, the individuals who, at the beginning of such period, constitute the Board, and any new director whose election by the Board or nomination for election by the Company's stockholders was approved by a vote of at least two-thirds (2/3) of the directors then still in office who either were directors at the beginning of the 12-month period or whose election or nomination for election was previously so approved, cease for any reason to constitute at least a majority of the Board; provided, however, that the following acquisitions shall not constitute a Change in Control for the purposes of this Plan: (A) any acquisitions of Common Stock or securities convertible, exercisable or exchangeable into Common Stock directly from the Company or (B) any acquisition of Common Stock or securities convertible, exercisable or exchangeable into Common Stock by any employee benefit plan (or related trust) sponsored by or maintained by the Company. Notwithstanding the foregoing, a Change in Control will only occur if such event constitutes a "Change in Control Event" as defined under Treasury Regulations issued under Code Section 409A.
- 15.2 Acceleration of Vesting. Without limiting the authority of the Committee under Sections 3.2 and 4.3 of the Plan, if a Change in Control of the Company occurs, then, unless otherwise provided by the Committee in its sole discretion either in the Incentive Award Agreement at the time of grant or at any time after the grant of an Incentive Award:
 - (a) all Options and Stock Appreciation Rights will become immediately exercisable in full and will remain exercisable in accordance with their terms;
 - (b) all restrictions and vesting requirements applicable to any Incentive Award based solely on the continued service of the Participant will terminate; and
 - (c) all Incentive Awards the vesting or payment of which are based on specified performance objectives will vest as though such performance objectives were fully achieved at target and will become immediately payable; provided, however, that no Incentive Award that provides for a deferral of compensation within the meaning of Section 409A of the Code will be cashed out upon the occurrence of a Change in Control unless the event or circumstances constituting the Change in Control also constitute a "change in the ownership" of the Company, a "change in the effective control" of the Company or a "change in the ownership of a substantial portion of the assets" of the Company, in each case as determined under Section 409A of the Code.

The treatment of any other Incentive Awards in the event of a Change in Control will be as determined by the Committee in connection with the grant thereof, as reflected in the applicable Incentive Award Agreement.

- Award or at any time after the grant of such an Incentive Award, may determine that any or all outstanding Incentive Award granted under the Plan, whether or not exercisable or vested, as the case may be, will be canceled and terminated and that in connection with such cancellation and termination the holder of such Incentive Award will receive for each share of Common Stock subject to such Incentive Award a cash payment (or the delivery of shares of stock, other securities or a combination of cash, stock and securities with a fair market value (as determined by the Committee in good faith) equivalent to such cash payment) equal to the difference, if any, between the consideration received by stockholders of the Company in respect of a share of Common Stock in connection with such Change in Control and the purchase price per share, if any, under the Incentive Award, multiplied by the number of shares of Common Stock subject to such Incentive Award (or in which such Incentive Award is denominated); provided that if such product is zero (\$0) or less or to the extent that the Incentive Award is not then exercisable, the Incentive Award may be canceled and terminated without payment therefor; provided, however, that no Incentive Award that provides for a deferral of compensation within the meaning of Section 409A of the Code will be cashed out upon the occurrence of a Change in Control unless payment was specified in the Incentive Award Agreement at the time of grant and the event or circumstances constituting the Change in Control also constitute a "change in the ownership" of the Company, a "change in the effective control" of the Company or a "change in the ownership of a substantial portion of the assets" of the Company, in each case as determined under Section 409A of the Code. If any portion of the consideration pursuant to a Change in Control may be received by holders of shares of Common Stock on a contingent or delayed basis, the Committee may, in its sole discretion, determine the fair market value per sh
- 15.4 <u>Limitation on Change in Control Payments</u>. Notwithstanding anything in Section 15.2 or 15.3 of the Plan to the contrary, if, with respect to a Participant, the acceleration of the vesting of an Incentive Award as provided in Section 15.2 of the Plan or the payment of cash in exchange for all or part of an Incentive Award as provided in Section 15.3 of the Plan (which acceleration or payment could be deemed a "payment" within the meaning of Section 280G(b)(2) of the Code), together with any other "payments" that such Participant has the right to receive from the Company or any corporation that is a member of an "affiliated group" (as defined in Section 1504(a) of the Code without regard to Section 1504(b) of the Code) of which the Company is a member, would constitute a "parachute payment" (as defined in Section 280G(b)(2) of the Code), then the "payments" to such Participant pursuant to Section 15.2 or 15.3 of the Plan will be reduced (or acceleration of vesting eliminated) to the largest amount as will result in no portion of such "payments" being subject to the excise tax imposed by Section 4999 of the Code; provided, that such reduction will be made only if the aggregate amount of the payments after such reduction exceeds the difference between (a) the amount of such payments absent such reduction minus (b) the aggregate amount of the excise tax imposed under Section 4999 of the Code attributable to any such excess parachute payments; and provided further that such payments will be reduced (or acceleration of vesting eliminated) in the following order: (i) options with an exercise price above fair market value that

19

have a positive value for purposes of Section 280G of the Code, (ii) pro rata among Incentive Awards that constitute deferred compensation under Section 409A of the Code, and (iii) finally, among the Incentive Awards that are not subject to Section 409A of the Code. Notwithstanding the foregoing sentence, if a Participant is subject to a separate agreement with the Company or an Affiliate or Subsidiary that expressly addresses the potential application of Section 280G or 4999 of the Code, then this Section 15.4 will not apply and any "payments" to a Participant pursuant to Section 15.2 or 15.3 of the Plan will be treated as "payments" arising under such separate agreement.

16. <u>Rights of Eligible Recipients and Participants; Transferability.</u>

- 16.1 <u>Employment or Service</u>. Nothing in the Plan or an Incentive Award Agreement will interfere with or limit in any way the right of the Company or any Subsidiary to terminate the employment or service of any Eligible Recipient or Participant at any time, nor confer upon any Eligible Recipient or Participant any right to continue in the employment or other service with the Company or any Subsidiary.
 - 16.2 No Rights to Awards. No Participant or Eligible Individual will have any claim to be granted any Incentive Award under the Plan.
- 16.3 Rights as a Stockholder. As a holder of Incentive Awards (other than Restricted Stock Awards), a Participant will have no rights as a stockholder unless and until such Incentive Awards are exercised for, settled or paid in the form of, shares of Common Stock and the Participant becomes the holder of record of such shares.

16.4 Restrictions on Transfer.

- (a) Except pursuant to testamentary will or the laws of descent and distribution or as otherwise expressly permitted by subsections (b) and (c) below, no right or interest of any Participant in an Incentive Award prior to the exercise (in the case of Options or Stock Appreciation Rights) or vesting, settlement or issuance (in the case of other Incentive Awards) of such Incentive Award will be assignable or transferable, or subjected to any lien, during the lifetime of the Participant, either voluntarily or involuntarily, directly or indirectly, by operation of law or otherwise.
- (b) A Participant will be entitled to designate a beneficiary to receive an Incentive Award upon such Participant's death, and in the event of such Participant's death, payment of any amounts due under the Plan will be made to, and exercise of any Options or Stock Appreciation Rights (to the extent permitted pursuant to Section 13 of the Plan) may be made by, such beneficiary. If a deceased Participant has failed to designate a beneficiary, or if a beneficiary designated by the Participant fails to survive the Participant, payment of any amounts due under the Plan will be made to, and exercise of any Options or Stock Appreciation Rights (to the extent permitted pursuant to Section 13 of the Plan) may be made by, the Participant's legal representatives, heirs and legatees. If a deceased Participant has designated a beneficiary and such beneficiary survives the Participant but dies before complete payment of all amounts due under the Plan or exercise of all exercisable Options or Stock Appreciation Rights, then such payments will be made to, and the exercise of such Options or Stock Appreciation Rights may be made by, the legal representatives, heirs and legatees of the beneficiary.
- (c) Upon a Participant's request, the Committee may, in its sole discretion, permit a transfer of all or a portion of a Non-Statutory Stock Option, other than for value, to such Participant's child, stepchild, grandchild, parent, stepparent, grandparent, spouse, former spouse,

20

sibling, niece, nephew, mother-in-law, father-in-law, son-in-law, daughter-in-law, brother-in-law, or sister-in-law, any person sharing such Participant's household (other than a tenant or employee), a trust in which any of the foregoing have more than fifty percent of the beneficial interests, a foundation in which any of the foregoing (or the Participant) control the management of assets, and any other entity in which these persons (or the Participant) own more than 50% of the voting interests. Any permitted transferee will remain subject to all the terms and conditions applicable to the Participant prior to the transfer. A permitted transfere may be conditioned upon such requirements as the Committee may, in its sole discretion, determine, including, execution and/or delivery of appropriate acknowledgements, opinion of counsel, or other documents by the transferee.

16.5 <u>Non-Exclusivity of the Plan</u>. Nothing contained in the Plan is intended to modify or rescind any previously approved compensation plans or programs of the Company or create any limitations on the power or authority of the Board to adopt such additional or other compensation arrangements as the Board may deem necessary or desirable.

Securities Law and Other Restrictions.

Notwithstanding any other provision of the Plan or any Incentive Award Agreement entered into pursuant to the Plan, the obligation of the Company to issue any shares of Common Stock under the Plan or settle Incentive Awards in shares of Common Stock or other consideration will be subject to all applicable law, rules and regulations, and to such approvals by governmental agencies as may be required. Notwithstanding the generality of the foregoing and notwithstanding any other provision of the Plan or any Incentive Award Agreement entered into pursuant to the Plan, the Company will not be required to issue any shares of Common Stock under the Plan, and a Participant may not sell, assign, transfer or otherwise dispose of shares of Common Stock issued pursuant to Incentive Awards granted under the Plan, unless (a) there is in effect with respect to such shares a registration statement under the Securities Act and any applicable securities laws of a state or foreign jurisdiction or an exemption from such registration under the Securities Act and applicable state or foreign securities laws, and (b) there has been obtained any other consent, approval or permit from any other U.S. or foreign regulatory body which the Committee, in its sole discretion, deems necessary or advisable. The Company will be under no obligation to register for sale under the Securities Act any of the shares of Common Stock to be offered or sold under the Plan. The Company may condition such issuance, sale or transfer upon the receipt of any representations or agreements from the parties involved, and the placement of any legends on certificates representing shares of Common Stock, as may be deemed necessary or advisable by the Company in order to comply with such securities law or other restrictions.

Deferred Compensation; Compliance with Section 409A.

It is intended that all Incentive Awards issued under the Plan be administered in a manner that will comply with the requirements of Section 409A of the Code, or the requirements of an exception to Section 409A of the Code and the Incentive Award Agreements and the Plan will be construed and administered in a manner that is consistent with and give effect to such intent. The Committee is authorized to adopt rules or regulations deemed necessary or appropriate to qualify for an exception from or to comply with the requirements of Section 409A of the Code (including any transition or grandfather rules relating thereto). Notwithstanding anything in this Section 18 or Section 19 to the contrary, with respect to any Incentive Award subject to Section 409A of the Code, no amendment to or payment under such Incentive Award will be made except and only to the extent permitted under Section 409A of the Code. If any amount is payable with respect to an Incentive Award that is subject to Section 409A of the Code as a result of the Participant's "separation from service" at such time the Participant is a "specified employee" within the

21

meaning of Section 409A of the Code, then no payment will be made, except as permitted under Section 409A of the Code, prior to the first business day after the earlier of (i) the date that is six months after the Participant's separation from service or (ii) the Participant's death.

- 19.1 Generally. Subject to other subsections of this Section 19 and Section 3.4 of the Plan, the Board at any time may suspend or terminate the Plan (or any portion thereof) or terminate any outstanding Incentive Award and the Committee, at any time and from time to time, may amend the Plan or amend or modify the terms of an outstanding Incentive Award and Incentive Award Agreement. The Committee's power and authority to amend or modify the terms of an outstanding Incentive Award Agreement includes the authority to modify the number of shares of Common Stock or other terms and conditions of an Incentive Award, excelerate the exercisability or vesting or otherwise terminate any restrictions relating to an Incentive Award, accept the surrender of any outstanding Incentive Award or, to the extent not previously exercised, settled or vested, authorize the grant of new Incentive Awards in substitution for surrendered Incentive Awards; provided, however that the amended or modified terms are permitted by the Plan as then in effect and that any Participant adversely affected by such amended or modified terms has consented to such amendment or modification.
- 19.2 Adjustment of Performance-Based Awards. The Committee may amend or modify the vesting criteria (including any performance objectives or performance periods) of any outstanding Incentive Awards based in whole or in part on the financial performance of the Company (or any Subsidiary or division, business unit or other sub-unit thereof) in recognition of unusual or nonrecurring events (including the events described in Section 4.3(a) of the Plan) affecting the Company or the financial statements of the Company or of changes in applicable law, regulations or accounting principles, whenever the Committee determines that such adjustments are appropriate in order to prevent unintended dilution or enlargement of the benefits or potential benefits intended to be made available under the Plan. The determination of the Committee as to the foregoing adjustments, if any, will be final, conclusive and binding on Participants under the Plan.
- 19.3 Stockholder Approval. No amendments to the Plan will be effective without approval of the Company's stockholders if: (a) stockholder approval of the amendment is then required pursuant to Section 422 of the Code, the rules of the primary stock exchange or stock market on which the Common Stock is then traded, applicable U.S. state corporate laws or regulations, applicable U.S. federal laws or regulations, and the applicable laws of any foreign country or jurisdiction where Incentive Awards are, or will be, granted under the Plan; or (b) such amendment would: (i) modify Section 3.4 of the Plan; (ii) materially increase benefits accruing to Participants; (iii) increase the aggregate number of shares of Common Stock issued or issuable under the Plan; (iv) increase any limitation set forth in the Plan on the number of shares of Common Stock which may be issued or the aggregate value of Incentive Awards which may be made, in respect of any type of Incentive Award; (v) modify the eligibility requirements for Participants in the Plan; or (vi) reduce the minimum exercise price as set forth in Sections 6.3 and 7.3 of the Plan.
- 19.4 <u>Incentive Awards Previously Granted.</u> Notwithstanding any other provision of the Plan to the contrary, no termination, suspension or amendment of the Plan may adversely affect any outstanding Incentive Award without the consent of the affected Participant; provided, however, that this sentence will not impair the right of the Committee to take whatever action it deems appropriate under Sections 3.2, 4.3, 13, 15, 18 or 19.5 of the Plan.

19.5 Amendments to Conform to Law. Notwithstanding any other provision of the Plan to the contrary, the Committee may amend the Plan or an Incentive Award Agreement, to take effect retroactively or otherwise, as deemed necessary or advisable for the purpose of conforming the Plan or an Incentive Award Agreement to any present or future law relating to plans of this or similar nature, and to the administrative regulations and rulings promulgated thereunder. By accepting an Incentive Award under the Plan, a Participant agrees to any amendment made pursuant to this Section 19.5 to any Incentive Award granted under the Plan without further consideration or action.

Effective Date and Duration of the Plan.

The Plan will be effective as of the Effective Date and will terminate at midnight on the day before the tenth (10th) anniversary of the Effective Date, and may be terminated prior to such time by Board action. No Incentive Award will be granted after termination of the Plan but Incentive Awards outstanding upon termination of the Plan will remain outstanding in accordance with their applicable terms and conditions and the terms and conditions of the Plan.

Miscellaneous.

- 21.1 <u>Usage</u>. In the Plan, except where otherwise indicated by clear contrary intention, (a) any masculine term used in the Plan also will include the feminine, (b) the plural will include the singular, and the singular will include the plural, (c) "including" (and with correlative meaning "include") means including without limiting the generality of any description preceding such term, and (d) "or" is used in the inclusive sense of "and/or"
- 21.2 <u>Unfunded Plan.</u> Participants will have no right, title or interest whatsoever in or to any investments that the Company or its Subsidiaries may make to aid it in meeting its obligations under the Plan. Nothing contained in the Plan, and no action taken pursuant to its provisions, will create or be construed to create a trust of any kind, or a fiduciary relationship between the Company and any Participant, beneficiary, legal representative, or any other individual. To the extent that any individual acquires a right to receive payments from the Company or any Subsidiary under the Plan, such right will be no greater than the right of an unsecured general creditor of the Company or the Subsidiary, as the case may be, all payments to be made hereunder will be paid from the general funds of the Company or the Subsidiary, as the case may be, and no special or separate fund will be established and no segregation of assets will be made to assure payment of such amounts except as expressly set forth in the Plan.
- 21.3 Relationship to Other Benefits. No payment under the Plan will be taken into account in determining any benefits under any pension, retirement, savings, profit sharing, group insurance, welfare, or benefit plan of the Company or any Subsidiary unless provided otherwise in such plan.
- 21.4 <u>Fractional Shares</u>. No fractional shares of Common Stock will be issued or delivered under the Plan or any Incentive Award. The Committee will determine whether cash, other Incentive Awards or other property will be issued or paid in lieu of fractional shares of Common Stock or whether such fractional shares of Common Stock or any rights thereto will be forfeited or otherwise eliminated by rounding up or down.
- 21.5 No Representations or Warranties Regarding Tax Effect. Notwithstanding any provision of the Plan to the contrary, the Company, its Subsidiaries, the Board and the Committee neither represent nor warrant the tax treatment under any federal, state, provincial, local, foreign or other laws of any Incentive Award granted or amounts paid to any Participant under this Plan, including when and to what extent such Incentive Award or amounts may be subject to tax, penalties and interest.

- 21.6 <u>Governing Law; Venue</u>. Except to the extent expressly provided in the Plan or in connection with other matters of corporate governance and authority (all of which will be governed by the laws of the Company's jurisdiction of incorporation), the validity, construction, interpretation, administration and effect of the Plan and any rules, regulations and actions relating to the Plan will be governed by and construed exclusively in accordance with the laws of the State of Illinois, notwithstanding the conflicts of laws principles of any jurisdictions. Unless otherwise provided in an Incentive Award Agreement, recipients of an Incentive Award and all issues that may arise out of or relate to the Plan or any related Incentive Award Agreement.
- 21.7 Successors. All obligations of the Company under the Plan with respect to Incentive Awards granted hereunder will be binding on any successor to the Company, whether the existence of such successor is the result of a direct or indirect purchase, merger, consolidation or otherwise, of all or substantially all of the business or assets of the Company.
- 21.8 Construction. Wherever possible, each provision of the Plan and any Incentive Award Agreement evidencing an Incentive Award granted under the Plan will be interpreted so that it is valid under the applicable law. If any provision of the Plan or any Incentive Award Agreement evidencing an Incentive Award granted under the Plan is to any extent invalid under the applicable law, that provision will still be effective to the extent it remains valid. The remainder of the Plan and the Incentive Award Agreement also will continue to be valid, and the entire Plan and Incentive Award Agreement will continue to be valid in other invasidations.
- 21.9 <u>Delivery and Execution of Electronic Documents</u>. To the extent permitted by applicable law, the Company may: (a) deliver by email or other electronic means (including posting on a Web site maintained by the Company or by a third party under contract with the Company) all documents relating to the Plan or any Incentive Award hereunder (including prospectuses required by the Securities and Exchange Commission) and all other documents that the Company is required to deliver to its security holders (including annual reports and proxy statements), and (b) permit Participants to use electronic, internet or other non-paper means to execute applicable Plan documents (including Incentive Award Agreements) and take other actions under the Plan in a manner prescribed by the Committee.

INCENTIVE STOCK OPTION AGREEMENT

THIS INCENTIVE STOCK OPTION AGREEMENT (this "Agreement") is entered into and effective as of [(the "Company"), and [] ("Optionee").

] (the "Grant Date") by and between RestorGenex Corporation, a Delaware corporation

- A. The Company has adopted the RestorGenex Corporation 2015 Equity Incentive Plan (as such plan may be amended from time to time, the "Plan") authorizing the Board of Directors (the "Board") of the Company, or a committee as provided for in the Plan (the Board or such a committee to be referred to as the "Committee"), to grant stock options, among other incentive awards, to certain individuals.
 - B. The Company desires to grant an incentive stock option to purchase shares of common stock, par value \$0.001 per share, of the Company (the "Common Stock") to Optionee pursuant to the Plan.
 - C. All of the capitalized terms used in this Agreement not otherwise defined in this Agreement have the same respective meanings as defined in the Plan.

NOW, THEREFORE, in consideration of the mutual promises and covenants hereinafter set forth, the Company and Optionee agree as follows:

- 2. <u>Vesting</u>. The Option shall vest and become exercisable in [12 quarterly/] equal (or as nearly equal as possible) installments on [the last calendar day of each calendar quarter/ over a [three-year/] period, beginning on []. [Optionee shall receive a full quarter of vesting for the [] calendar quarter of [].]

[OR, IN THE CASE OF PERFORMANCE-BASED VESTING]

[The Option shall vest and become exercisable []].

- Exercise of Option
 - 3.1. Notice; Payment. Subject to the terms and conditions set forth in this Agreement, including vesting of the Option in Section 2 of this Agreement and termination of

the Option in Section 4 of this Agreement, and the Plan, the Option may be exercised, in whole or in part, at any time and from time to time, by delivery to the Company of written notice of the exercise of the Option, in substantially the form as provided by the Company, stating the number of Option Shares being purchased (the "Purchased Shares"), and accompanied by payment in full of the total aggregate Exercise Price of the Purchased Shares. The Exercise Price shall be payable in full in any one of the following alternative forms:

- (a) Full payment in cash, personal check or certified bank or cashier's check;
- (b) Any broker assisted cashless exercise procedure which is acceptable to the Company; or
- (c) Cashless net exercise.

Upon a cashless net exercise, Optionee shall receive the number of shares of Common Stock equal to a number (as determined below) of shares of Common Stock computed using the following formula:

 $X = Y - [\frac{(A)(Y)}{B}$

Where X = the number of shares of Common Stock to be issued to Optionee.

Y = the number of Purchased Shares.

A = the Exercise Price.

B = the Fair Market Value of one share of Common Stock on the date of exercise.

3.2. <u>Issuance of Purchased Shares; No Fractional Shares.</u> Following receipt of the exercise notice and the payment referred to above, the Company shall, as soon as reasonably practicable thereafter, cause certificates (or book-entry notations) representing the Purchased Shares (or such fewer number of Purchased Shares if a cashless net exercise is used) to be delivered to Optionee identificates (or book-entry notations) representing the Purchased Shares (or such fewer number of Purchased Shares if a cashless net exercise is used) to be delivered to Optionee identificates (or book-entry notations) representing the Purchased Shares for Optionee's benefit with any broker with which Optionee has an account relationship or the Company has engaged to provide such services under the Plan; <u>provided, however,</u> that the Company shall not be obligated to issue a fraction or fractions of a share otherwise issuable upon exercise of the Option, and may pay to Optionee, in cash or cash equivalent, the Fair Market Value of any such fraction or fractions of a share as of the date of exercise. If requested by the Company in connection with any exercise of the Option, Optionee shall also deliver this Agreement to the Company, which shall endorse hereon a notation of the exercise and, and if the Option is exercised in part, shall return this Agreement to Optionee. The date of exercise of an Option that is validly exercised shall be deemed to be the date on which there shall have been delivered to the Company the notice referred to in <u>Section 3.1</u> of this

2

Agreement and full payment of the Exercise Price of the Purchased Shares. Optionee shall not be deemed to be a holder of any Purchased Shares pursuant to exercise of the Option until the date of issuance of a stock certificate or book-entry notation to Optionee for such shares following payment in full for the Purchased Shares.

- 3.3. Tax Withholding. The Company is entitled to (a) withhold and deduct from future wages of Optionee (or from other amounts that may be due and owing to Optionee from the Company or a Subsidiary), or make other arrangements for the collection of, all amounts the Company reasonably determines are necessary to satisfy any and all federal, foreign, state and local withholding and employment related tax requirements attributable to the Option, including, without limitation, the grant, exercise or vesting of, the Option; (b) withhold cash paid or payable or shares of Common Stock from the shares issued or otherwise issuable to Optionee in connection with the Option; or (c) require Optionee promptly to remit the amount of such withholding to the Company before taking any action, including issuing any shares of Common Stock, with respect to the Option. Shares of Common Stock issued or otherwise issuable to Optionee in connection with the Option that gives rise to the tax withholding obligation that are withheld for purposes of satisfying Optionee's withholding or employment-related tax obligation will be valued at their Fair Market Value on the Tax Date.
- 3.4. Remaining Option Shares. Option Shares will no longer be outstanding under the Option (and will therefore not thereafter be exercisable) following the exercise of the Option to the extent of (a) shares used to pay the Exercise Price of an Option under the "cashless net exercise" method (b) shares actually delivered to Optionee as a result of such exercise and (c) any shares withheld for purposes of tax withholding.
 - 4. <u>Termination of Option</u>
 - 4.1. <u>Time of Termination</u>. Except as provided in this <u>Section 4</u> and <u>Section 5</u> of this Agreement, the Option shall terminate, no longer be exercisable and expire at 5:00 p.m., Eastern Time, on [(the "Time of Termination").
- 4.2. <u>Termination for Cause</u>. In the event Optionee's employment with the Company and all Subsidiaries is terminated by the Company for Cause, the Option will immediately terminate without notice of any kind, and the Option will no longer be exercisable.
- 4.3. <u>Termination Due to Death, Disability or Retirement</u>. In the event Optionee's employment with the Company and all Subsidiaries is terminated by reason of Optionee's death, Disability or Retirement, the Option will remain exercisable, to the extent exercisable as of the date of such termination, for a period of one (1) year after such termination (but in no event after the Time of Termination).
- 4.4. <u>Termination for Other Reasons</u>. In the event Optionee's employment with the Company and all Subsidiaries is terminated for any other reason, the Option will, to the extent exercisable as of such termination, remain exercisable for a period of three (3) months after such termination (but in no event after the Time of Termination).
 - 4.5. Effect of Actions Constituting Cause or Adverse Action. Notwithstanding anything in this Agreement to the contrary and in addition to the rights of the Committee under

Sections 13.5 and 13.6 of the Plan, if Optionee is determined by the Committee, acting in its sole discretion, to have taken any action that would constitute Cause or an Adverse Action during or after the termination of employment with the Company or a Subsidiary, irrespective of whether such action or the Committee's determination occurs before or after termination of Optionee's employment with the Company or any Subsidiary and irrespective of whether or not Optionee was terminated as a result of such Cause or Adverse Action, (a) all rights of Optionee under the Option and this Agreement will terminate and be forfeited without notice of any kind, and (b) the Committee in its sole discretion will have the authority to rescind the exercise, vesting, settlement or issuance of, or payment in respect of, the Option that was exercised, vested, settled or issued, or as to which such payment was made, and to require Optionee to pay to the Company, within ten (10) days of receipt from the Company of notice of such rescission, any amount received or the amount of any gain realized as a result of such rescinded exercise, vesting, settlement, issuance or payment (including any dividends paid or other distributions made with respect to any shares of Common Stock subject to the Option). The Company may defer the exercise of the Option for a period of up to six (6) months after receipt of Optionee's written notice of exercise or the issuance of Purchased Shares upon the vesting of the Option for a period of up to six (6) months after the date of such vesting in order for the Committee to make any determination as to the existence of Cause or an Adverse Action. The Company will be entitled to withhold and deduct from future wages of Optionee (or from other amounts that may be due and owing to Optionee from the Company or a Subsidiary) or make other arrangements for the collection of all amounts necessary to satisfy such payment obligations. This Section 4.5 will not apply to the Option following a Change in Control.

- 4.6. <u>Clawback/Forfeiture</u>. The Option and Option Shares issued or issuable pursuant to the Option are subject to forfeiture or clawback by the Company to the extent required and allowed by law, including the Dodd-Frank Wall Street Reform and Consumer Protection Act of 2010 and the Sarbanes Oxley Act of 2002 and any implementing rules and regulations promulgated thereunder, and pursuant to any forfeiture, clawback or similar policy of the Company, as such laws, rules, regulations and policy may be in effect from time to time.
- 5. Adjustments. In the event of any reorganization, merger, consolidation, recapitalization, liquidation, reclassification, stock dividend, stock split, combination of shares, rights offering, divestiture or extraordinary dividend (including a spin-off), or any other similar change in the corporate structure or shares of the Company, the Committee (or, if the Company is not the surviving corporation in any such transaction, the board of directors of the surviving corporation), will make appropriate adjustment (which determination will be conclusive) as to the number and kind of securities or other property (including cash) subject to, and the Exercise Price of, the Option in order to prevent dilution or enlargement of the rights of Optionee.
- 6. <u>Change in Control.</u> The Option shall become immediately vested and exercisable upon completion of a Change in Control and remain exercisable through the Time of Termination regardless of whether Optionee remains in the employment of the Company. Notwithstanding any of the foregoing, in connection with a Change in Control, the Committee, in its sole discretion, at any time after the grant of the Option, may take whatever action it deems appropriate pursuant to Section 15.3 of the Plan.

4

- 7. Rights as a Stockholder. Optionee will have no rights as a stockholder of the Company unless and until all conditions to the effective exercise of the Option (including, without limitation, the conditions set forth in Section 3 of this Agreement) have been satisfied and Optionee has become the holder of record of such shares. No adjustment will be made for dividends or distributions with respect to the Option as to which there is a record date preceding the date Optionee becomes the holder of record of such shares, except as may otherwise be provided in the Plan or determined by the Committee in its sole discretion.
- 8. Restrictions on Transfer. Except pursuant to testamentary will or the laws of descent and distribution or as otherwise expressly permitted by the Plan, no right or interest of Optionee in the Option prior to exercise may be assigned or transferred, or subjected to any lien, during the lifetime of Optionee, either voluntarily or involuntarily, directly or indirectly, by operation of law or otherwise. Optionee, however, will be entitled to designate a beneficiary to receive the Option upon Optionee's death, and, in the event of Optionee's death, exercise of the Option (to the extent permitted pursuant to Sections 2 and 4 of this Agreement) may be made by Optionee's legal representatives, heirs and legatees.
- 9. <u>Market Stand-off.</u> Optionee, if so requested by the Company or any representative of the underwriters in connection with a firmly underwritten public offering of securities by the Company pursuant to a registration statement under the Securities Act following the date of this Agreement, shall not sell or otherwise transfer any Option Shares during the 180-day period following the effective date of such registration statement. The Company may impose stop-transfer instructions with respect to securities subject to the foregoing restriction until the end of such 180-day period. This <u>Section 9</u> will not apply to the sale of any Option Shares to an underwriter pursuant to an underwriting agreement and shall only be applicable to Optionee if all then current executive officers and directors of the Company enter into similar agreements.
- 10. <u>Employment</u>. Nothing in this Agreement or the Plan will interfere with or limit in any way the right of the Company or any Subsidiary to terminate the employment of Optionee at any time, nor confer upon Optionee any right to continue in the employment with the Company or any Subsidiary.
- 11. Option Subject to Plan. The Option and the Option Shares granted and issued pursuant to this Agreement have been granted and issued under, and are subject to the terms of, the Plan. The terms of the Plan are incorporated by reference in this Agreement in their entirety, and Optionee, by execution of this Agreement, acknowledges having received a copy of the Plan. The provisions of this Agreement will be interpreted as to be consistent with the Plan, and any ambiguities in this Agreement will be interpreted by reference to the Plan. In the event that any provision of this Agreement is inconsistent with the terms of the Plan, the terms of the Plan will prevail. All of the capitalized terms used in this Agreement not otherwise defined in this Agreement have the same respective meanings as defined in the Plan.
 - Incentive Stock Option Limitations.
 - 12.1. <u>Limitation on Amount</u>. To the extent that the aggregate Fair Market Value (determined as of the Grant Date) of the shares of Common Stock with respect to which

5

incentive stock options (within the meaning of Section 422 of the Code) are exercisable for the first time by the Optionee during any calendar year (under the Plan and any other incentive stock option plans of the Company or any subsidiary or parent corporation of the Company (within the meaning of the Code)) exceeds \$100,000 (or such other amount as may be prescribed by the Code from time to time), such excess incentive stock options will be treated as non-statutory stock options in the manner set forth in the Plan.

- 12.2. <u>Limitation on Exercisability; Disposition of Option Shares</u>. Any incentive stock option that remains unexercised more than one year following termination of employment by reason of death or disability or more than three months following termination for any reason other than death or disability will thereafter be deemed to be a non-statutory stock option. In addition, in the event that a disposition (as defined in Section 424(c) of the Code) of shares of Common Stock acquired pursuant to the exercise of an incentive stock option occurs prior to the expiration of two years after its date of grant or the expiration of one year after its date of exercise (a "disqualifying disposition"), such incentive stock option will, to the extent of such disqualifying disposition, be treated in a manner similar to a non-statutory stock option.
- 12.3. No Representation or Warranty. Section 422 of the Code and the rules and regulations thereunder are complex, and neither the Plan nor this Agreement purports to summarize or otherwise set forth all of the conditions that need to be satisfied in order for this Option to qualify as an incentive stock option. In addition, this Option may contain terms and conditions that allow for exercise of this Option beyond the periods permitted by Section 422 of the Code, including, without limitation, the periods described in Sections 2 and 4 of this Agreement. Accordingly, the Company makes no representation or warranty regarding whether the exercise of this Option will qualify as the exercise of an incentive stock option, and the Company recommends that the Optionee consult with the Optionee's own advisors before making any determination regarding the exercise of this Option or the sale of the Option Shares.

13. <u>General Provisions</u>.

- 13.1. Governing Law; Venue. This Agreement and all rights and obligations under this Agreement will be governed by and construed exclusively in accordance with the laws of the State of Delaware, notwithstanding the conflicts of laws principles of any jurisdictions. By acceptance of the Option, Optionee is deemed to submit to the exclusive jurisdiction and venue of the federal or state courts of the State of Illinois to resolve any and all issues that may arise out of or relate to the Option or this Agreement.
- 13.2. Entire Agreement. This Agreement and the Plan set forth the entire agreement and understanding of the parties to this Agreement with respect to the grant and exercise of the Option and the administration of the Plan and supersede all prior agreements, arrangements, plans and understandings relating to the grant and exercise of the Option and the administration of the Plan.
- 13.3. Failure to Enforce Not a Waiver. The failure of the Company or Optionee to enforce at any time any provision of this Agreement shall in no way be construed to be a waiver of such provision or of any other provision hereof.

- 13.4. Notices. All notices, requests, demands and other communications (collectively, "Notices") given pursuant to this Agreement shall be in writing, and shall be delivered by personal service, courier, facsimile transmission, email transmission of a pdf format data file or by United States first class, registered or certified mail, postage prepaid, addressed to the party at the address set forth on the signature page of this Agreement. Any Notice, other than a Notice sent by registered or certified mail, shall be effective when received; a Notice sent by registered or certified mail, postage prepaid return receipt requested, shall be effective on the earlier of when received or the third day following deposit in the United States mails. Any party may from time to time change its address for further Notices hereunder by giving notice to the other party in the manner prescribed in this Section 13.5.
- 13.5. Successors and Assigus. Except to the extent specifically limited by the terms and provision of this Agreement, this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors, assigns, heirs and personal representatives.
- 13.6. Execution. This Agreement may be executed in two or more counterparts, all of which when taken together shall be considered one and the same agreement and shall become effective when counterparts have been signed by each party and delivered to the other party, it being understood that both parties need not sign the same counterpart. In the event that any signature is delivered by facsimile transmission or by email delivery of a "pdf" format data file, such signature shall create a valid and binding obligation of the party executing (or on whose behalf such signature is executed) with the same force and effect as if such facsimile or "pdf" signature page were an original thereof.
- 13.7. <u>Titles, Captions and Sections</u>. Titles and captions contained in this Agreement are inserted for convenience of reference only and do not constitute a part of this Agreement for any other purpose. References to Sections in this Agreement refer to Sections of this Agreement unless otherwise stated.

13.0	8. <u>Nature</u>	e of the Grant. In accepting the Option and by execution of this Agreement, Optionee acknowledges that:					
me, unless otherwise	(a) e provided ir	The Plan is established voluntarily by the Company, it is discretionary in nature and it may be modified, amended, suspended or terminated by the Company in its sole discretion at an the Plan.					
rants have been gran	(b) nted repeated	The grant of the Option is voluntary and occasional and does not create any contractual or other right to receive future Option grants, or benefits in lieu of Option grants, even if Optily in the past.					
	(c)	All decisions with respect to future Option grants, if any, will be at the sole discretion of the Company.					
(d) Optionee is voluntarily participating in the Plan.							
	(e)	The Option grant is not part of normal or expected compensation or salary for any purposes, including, but not limited to, calculating any severance, resignation,					
ermination, redundan ervices for the Comp		service payments, bonuses, long-service awards, pension or retirement benefits or similar payments and in no event shall be considered as compensation for, or relating in any way to, p					
	(f)	In the event that Optionee is not an employee of the Company, the Option will not be interpreted to form an employment contract or relationship with the Company.					
his Agreement and is	(g) s issued Purc	(g) The future value of the Common Stock is unknown and cannot be predicted with certainty and if the Option vests and Optionee exercises the Option in accordance with the terms of ssued Purchased Shares, the value of those shares may increase or decrease.					
ourchased Shares acq		In consideration of the grant of the Option, no claim or entitlement to compensation or damages shall arise from termination of the Option or diminution in value of the Option or exercise of the Option resulting from termination of Optionee's employment or service by the Company (for any reason whatsoever and whether or not in breach of local labor laws) are					
ound by a court of co		exercise of the Option resulting from refinition of Optionee's employment of service by the Company (tot any reason whatsoever and whether of not in treat of noted habor laws) are company and its Subsidiaries, and their respective directors, officers, employees and agents, from any such claim that may arise; if, notwithstanding the foregoing, any such claim is risdiction to have arisen, then, by acceptance of the Option and execution of this Agreement, Optionee shall be deemed irrevocably to have waived his or her entitlement to pursue such					
	ompetent jur	Company and its Subsidiaries, and their respective directors, officers, employees and agents, from any such claim that may arise; if, notwithstanding the foregoing, any such claim is risdiction to have arisen, then, by acceptance of the Option and execution of this Agreement, Optionee shall be deemed irrevocably to have waived his or her entitlement to pursue such. The Company is not providing any tax, legal or financial advice, nor is the Company making any recommendations regarding Optionee's participation in the Plan, or Optionee's goption Shares.					
ound by a court of colaim. urchase or sale of the	ompetent jur	Company and its Subsidiaries, and their respective directors, officers, employees and agents, from any such claim that may arise; if, notwithstanding the foregoing, any such claim is risdiction to have arisen, then, by acceptance of the Option and execution of this Agreement, Optionee shall be deemed irrevocably to have waived his or her entitlement to pursue such. The Company is not providing any tax, legal or financial advice, nor is the Company making any recommendations regarding Optionee's participation in the Plan, or Optionee's					
ound by a court of colaim. urchase or sale of the	ompetent jur (i) e underlying	Company and its Subsidiaries, and their respective directors, officers, employees and agents, from any such claim that may arise; if, notwithstanding the foregoing, any such claim is risdiction to have arisen, then, by acceptance of the Option and execution of this Agreement, Optionee shall be deemed irrevocably to have waived his or her entitlement to pursue such. The Company is not providing any tax, legal or financial advice, nor is the Company making any recommendations regarding Optionee's participation in the Plan, or Optionee's goption Shares.					
ound by a court of collaim.	ompetent jur (i) e underlying	e Company and its Subsidiaries, and their respective directors, officers, employees and agents, from any such claim that may arise; if, notwithstanding the foregoing, any such claim is risdiction to have arisen, then, by acceptance of the Option and execution of this Agreement, Optionee shall be deemed irrevocably to have waived his or her entitlement to pursue such. The Company is not providing any tax, legal or financial advice, nor is the Company making any recommendations regarding Optionee's participation in the Plan, or Optionee's goption Shares. Optionee is hereby advised to consult with his or her own personal tax, legal and financial advisors regarding his or her participation in the Plan before taking any action related to the participation in the Plan before taking any action related to the participation in the Plan before taking any action related to the participation in the Plan before taking any action related to the participation in the Plan before taking any action related to the participation in the Plan before taking any action related to the participation in the Plan before taking any action related to the participation in the Plan before taking any action related to the participation in the Plan before taking any action related to the participation in the Plan before taking any action related to the participation in the Plan before taking any action related to the participation in the Plan before taking any action related to the participation in the Plan before taking any action related to the participation in the Plan before taking any action related to the participation in the Plan before taking any action related to the participation in the Plan before taking any action related to the participation in the Plan before taking any action related to the participation in the Plan before taking any action related to the participation in the Plan before taking any action related to the participation in the Plan before taking any action related to the participation and the participation in the					
ound by a court of colaim. urchase or sale of the lan or the Option. IN WITNES	ompetent jur (i) e underlying (j)	e Company and its Subsidiaries, and their respective directors, officers, employees and agents, from any such claim that may arise; if, notwithstanding the foregoing, any such claim is risdiction to have arisen, then, by acceptance of the Option and execution of this Agreement, Optionee shall be deemed irrevocably to have waived his or her entitlement to pursue such The Company is not providing any tax, legal or financial advice, nor is the Company making any recommendations regarding Optionee's participation in the Plan, or Optionee's goption Shares. Optionee is hereby advised to consult with his or her own personal tax, legal and financial advisors regarding his or her participation in the Plan before taking any action related to the [Remainder of page intentionally left blank; signature page follows]					
ound by a court of colaim. Purchase or sale of the Plan or the Option.	ompetent jur (i) e underlying (j)	e Company and its Subsidiaries, and their respective directors, officers, employees and agents, from any such claim that may arise; if, notwithstanding the foregoing, any such claim is risdiction to have arisen, then, by acceptance of the Option and execution of this Agreement, Optionee shall be deemed irrevocably to have waived his or her entitlement to pursue such. The Company is not providing any tax, legal or financial advice, nor is the Company making any recommendations regarding Optionee's participation in the Plan, or Optionee's goption Shares. Optionee is hereby advised to consult with his or her own personal tax, legal and financial advisors regarding his or her participation in the Plan before taking any action related to the [Remainder of page intentionally left blank; signature page follows] 8 OF, the parties to this Agreement have executed this Agreement effective as of the Grant Date.					
ound by a court of colaim. urchase or sale of the lan or the Option. IN WITNES OPTIONEE:	ompetent jur (i) e underlying (j)	Company and its Subsidiaries, and their respective directors, officers, employees and agents, from any such claim that may arise; if, notwithstanding the foregoing, any such claim is risdiction to have arisen, then, by acceptance of the Option and execution of this Agreement, Optionee shall be deemed irrevocably to have waived his or her entitlement to pursue such The Company is not providing any tax, legal or financial advice, nor is the Company making any recommendations regarding Optionee's participation in the Plan, or Optionee's goption Shares. Optionee is hereby advised to consult with his or her own personal tax, legal and financial advisors regarding his or her participation in the Plan before taking any action related to the [Remainder of page intentionally left blank; signature page follows] 8 OF, the parties to this Agreement have executed this Agreement effective as of the Grant Date. RESTORGENEX CORPORATION By:					
ound by a court of colaim. urchase or sale of the lan or the Option. IN WITNES OPTIONEE:	ompetent jur (i) e underlying (j)	Company and its Subsidiaries, and their respective directors, officers, employees and agents, from any such claim that may arise; if, notwithstanding the foregoing, any such claim is risdiction to have arisen, then, by acceptance of the Option and execution of this Agreement, Optionee shall be deemed irrevocably to have waived his or her entitlement to pursue such The Company is not providing any tax, legal or financial advice, nor is the Company making any recommendations regarding Optionee's participation in the Plan, or Optionee's goption Shares. Optionee is hereby advised to consult with his or her own personal tax, legal and financial advisors regarding his or her participation in the Plan before taking any action related to the International Internationa					
ound by a court of colaim. urchase or sale of the colaim. IN WITNES OPTIONEE: Signature] Jame:	(i) e underlying (j) SS WHEREC	Company and its Subsidiaries, and their respective directors, officers, employees and agents, from any such claim that may arise; if, notwithstanding the foregoing, any such claim is risdiction to have arisen, then, by acceptance of the Option and execution of this Agreement, Optionee shall be deemed irrevocably to have waived his or her entitlement to pursue such a Company is not providing any tax, legal or financial advice, nor is the Company making any recommendations regarding Optionee's participation in the Plan, or Optionee's goption Shares. Optionee is hereby advised to consult with his or her own personal tax, legal and financial advisors regarding his or her participation in the Plan before taking any action related to the International state of the Interna					

STOCK OPTION AGREEMENT

The Company has adopted the RestorGenex Corporation 2015 Equity Incentive Plan (as such plan may be amended from time to time, the "Plan") authorizing the Board of Directors (the "Board") of the

THIS STOCK OPTION AGREEMENT (this "Agreement") is entered into and effective as of [] (the "Gran "Company"), and [] ("Optionee").

] (the "Grant Date") by and between RestorGenex Corporation, a Delaware corporation (the

- Company, or a committee as provided for in the Plan (the Board or such a committee to be referred to as the "Committee"), to grant stock options, among other incentive awards, to certain individuals.
 - B. The Company desires to grant an option to purchase shares of common stock, par value \$0.001 per share, of the Company (the "Common Stock") to Optionee pursuant to the Plan.
 - C. All of the capitalized terms used in this Agreement not otherwise defined in this Agreement have the same respective meanings as defined in the Plan.

NOW, THEREFORE, in consideration of the mutual promises and covenants hereinafter set forth, the Company and Optionee agree as follows:

[OR, IN THE CASE OF PERFORMANCE-BASED VESTING]

[The Option shall vest and become exercisable [

- Exercise of Option.
- 3.1. Notice; Payment. Subject to the terms and conditions set forth in this Agreement, including vesting of the Option in Section 2 of this Agreement and termination of the Option in Section 4 of this Agreement, and the Plan, the Option may be exercised, in whole or in part, at any time and from time to time, by delivery to the Company of written notice of the

exercise of the Option, in substantially the form as provided by the Company, stating the number of Option Shares being purchased (the "Purchased Shares"), and accompanied by payment in full of the total aggregate Exercise Price of the Purchased Shares. The Exercise Price shall be payable in full in any one of the following alternative forms:

- (a) Full payment in cash, personal check or certified bank or cashier's check;
- (b) Any broker assisted cashless exercise procedure which is acceptable to the Company; or
- (c) Cashless net exercise.

Upon a cashless net exercise, Optionee shall receive the number of shares of Common Stock equal to a number (as determined below) of shares of Common Stock computed using the following formula:

 $X = Y - [\frac{(A)(Y)}{B}$

Where X = the number of shares of Common Stock to be issued to Optionee.

Y = the number of Purchased Shares.

A = the Exercise Price.

B = the Fair Market Value of one share of Common Stock on the date of exercise.

3.2. <u>Issuance of Purchased Shares; No Fractional Shares.</u> Following receipt of the exercise notice and the payment referred to above, the Company shall, as soon as reasonably practicable thereafter, cause certificates (or book-entry notations) representing the Purchased Shares (or such fewer number of Purchased Shares if a cashless net exercise is used) to be delivered to Optionee either at Optionee's address set forth in the records of the Company or at such other address as Optionee may designate in writing to the Company or issue and deposit the Purchased Shares for Optionee's benefit with any broker with which Optionee has an account relationship or the Company has engaged to provide such services under the Plan; <u>provided, however</u>, that the Company shall not be obligated to issue a fraction or fractions of a share otherwise issuable upon exercise of the Option, and may pay to Optionee, in cash or cash equivalent, the Fair Market Value of any such fraction or fractions of a share as of the date of exercise. If requested by the Company in connection with any exercise of the Option, Optionee shall also deliver this Agreement to the Company, which shall endorse hereon a notation of the exercise and, and if the Option is exercised in part, shall return this Agreement to Optionee. The date of exercise of an Option that is validly exercised shall be deemed to be the date on which there shall have been delivered to the Company the notice referred to in <u>Section 3.1</u> of this Agreement and full payment of the Exercise Price of the Purchased Shares. Optionee shall not be deemed to be a holder of any Purchased Shares pursuant to exercise of the Option until the

2

date of issuance of a stock certificate or book-entry notation to Optionee for such shares following payment in full for the Purchased Shares.

- 3.3. Tax Withholding. The Company is entitled to (a) withhold and deduct from future wages of Optionee (or from other amounts that may be due and owing to Optionee from the Company or a Subsidiary), or make other arrangements for the collection of, all amounts the Company reasonably determines are necessary to satisfy any and all federal, foreign, state and local withholding and employment related tax requirements attributable to the Option, including, without limitation, the grant, exercise or vesting of, the Option; (b) withhold cash paid or payable or shares of Common Stock from the shares issued or otherwise issuable to Optionee in connection with the Option; or (c) require Optionee promptly to remit the amount of such withholding to the Company before taking any action, including issuing any shares of Common Stock with respect to the Option. Shares of Common Stock issued or otherwise issuable to Optionee in connection with the Option that gives rise to the tax withholding obligation that are withheld for purposes of satisfying Optionee's withholding or employment-related tax obligation will be valued at their Fair Market Value on the Tax Date.
- 3.4. Remaining Option Shares. Option Shares will no longer be outstanding under the Option (and will therefore not thereafter be exercisable) following the exercise of the Option to the extent of (a) shares used to pay the Exercise Price of an Option under the "cashless net exercise" method (b) shares actually delivered to Optionee as a result of such exercise and (c) any shares withheld for purposes of tax withholding.
 - 4. <u>Termination of Option</u>
 - 4.1. <u>Time of Termination</u>. Except as provided in this <u>Section 4</u> and <u>Section 5</u> of this Agreement, the Option shall terminate, no longer be exercisable and expire at 5:00 p.m., Eastern Time, on [(the "Time of Termination").
- 4.2. <u>Termination for Cause</u>. In the event Optionee's employment (in the event that Optionee is an Employee) or other service (in the event that Optionee is a Consultant) with the Company and all Subsidiaries is terminated by the Company for Cause, the Option will immediately terminate without notice of any kind, and the Option will no longer be exercisable.
- 4.3. <u>Termination Due to Death, Disability or Retirement.</u> In the event Optionee's employment (in the event that Optionee is an Employee) or other service (in the event that Optionee is a Consultant) with the Company and all Subsidiaries is terminated by reason of Optionee's death, Disability or Retirement, the Option will remain exercisable, to the extent exercisable as of the date of such termination, for a period of one (1) year after such termination (but in no event after the Time of Termination).
- 4.4. <u>Termination for Other Reasons</u>. In the event Optionee's employment (in the event that Optionee is an Employee) or other service (in the event that Optionee is a Consultant) with the Company and all Subsidiaries is terminated for any other reason, the Option will, to the extent exercisable as of such termination, remain exercisable for a period of three (3) months after such termination (but in no event after the Time of Termination).

- 4.5. Effect of Actions Constituting Cause or Adverse Action. Notwithstanding anything in this Agreement to the contrary and in addition to the rights of the Committee under Sections 13.5 and 13.6 of the Plan, if Optionee is determined by the Committee, acting in its sole discretion, to have taken any action that would constitute Cause or an Adverse Action during or after the termination of employment or other service with the Company or a Subsidiary, irrespective of whether such action or the Committee's determination occurs before or after termination of Optionee's employment or other service with the Company or any Subsidiary and irrespective of whether or not Optionee was terminated as a result of such Cause or Adverse Action, (a) all rights of Optionee under the Option and this Agreement will terminate and be forfeited without notice of any kind, and (b) the Committee in its sole discretion will have the authority to rescind the exercise, vesting, settlement or issuance of, or payment in respect of, the Option that was exercised, vested, ested or issued, or as to which such payment was made, and to require Optionee to pay to the Company, within ten (10) days of receipt from the Company of notice of such rescission, any amount received or the amount of any gain realized as a result of such rescinded exercise, vesting, settlement, issuance or payment (including any dividends paid or other distributions made with respect to any shares of Common Stock subject to the Option). The Company may defer the exercise of the Option for a period of up to six (6) months after the date of such vesting in order for the Committee to make any determination as to the existence of exercise or the issuance of Purchased Shares upon the vesting of the Option for a period of up to six (6) months after the date of such vesting in order for the Committee to make any determination as to the existence of Cause or an Adverse Action. The Company will be entitled to withhold and deduct from future wages of Optionee (or from other amounts
- 4.6. <u>Clawback/Forfeiture</u>. The Option and Option Shares issued or issuable pursuant to the Option are subject to forfeiture or clawback by the Company to the extent required and allowed by law, including the Dodd-Frank Wall Street Reform and Consumer Protection Act of 2010 and the Sarbanes Oxley Act of 2002 and any implementing rules and regulations promulgated thereunder, and pursuant to any forfeiture, clawback or similar policy of the Company, as such laws, rules, regulations and policy may be in effect from time to time.
- 5. Adjustments. In the event of any reorganization, merger, consolidation, recapitalization, liquidation, reclassification, stock dividend, stock split, combination of shares, rights offering, divestiture or extraordinary dividend (including a spin-off), or any other similar change in the corporate structure or shares of the Company, the Committee (or, if the Company is not the surviving corporation in any such transaction, the board of directors of the surviving corporation), will make appropriate adjustment (which determination will be conclusive) as to the number and kind of securities or other property (including cash) subject to, and the Exercise Price of, the Option in order to prevent dilution or enlargement of the rights of Optionee.
- 6. <u>Change in Control.</u> The Option shall become immediately vested and exercisable upon completion of a Change in Control and remain exercisable through the Time of Termination regardless of whether Optionee remains in the employment or service of the Company. Notwithstanding any of the foregoing, in connection with a Change in Control, the Committee, in its sole discretion, at any time after the grant of the Option, may take whatever action it deems appropriate pursuant to <u>Section 15.3</u> of the Plan.

- 7. Rights as a Stockholder. Optionee will have no rights as a stockholder of the Company unless and until all conditions to the effective exercise of the Option (including, without limitation, the conditions set forth in Section 3 of this Agreement) have been satisfied and Optionee has become the holder of record of such shares. No adjustment will be made for dividends or distributions with respect to the Option as to which there is a record date preceding the date Optionee becomes the holder of record of such shares, except as may otherwise be provided in the Plan or determined by the Committee in its sole discretion.
- 8. Restrictions on Transfer. Except pursuant to testamentary will or the laws of descent and distribution or as otherwise expressly permitted by the Plan, no right or interest of Optionee in the Option prior to exercise may be assigned or transferred, or subjected to any lien, during the lifetime of Optionee, either voluntarily, directly or indirectly, by operation of law or otherwise. Optionee, however, will be entitled to designate a beneficiary to receive the Option upon Optionee's death, and, in the event of Optionee's death, exercise of the Option (to the extent permitted pursuant to Sections 2 and 4 of this Agreement) may be made by Optionee's legal representatives, heirs and legatees.
- 9. <u>Market Stand-off.</u> Optionee, if so requested by the Company or any representative of the underwriters in connection with a firmly underwritten public offering of securities by the Company pursuant to a registration statement under the Securities Act following the date of this Agreement, shall not sell or otherwise transfer any Option Shares during the 180-day period following the effective date of such registration statement. The Company may impose stop-transfer instructions with respect to securities subject to the foregoing restriction until the end of such 180-day period. This <u>Section 9</u> will not apply to the sale of any Option Shares to an underwriter pursuant to an underwriting agreement and shall only be applicable to Optionee if all then current executive officers and directors of the Company enter into similar agreements.
- 10. <u>Employment or Service</u>. Nothing in this Agreement or the Plan will interfere with or limit in any way the right of the Company or any Subsidiary to terminate the employment or service of Optionee at any time, nor confer upon Optionee any right to continue in the employment or other service with the Company or any Subsidiary.
- 11. Option Subject to Plan. The Option and the Option Shares granted and issued pursuant to this Agreement have been granted and issued under, and are subject to the terms of, the Plan. The terms of the Plan are incorporated by reference in this Agreement in their entirety, and Optionee, by execution of this Agreement, acknowledges having received a copy of the Plan. The provisions of this Agreement will be interpreted as to be consistent with the Plan, and any ambiguities in this Agreement will be interpreted by reference to the Plan. In the event that any provision of this Agreement is inconsistent with the terms of the Plan, the terms of the Plan will prevail. All of the capitalized terms used in this Agreement not otherwise defined in this Agreement have the same respective meanings as defined in the Plan.
 - 12. <u>General Provisions</u>.
 - 12.1. Governing Law; Venue. This Agreement and all rights and obligations under this Agreement will be governed by and construed exclusively in accordance with the laws

5

of the State of Delaware, notwithstanding the conflicts of laws principles of any jurisdictions. By acceptance of the Option, Optionee is deemed to submit to the exclusive jurisdiction and venue of the federal or state courts of the State of Illinois to resolve any and all issues that may arise out of or relate to the Option or this Agreement.

- 12.2. Entire Agreement. This Agreement and the Plan set forth the entire agreement and understanding of the parties to this Agreement with respect to the grant and exercise of the Option and the administration of the Plan and supersede all prior agreements, arrangements, plans and understandings relating to the grant and exercise of the Option and the administration of the Plan.
- 12.3. <u>Failure to Enforce Not a Waiver</u>. The failure of the Company or Optionee to enforce at any time any provision of this Agreement shall in no way be construed to be a waiver of such provision or of any other provision hereof.
- 12.4. Notices. All notices, requests, demands and other communications (collectively, "Notices") given pursuant to this Agreement shall be in writing, and shall be delivered by personal service, courier, facsimile transmission, email transmission of a pdf format data file or by United States first class, registered or certified mail, postage prepaid, addressed to the party at the address set forth on the signature page of this Agreement. Any Notice, other than a Notice sent by registered or certified mail, shall be effective when received; a Notice sent by registered or certified mail, postage prepaid return receipt requested, shall be effective on the earlier of when received or the third day following deposit in the United States mails. Any party may from time to time change its address for further Notices hereunder by giving notice to the other party in the manner prescribed in this Section 12.5.
- 12.5. <u>Successors and Assigns</u>. Except to the extent specifically limited by the terms and provision of this Agreement, this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors, assigns, heirs and personal representatives.
- 12.6. Execution. This Agreement may be executed in two or more counterparts, all of which when taken together shall be considered one and the same agreement and shall become effective when counterparts have been signed by each party and delivered to the other party, it being understood that both parties need not sign the same counterpart. In the event that any signature is delivered by facsimile transmission or by email delivery of a "pdf" format data file, such signature shall create a valid and binding obligation of the party executing (or on whose behalf such signature is executed) with the same force and effect as if such facsimile or "pdf" signature page were an original thereof.
- 12.7. <u>Titles, Captions and Sections</u>. Titles and captions contained in this Agreement are inserted for convenience of reference only and do not constitute a part of this Agreement for any other purpose. References to Sections in this Agreement refer to Sections of this Agreement unless otherwise stated.
 - 12.8. Nature of the Grant. In accepting the Option and by execution of this Agreement, Optionee acknowledges that:

- (a) The Plan is established voluntarily by the Company, it is discretionary in nature and it may be modified, amended, suspended or terminated by the Company in its sole discretion at any time, unless otherwise provided in the Plan.
- (b) The grant of the Option is voluntary and occasional and does not create any contractual or other right to receive future Option grants, or benefits in lieu of Option grants, even if Option grants have been granted repeatedly in the past.
 - (c) All decisions with respect to future Option grants, if any, will be at the sole discretion of the Company.
 - (d) Optionee is voluntarily participating in the Plan.
- (e) The Option grant is not part of normal or expected compensation or salary for any purposes, including, but not limited to, calculating any severance, resignation, termination, redundancy, end of service payments, bonuses, long-service awards, pension or retirement benefits or similar payments and in no event shall be considered as compensation for, or relating in any way to, past services for the Company.

	(f)	In the event that Optionee is not an employee of the Company, the Option will not be interpreted to form an employment contract or relationship with the Company.					
	(g) ed Purch	The future value of the Common Stock is unknown and cannot be predicted with certainty and if the Option vests and Optionee exercises the Option in accordance with the terms of rchased Shares, the value of those shares may increase or decrease.					
(h) In consideration of the grant of the Option, no claim or entitlement to compensation or damages shall arise from termination of the Option or diminution in value of the Option or Purchased Shares acquired upon exercise of the Option resulting from termination of Optionee's employment or service by the Company (for any reason whatsoever and whether or not in breach of local labor laws) and Optionee irrevocably releases the Company and its Subsidiaries, and their respective directors, officers, employees and agents, from any such claim that may arise; if, notwithstanding the foregoing, any such claim is found by a court of competent jurisdiction to have arisen, then, by acceptance of the Option and execution of this Agreement, Optionee shall be deemed irrevocably to have waived his or her entitlement to pursue such claim.							
(i) The Company is not providing any tax, legal or financial advice, nor is the Company making any recommendations regarding Optionee's participation in the Plan, or Optionee's purchase or sale of the underlying Option Shares.							
Plan or the Option.	(j) Optionee is hereby advised to consult with his or her own personal tax, legal and financial advisors regarding his or her participation in the Plan before taking any action related to that or the Option.						
[Remainder of page intentionally left blank; signature page follows]							
			7				
IN WITNESS WHEREOF, the parties to this Agreement have executed this Agreement effective as of the Grant Date. OPTIONEE: RESTORGENEX CORPORATION							
[C:zantum]			By: Name:				
[Signature] Name:			Title:				
Address:			Address:	2150 East Lake Cook Road, Suite 750 Buffalo Grove, IL 60089			
By execution of this Agreement, Optionee acknowledges having received a copy of the Plan.							
			8				